Mechanisms of anthelmintic resistance in Cooperia oncophora,

a nematode parasite of cattle

Annete Igandu Njue

Institute of Parasitology

McGill University,

Montreal

December 2003

A thesis submitted to McGill University in partial fulfillment of the requirements of the degree of Doctor of Philosophy

©Annete I. Njue 2003



Library and Archives Canada

Published Heritage Branch

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque et Archives Canada

Direction du Patrimoine de l'édition

395, rue Wellington Ottawa ON K1A 0N4 Canada

> Your file Votre référence ISBN: 0-612-98337-4 Our file Notre référence ISBN: 0-612-98337-4

NOTICE:

The author has granted a nonexclusive license allowing Library and Archives Canada to reproduce, publish, archive, preserve, conserve, communicate to the public by telecommunication or on the Internet, loan, distribute and sell theses worldwide, for commercial or noncommercial purposes, in microform, paper, electronic and/or any other formats.

The author retains copyright ownership and moral rights in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission.

AVIS:

L'auteur a accordé une licence non exclusive permettant à la Bibliothèque et Archives Canada de reproduire, publier, archiver, sauvegarder, conserver, transmettre au public par télécommunication ou par l'Internet, prêter, distribuer et vendre des thèses partout dans le monde, à des fins commerciales ou autres, sur support microforme, papier, électronique et/ou autres formats.

L'auteur conserve la propriété du droit d'auteur et des droits moraux qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

In compliance with the Canadian Privacy Act some supporting forms may have been removed from this thesis.

While these forms may be included in the document page count, their removal does not represent any loss of content from the thesis.



Conformément à la loi canadienne sur la protection de la vie privée, quelques formulaires secondaires ont été enlevés de cette thèse.

Bien que ces formulaires aient inclus dans la pagination, il n'y aura aucun contenu manquant. ABSTRACT -

Anthelmintic resistance is a major problem in livestock, and while it has been slower to emerge in cattle, there are reports of its occurrence. Three broad-spectrum anthelmintics are available for use, and one mechanism of resistance that is common to all is target site Glutamate-gated chloride channels (GluCls) are an important target for alteration. the lactone anthelmintics (MLs), while beta-tubulin represents macrocyclic benzimidazole (BZ) target. The objectives of this thesis were to determine whether GluCls are involved in ML resistance in the cattle parasite Cooperia oncophora, and whether beta-tubulin is involved in BZ and ML resistance. Two isolates of C. oncophora were used. In a fecal egg-count reduction test, ivermectin was found to be 100% effective against one isolate (IVS), and only 77.8% effective against the second isolate (IVR). Two full-length GluCl cDNAs, encoding GluCl α 3 and β subunits, were cloned. These subunits share high sequence identity with similar GluCl subunits from Haemonchus contortus and Caenorhabditis elegans. Genetic variability analysis of the two genes showed significant differences in allele frequencies between IVS and IVR worms at the GluCl α 3 gene, but not the GluCl β gene, suggesting that the GluCl α 3 gene is involved in ivermectin resistance. Sequencing of full-length GluCl subunit cDNAs from IVS and IVR worms revealed the presence of mutations in the N-terminal domains. Mutations in the GluCla3 caused modest but significant reductions in glutamate, ivermectin and moxidectin sensitivity, while mutations in the GluCl β abolished glutamate sensitivity. Of the three mutations identified in the IVR GluCl α 3, the L256F mutation accounted for the difference in glutamate and ivermectin response between IVS and IVR GluClo3 channels. Two beta-tubulin isotypes cloned from C. oncophora were found to share a high homology with beta-tubulin isotypes from other trichostrongylids. Genetic variability analysis of the two isotype genes revealed no association between beta-tubulin and ivermectin resistance. However, a small proportion of the IVR worms were found to carry the BZ resistance allele. The results of this thesis demonstrate the involvement of the GluCls in ivermectin resistance in C. oncophora, and the presence of a BZ resistance allele similar to that identified in trichostrongylids of sheep.

AGRÉGÉ -

La résistance anthelminthique est un problème d'importance et quoique ce phénomène ait été plus lent à apparaître chez les bovins, on retrouve des comptes rendus de son occurrence. Trois anthelminthiques à spectre étendue sont disponibles dont le mécanisme de résistance commun est l'altération de récepteurs cibles. Les canaux de chlorure glutamate dépendants (GluCls) sont une cible importante des anthelminthiques à lactones macrocycliques (MLs), tout comme l'est la tubuline-bêta pour les benzimidazoles (BZ). Les objectifs de cette thèse étaient de démontrer que les GluCls jouent un rôle au niveau de la résistance aux MLs du parasite des bovins, Cooperia oncophora, et que la tubulinebêta est impliquée dans la résistance aux BZ et MLs. Deux isolats de C. oncophora ont été utilisés. Lors d'un test coproscopique, Feacal egg count reduction test, l'ivermectine a eu une efficacité de 100% contre un des isolats (IVS), et de seulement 77,8% contre l'autre isolat (IVR). L'ADNc complet des GluCls qui encodent GluCla3 et les sousunités ß ont été clonés. Ces sous-unités partagent une grande identité séquentielle avec des sous-unités GluCls similaires d'Haemonchus contortus et Caenorhabditis elegans. L'analyse de la variabilité génétique de ces deux gènes a démontré des différences significatives dans la fréquence des allèles du gène GluCla3 entre des vers adultes IVS et IVR mais non avec le gène GluClβ. Ceci suggère que le gène GluClα3 joue un rôle dans la résistance à l'ivermectine. Le séquençage de l'ADNc complet de la sous-unité GluCl de vers adultes IVS et IVR a révélé la présence de mutations dans le domaine N-terminal. Les mutations de GluCla3 causent de modestes réductions significatives à la sensibilité au glutamate, à l'ivermectine et à la moxidectine. Cependant les mutations chez GluClß ont uniquement supprimé la sensibilité au glutamate. Des trois mutations identifiées chez GluCla3 IVR, la mutation L256F explique le type de réaction des GluCla3 IVS et IVR à l'ivermectine et le glutamate. Deux isotopes tubuline-bêta clonés de C. oncophora partagent une forte homologie avec les isotypes tubuline-bêta d'autres trichostrongles. L'analyse de la variabilité génétique de ces deux gènes isotypes n'a révélée aucune association entre la tubuline-bêta et la résistance à l'ivermectine. Cependant, une petite proportion des vers adultes IVR présentait des allèles de résistance aux BZ. Les résultats

iii

de cette thèse démontrent la participation des GluCls à la résistance à l'ivermectine chez *C. oncophora* et la présence d'un allèle de résistance aux BZ similaire à celle qui a été identifiée chez les trichostrongles du mouton. ACKNOWLEDGEMENTS

I would like to thank my advisor, Roger Prichard, for giving me the opportunity to study in Canada, where I've learned a great deal, both in and out of the lab. I am truly grateful for his mentorship and financial support. Other people I'd like to thank: Christiane, Marcelo and Yaroslav for their help with identifying and isolating the Cooperia eggs and larvae; Christiane, for translating my abstract; Shirley, for help with administrative matters; Gordy, for looking after the calves for all those years; Yeu-jin, for patiently teaching me the basics of molecular biology; Jeff, for being a great person to work with, for the useful discussions we've had over the years, and for sharing some valuable 'quick and easy' techniques; Dr. David Zadworny and his student, Babu, for help with the SSCP technique: Dayle, for her encouragement, for being a great friend, and a joy to be around; Xiao-peng and Bernie, for their insight, support and friendship- I am indebted to Bernie for kindly hosting me during the defense period; Sean and Bill, for taking the time to share their expertise when I started; Guan hua, Cat, Meredith, Jie, Yemisi, Megan, Anne, Amy, Mike, and Alain, for their friendship, and for the many discussions and laughs that we have shared; Sean and Alain, for helping me with the data analysis; Peter and Aws, for their friendship and kindness; Dr. Jim Smith, for help with computer programs; Dr. Jon Hayashi, for his tireless determination to get the McGill-FMC Agreement signed, and, with Lyle, for making electrophysiology such an interesting subject; Dr. Armando Jardim, for suggestions regarding the functional analysis of the GluCl receptors: Dr. Paula Ribeiro, for giving me the opportunity to be a TA for the biotech course, where I learnt a lot through interacting with the students; Mum, Dad, my siblings Larry, Jennifer, Tony, Nana and Kaari, for showing an interest in my work, for their encouragement, love and prayers; and my husband, Trevor, for his patience, love and support. In memory of my late grandfather,

Paul Njagi

CONTRIBUTIONS OF AUTHORS -

All experiments were designed and performed by the author, under the supervision of Prof. Roger Prichard. All data presented were obtained by the author, and all analyses were performed by the author. For the electrophysiology experiments (Chapter 4), Dr. Xiao-Peng Feng, who co-authored the paper, made key suggestions on the experimental design. The experiments were carried out at the FMC labs in New Jersey, under the expert guidance of Dr. Jon Hayashi and Lyle Kinne, who helped with frog dissections and recordings. The degenerate primers used for the initial isolation of a GABA β subunit fragment (Appendix A) were designed by Dr. Xiao-Peng Feng. All other primers were designed by the author. Animal infections and treatments were carried out by the author, with the assistance of Dayle Eshelby. This thesis was written by the author with editorial contributions from Prof. Roger Prichard.

CONTRIBUTION TO KNOWLEDGE IN THE FIELD –

In this thesis, we report the cloning of six full-length *C. oncophora* cDNAs, which have not previously been reported. The sequences have all been deposited in GenBank. This is the first study of mechanisms of anthelmintic resistance in a cattle nematode.

Glutamate-gated chloride channels (GluCls) have been implicated in ivermectin resistance in *Caenorhabditis elegans* and *Haemonchus contortus*, and here, we demonstrate that they are also involved in ivermectin resistance in *C. oncophora*.

Functional analysis of heterologously expressed *C. oncophora* GluCl subunits demonstrated that mutations affect agonist sensitivity. The findings that *C. oncophora* GluCl channels appear to be more sensitive to moxidectin than ivermectin, though ivermectin resistance selects for moxidectin resistance, are significant. Because of the higher sensitivity of moxidectin, this drug may be useful against ivermectin-resistant parasites.

Analysis of *C. oncophora* beta-tubulin isotype 1 demonstrated the presence of a mutation that has previously been shown to confer benzimidazole resistance in nematodes of sheep.

TABLE OF CONTENTS

ABSTRACT	
ABRÉGÉ	 111
ACKNOWLEDGEMENTS	,V
DEDICATION	vii
CONTRIBUTIONS OF AUTHORS	•••••• viii
CONTRIBUTION TO KNOWLEDGE IN THE FIELD	•••••• ix
TABLE OF CONTENTS	· · · · · · · · · · · · · · · · · · ·
LIST OF FIGURES	xiv
LIST OF TABLES	••••• XVİ
INTRODUCTION	XVIII

CHAPTER 1- Literature Review	1
1.1 Cooperia oncophora	2
1.2 Control of Gastrointestinal Helminths Using Anthelmintics:	
Its Effect on the Development of Immunity	
1.3 Control of Gastrointestinal Helminths: Anthelmintics Used in Cattle	
1.4 The Benzimidazoles and their Target Site	6
1.5 The Macrocyclic Lactones and their Target Sites	
1.5.1 The Macrocyclic Lactones	8
1.5.2 Biological Effects of the Macrocyclic Lactones on Nematodes	12
1.5.3 Ivermectin Sites of Action- the GABACls	
1.5.4 GABA- and Glutamate-gated Chloride Channels Belong to	
the 'cys loop' Superfamily of Ligand Gated Ion Channels	15
1.5.5 The GluCls	
1.5.6 Ivermectin's Interaction with Mammalian Host Targets	
1.6 Anthelmintic Resistance	
1.7 Benzimidazole Resistance Mechanisms	
1.8 Macrocyclic Lactone Resistance Mechanisms	
1.9 Anthelmintic Resistance Detection Methods	
References	

VARIAL BALLON MO	Field Isolata 57
	riela isolate
Abstract	50
2.1 Introduction	20
2.2 Materials a	ind Memods
2.2.1 Pa	
2.2.2 An	Imais
2.2.3 Fe	cal Egg Count Reduction Test (FECRT)
2.2.4 Ca	iculating Reduction in worm Burdens
2.3 Results	۵۷ ۵۷ ۸۸
2.4 Discussion	۵۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰
Acknowledgen	1ents
References	
CONNECTIN	IG STATEMENT I
	Constant Vanishility of Clastomate acted Chamida Champal Conog
CHAPTER 3:	Genetic Variability of Glutamate-gated Chloride Channel Genes
CHAPTER 3:	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant <i>Cooperia oncophora</i>
CHAPTER 3: Abstract	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant <i>Cooperia oncophora</i>
CHAPTER 3: Abstract 3.1 Introduction 2.2 Mathematical and the second sec	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant <i>Cooperia oncophora</i>
CHAPTER 3: Abstract 3.1 Introduction 3.2 Materials a	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant <i>Cooperia oncophora</i>
CHAPTER 3: Abstract 3.1 Introduction 3.2 Materials a 3.2.1 Par	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora
Abstract 3.1 Introductic 3.2 Materials a 3.2.1 Par 3.2.2 Iso	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora
CHAPTER 3: Abstract 3.1 Introduction 3.2 Materials a 3.2.1 Part 3.2.2 Iso 3.2.3 PC	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora
CHAPTER 3: Abstract 3.1 Introductic 3.2 Materials a 3.2.1 Par 3.2.2 Iso 3.2.3 PC Ma	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora
CHAPTER 3: Abstract 3.1 Introduction 3.2 Materials a 3.2.1 Part 3.2.2 Iso 3.2.3 PC Ma 3.2.4 SS	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora
Abstract 3.1 Introduction 3.2 Materials a 3.2.1 Part 3.2.2 Iso 3.2.3 PC Ma 3.2.4 SS 3.2.5 Sec	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora
Abstract 3.1 Introduction 3.2 Materials a 3.2.1 Part 3.2.2 Iso 3.2.3 PC Ma 3.2.4 SS 3.2.5 Sec 3.3 Results	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora 74 75 76 77 76 77 78 1ation of Full-length C. oncophora GluCl cDNAs Using RT-PCR 78 78 79 78 79 78 120 78 121 78 122 78 123 74 75 76 77 76 77 78 124 78 125 78 126 79 127 78 79 79 70 71 72 73 74 75 76 77 78 79
Abstract 3.1 Introduction 3.2 Materials a 3.2.1 Part 3.2.2 Iso 3.2.3 PC Ma 3.2.4 SS 3.2.5 Sec 3.3 Results 3.4 Discussion	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora 74 75 75 76 76 77 76 78 78 lation of Full-length C. oncophora GluCl cDNAs Using RT-PCR 78 ?R Amplification of Genomic DNA from Individual 81 CP Analysis 82 quence Analysis 83 95
Abstract 3.1 Introduction 3.2 Materials a 3.2.1 Par 3.2.2 Iso 3.2.3 PC Ma 3.2.4 SS 3.2.5 Sec 3.3 Results 3.4 Discussion Acknowledgem	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora
Abstract 3.1 Introduction 3.2 Materials a 3.2.1 Part 3.2.2 Iso 3.2.3 PC Ma 3.2.4 SS 3.2.5 Sec 3.3 Results 3.4 Discussion Acknowledgerr References	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora 74 75 75 on 76 and Methods 78 rasites 78 lation of Full-length C. oncophora GluCl cDNAs Using RT-PCR 78 R Amplification of Genomic DNA from Individual 81 CP Analysis 82 quence Analysis 83 95 99 100 100
Abstract 3.1 Introductic 3.2 Materials a 3.2.1 Par 3.2.2 Iso 3.2.3 PC Ma 3.2.4 SS 3.2.5 Sec 3.3 Results 3.4 Discussion Acknowledgem References	Genetic Variability of Glutamate-gated Chloride Channel Genes in Ivermectin-susceptible and -resistant Cooperia oncophora 75 on 76 nd Methods 78 rasites 78 lation of Full-length C. oncophora GluCl cDNAs Using RT-PCR 78 79 78 78 PR Amplification of Genomic DNA from Individual 100 100

CHAPTER 4: Mutations in the Extracellular Domains of Glutamate-gated
Chloride Channel $\alpha 3$ and β Subunits from Ivermectin-resistant
Cooperia oncophora Affect Agonist Sensitivity
Abstract 109
4.1 Introduction
4.2 Materials and Methods 112
4.2 1 Generation of Full-length cDNAs, and cRNA Synthesis 112
4.2.2 Oocyte Preparation and Injection 113
4.2.3 Electrophysiological Recordings 114
4.2.4 Drugs 116
4.3 Results
4.4 Discussion
Acknowledgements 143
References144
CONNECTING STATEMENT III 151
CONNECTING STATEMENT III
CONNECTING STATEMENT III
CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from
CONNECTING STATEMENT III
CONNECTING STATEMENT III
CONNECTING STATEMENT III
CONNECTING STATEMENT III CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from <i>Cooperia oncophora</i> , and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates
CONNECTING STATEMENT III CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from <i>Cooperia oncophora</i> , and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates
CONNECTING STATEMENT III CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from <i>Cooperia oncophora</i> , and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates
CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from <i>Cooperia oncophora</i> , and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates
CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from <i>Cooperia oncophora</i> , and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates
CONNECTING STATEMENT III CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from <i>Cooperia oncophora</i> , and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates
CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from <i>Cooperia oncophora</i> , and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates
CONNECTING STATEMENT III
CONNECTING STATEMENT III CONNECTING THE INTERTION OF COMPARIANCE INTERSTITE Abstract 153 5.2.1 Parasites 156 5.2.1 Parasites 156 5.2.2 Cloning of the Full-length Beta-tubulin Isotypes 1 and 2 cDNAs from <i>C. oncophora</i> 156 5.2.3 Determining the Presence/Absence of BZ Resistance in IVS and IVR <i>C. oncophora</i> 159 5.2.4 Analyzing Genetic Variability Using SSCP, and Sequencing of Alleles 159 5.2.5 Sequence Analysis 160
CONNECTING STATEMENT III
CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from <i>Cooperia oncophora</i> , and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates
CONNECTING STATEMENT III CHAPTER 5: Cloning Two Full-length Beta-tubulin Isotype cDNAs from Cooperia oncophora, and Screening for Benzimidazole Resistance-associated Mutations in Two Isolates Abstract 152 Abstract 153 5.1 Introduction 154 5.2.1 Parasites 156 5.2.2 Cloning of the Full-length Beta-tubulin Isotypes 1 and 2 cDNAs from C. oncophora 156 5.2.3 Determining the Presence/Absence of BZ Resistance in IVS and IVR C. oncophora 159 5.2.4 Analyzing Genetic Variability Using SSCP, and Sequencing of Alleles 160 5.2.4 Analyzing Genetic Variability Using SSCP, and Sequencing of Alleles 160 5.2.5 Sequence Analysis 160 5.3 Results 174 Acknowledgements 178

CHAPTER 6: General Discussion
References 195
INTRODUCTION TO APPENDICES
APPENDIX A: Cloning a GABA-gated Chloride Channel cDNA from Cooperia
oncophora
A.1 Introduction
A.2 Materials and Methods 203
A.2.1 Cloning of the full-length C. oncophora GABAβ cDNA
A.3 Results
A.4 Discussion
References
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluCl of and
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co $GluCl \alpha$ and $GluCl\beta$ genes
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClα3 andGluClβ genesB.1 Analysis of the Co GluClα3 gene218
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClc3 and GluCl β genes
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClc3 and GluCl β genes
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClo3 and GluClβ genes 216 B.1 Analysis of the Co GluClo3 gene 218 B.2 Analysis of the Co GluClβ gene 221 APPENDIX C: GenBank Submissions of C. oncophora Sequences Cloned 224
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClo3 and 216 B.1 Analysis of the Co GluClo3 gene 218 B.2 Analysis of the Co GluClβ gene 218 APPENDIX C: GenBank Submissions of C. oncophora Sequences Cloned 224 C.1 GluClo3 225
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClα3 and 216 B.1 Analysis of the Co GluClα3 gene 218 B.2 Analysis of the Co GluClβ gene 218 APPENDIX C: GenBank Submissions of C. oncophora Sequences Cloned 224 C.1 GluClα3 225 C.2 GluClβ 227
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClc3 and GluCl β genesB.1 Analysis of the Co GluCl α 3 gene218B.2 Analysis of the Co GluCl β gene221APPENDIX C: GenBank Submissions of C. oncophora Sequences ClonedC.1 GluCl α 3225C.2 GluCl β 227C.3 Beta-tubulin Isotype 1229
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClc3 and GluCl β genes216B.1 Analysis of the Co GluClc3 gene218B.2 Analysis of the Co GluCl β gene221APPENDIX C: GenBank Submissions of C. oncophora Sequences Cloned224C.1 GluClc3225C.2 GluCl β 227C.3 Beta-tubulin Isotype 1229C.4 Beta-tubulin Isotype 2231
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluCl α 3 and GluCl β genes216B.1 Analysis of the Co GluCl α 3 gene218B.2 Analysis of the Co GluCl β gene221APPENDIX C: GenBank Submissions of C. oncophora Sequences Cloned224C.1 GluCl α 3225C.2 GluCl β 227C.3 Beta-tubulin Isotype 1229C.4 Beta-tubulin Isotype 2231C.5 Actin233
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClc3 and GluCl\beta genesB.1 Analysis of the Co GluClc3 gene216B.2 Analysis of the Co GluCl β gene221APPENDIX C: GenBank Submissions of C. oncophora Sequences ClonedC.1 GluClc3225C.2 GluCl β 227C.3 Beta-tubulin Isotype 1229C.4 Beta-tubulin Isotype 2231C.5 Actin233C.6 GABA β 235
APPENDIX B: SSCP analysis of intracytoplasmic loop region of Co GluClc3 and GluCl β genes216B.1 Analysis of the Co GluCl α 3 gene218B.2 Analysis of the Co GluCl β gene221APPENDIX C: GenBank Submissions of C. oncophora Sequences Cloned224C.1 GluCl α 3225C.2 GluCl β 227C.3 Beta-tubulin Isotype 1225C.4 Beta-tubulin Isotype 2231C.5 Actin233C.6 GABA β 235

xiii

------ LIST OF FIGURES ------

CHAPTER 1

Figure 1:	Chemical structures of ivermectin and moxidectin	12
Figure 2:	Topological organization of cys-loop receptors	17
Figure 3:	Schematic drawing of the muscle-type $(\alpha 1)_2\beta 1\gamma\delta$ nAchR	19

CHAPTER 3

Figure 1:	Neighbor-joining tree showing relationships of the C. oncophora GluCl	
	subunits with full-length GluCls from H. contortus and C. elegans	36
Figure 2A:	Alignment of GluCla3 deduced amino acid sequences from	
	C. oncophora, H. contortus and C. elegans	39
Figure 2B:	Alignment of GluCl β deduced amino acid sequences from	
	C. oncophora, H. contortus and C. elegans) 0
Figure 3A:	SSCP analysis: Agarose gel showing quality of the PCR products	
	derived from 10 individual male worms) 1
Figure 3B:	SSCP analysis: SSCP patterns of 11 individual IVS worms at the	
	Co $GluCl\beta$ gene) 1
Figure 4:	Co GluClo3 allele frequencies in IVS and IVR worms	92
Figure 5:	Co $GluCl\beta$ allele frequencies in IVS and IVR worms) 3
Figure 6A:	Positions of polymorphic sites in the region of the Co GluClas	
	gene analyzed	94
Figure 6B:	Positions of polymorphic sites in the region of the Co $GluCl\beta$	
	gene analyzed	94

CHAPTER 4

Figure 1A:	Comparing the predicted amino acid sequences of the
	C. oncophora GluCla3 subunits from IVS and IVR worms 118
Figure 1B:	Comparing the predicted amino acid sequences of the
	<i>C. oncophora</i> GluClβ subunits from IVS and IVR worms
Figure 2A:	Representative current traces for homomeric and heteromeric
	receptors following activation with glutamate 123

Figure 2B:	Dose-response curves for glutamate-induced currents for oocytes
	expressing homomeric (α 3 or β) receptors
Figure 2C:	Dose-response curves for glutamate-induced currents for oocytes
	expressing homomeric (α 3) and heteromeric (α 3 + β) receptors 124
Figure 3:	Current-voltage relationship of the glutamate-gated currents
	of the IVS GluCla3 receptor
Figure 4A:	Currents activated by glutamate, ivermectin and moxidectin
	in oocytes expressing IVS GluCla3
Figure 4B:	Ivermectin and moxidectin dose-response curves for oocytes
	expressing IVS and IVR GluCla3 128
Figure 5A:	Glutamate dose-response curves for mutant GluCl α 3 receptors 131
Figure 5B:	Ivermectin dose-response curves for mutant GluCl α 3 receptors 131
Figure 6:	Currents activated by glutamate and ibotenate in oocytes
	expressing IVS GluCl α 3 and GluCl β receptors

CHAPTER 5

Figure 1:	Alignment of Co b1 and Co b2 sequences with other	
	trichostrongylid beta-tubulins	164
Figure 2:	Neighbour-joining tree showing the relationships among	
	beta-tubulin predicted proteins	169
Figure 3:	SSCP of Co b2 gene	

APPENDIX A

Figure 1:	Sequence of the <i>C. oncophora</i> GABAβ cDNA	207
Figure 2:	Alignment of GABA β predicted amino acid sequences	
	from C. oncophora, H. contortus and C. elegans	209

----- LIST OF TABLES ------

CHAPTER 1

Table 1:	Ivermectin given subcutaneously at 200 μ g kg ⁻¹ is effective against
	a wide range of endo- and ecto-parasites in cattle

CHAPTER 2

Table 1:	Pre- and post-treatment fecal egg counts from untreated calves and	
	calves treated with injectable ivermectin \dots 6	53
Table 2:	Post-treatment worm burdens in treated and untreated calves	54

CHAPTER 3

Table 1:	Primers used to amplify fragments of Co $GluClo3$ and Co $GluCl\beta$	
	genes from individual male worms for SSCP	32
Table 2:	Amino acid sequence identities of the Co GluCl α 3 and GluCl β	
	Subunits with other GluCl sequences reported in GenBank	35

CHAPTER 4

Table 1:	C. oncophora GluCla3 mutants tested	113
Table 2:	Pharmacology of Co GluCl α 3 and GluCl β homomeric channels	133
Table 3A:	Summary of effects of glutamate on IVS, IVR and mutant	
	GluCl receptors	135
Table 3B:	Summary of effects of ivermectin and moxidectin on IVS, IVR	
	and mutant GluCl receptors	135

CHAPTER 5

Table 1:	Comparison of <i>C. oncophora</i> beta-tubulin (Co b1 and Co b2)	
	predicted proteins with other beta-tubulins for identity 16	57
Table 2:	Determining the genotypes of individual worms based on amino	
	acids at positions 167 and 200 of Co b1 gene 17	70
Table 3:	Allele frequencies for Co b1 and Co b2 genes in IVS and IVR worms 17	73



APPENDIX A

 INTRODUCTION -

The benzimidazoles and the macrocyclic lactone (ML) anthelmintics are broad spectrum anthelmintics that are widely used to treat helminth infections in domestic animals. The benzimidazoles act by binding to beta-tubulin, thereby interfering with microtubule-based processes such as cell division. The MLs target chloride channels gated by glutamate and gamma aminobutyric acid (GABA) and interfere with neurotransmission. These chloride channels belong to the cys-loop superfamily of receptors, which mediate fast chemical neurotransmission, and include the nicotinic acetylcholine receptors and the mammalian glycine receptors.

The effectiveness of anthelmintics in controlling helminth infections is threatened by the development of resistance. Resistance is widespread in nematodes of sheep, and has been slower to develop in cattle parasites. However, there are reports of anthelmintic resistance in cattle. Benzimidazole resistance has been found in several cattle nematode species, including *Cooperia oncophora*, *Ostertagia ostertagi* and *Trichostrongylus axei*. With ML resistance, all cases that have been reported so far involve *Cooperia* species. The mechanisms of benzimidazole resistance are well established, and are known to involve the target, beta-tubulin. ML resistance mechanisms are less well understood. The main objective of this thesis is to determine the role of the glutamate-gated chloride channel (GluCl) genes in ML resistance in *Cooperia oncophora*. As well, the role of beta-tubulin in benzimidazole and ML resistance is explored.

A review of what is currently known about the modes of action and mechanisms of resistance to the benzimidazoles and MLs is presented in Chapter 1. Information relating to anthelmintic use in cattle is also provided. Chapter 2 describes the ivermectin susceptibility and resistance status of the two *C. oncophora* field isolates used throughout the study. The two isolates differ in their sensitivities to ivermectin, with one being 100% susceptible, and the other, resistant, providing a basis for comparison of the two at the genetic and functional levels for changes which can be linked to ivermectin resistance. Chapter 3 describes the cloning of two GluCl genes from *C. oncophora*, and the genetic variability data presented provides indirect evidence that one of these genes, GluCl α 3, may be involved in ivermectin resistance in this nematode. Previous studies have shown that the *C. elegans* orthologue of this subunit has a binding site for ivermectin. Chapter 4 provides direct evidence that ivermectin interacts with receptors formed by the *C. oncophora* GluCl α 3 subunit in a heterologous expression system, and that mutations found in GluCl subunits of ivermectin-resistant worms significantly reduce agonist sensitivity.

Beta-tubulin has been implicated in ivermectin resistance in *Haemonchus* contortus, a nematode parasite of sheep. Chapter 5 describes the cloning of two *C*. oncophora beta-tubulin isotypes cDNAs. While no association was found between beta-tubulin and ivermectin resistance in *C. oncophora*, the benzimidazole resistance-associated mutation was found in one of the field isolates. In Chapter 6, the findings presented in this thesis are discussed in light of what is currently known about cys-loop receptors, as well as anthelmintic resistance.



LITERATURE REVIEW

1.1 Cooperia oncophora

Cooperia oncophora is a parasitic gastrointestinal nematode of cattle belonging to the superfamily Trichostrongyloidea. It is a common and economically important nematode in temperate regions (Coop et al. 1979; Kloosterman et al. 1984; De Graaf et al. 1992; Parmentier et al. 1995). It is found in the small intestine, and has a direct life cycle, with free-living first and second larval stages, and an infective third larval stage (L3) (Bowman et al. 2003). The first and second stage larvae feed on bacteria, while thirdstage larvae, which retain the cuticle of the second-stage larvae during the second molt, do not feed. Infection is by ingestion of L3, which exsheath in the abomasum before reaching their predilection site. Once in the small intestine, they penetrate the crypts between the villi, and while in the crypts, molt twice to reach the immature adult stage (L5). Maturation to adult stage (from ingestion) takes 17-22 days (Isenstein 1963). C. oncophora is a lumen-dwelling nematode that, as an adult, lives in the crypts between the villi, or attaches to the mucus lining the intestine (Herlich 1965). However, the fourthstage larvae penetrate the intestinal mucosa during development, contributing to the pathology of infection (Armour et al. 1987; Parkins et al. 1990). While worms are typically restricted to the anterior part of the small intestine during infection, they can also be found in the ileum and jejunum when infection levels are high (Armour et al. 1987; Parkins et al. 1990; Bowman et al. 2003).

Immunity to *Cooperia* infections develops with age, and young animals up to two years of age are most at risk of succumbing to infection (Armour 1989; Kloosterman et al. 1991; Satrija and Nansen 1993). However, mature cattle that have not been exposed to sufficient contact with worms remain susceptible to infection. Mature animals that have acquired immunity harbor small populations of nematode parasites that are a source of constant pasture contamination (Corwin 1997). *C. oncophora* infections are usually sub-clinical, limiting weight gains in young growing calves, and can lead to poor production in later years (Herlich 1965; Ploeger et al. 1990; Hawkins 1993). In older dairy animals, sub-clinical parasitism reduces milk production (Thomas et al. 1984; Bisset et al. 1987a, b; Gross et al. 1999). Massive infections can, however, cause clinical disease, and the severity of the pathological changes and clinical signs correlates directly with the level of infection (Herlich 1965; Armour et al. 1987). A tentative diagnosis of helminth infections in general can be made based on the clinical signs, and confirmed by demonstrating strongyle-type eggs in feces (Fraser 1991). While *Cooperia* eggs can usually be differentiated from those of other trichostrongylid species by their shape, a definitive diagnosis of *Cooperia* infection can only be made by culturing feces and identifying the L3 stage (Fraser 1991). However, for treatment purposes, and when anthelmintic resistance is not an issue, this is not warranted, since the broad spectrum anthelmintics used are effective against all trichostrongylid species (Fraser 1991).

1.2 Control of Gastrointestinal Helminths Using Anthelmintics: Its Effect on the Development of Immunity

Control of gastrointestinal helminths, among them *C. oncophora*, relies on the use of broad-spectrum anthelmintics. These drugs are often used prophylactically to limit production loss caused by gastrointestinal parasites, and to reduce pasture contamination by limiting fecal egg output (Vercruysse et al. 1994, 1995). With the strategic use of these highly effective drugs, clinical parasitism is rare in the developed world, and

economic losses are mostly associated with sub-clinical parasitism (Corwin 1997; Gasbarre et al. 2001; Vercruysse and Claerebout 2001). However, suppression of nematode infections through intensive treatment of calves reduces host-parasite contact and interferes with the development of immunity during the first grazing season (FGS) (Kloosterman et al. 1991; Ploeger and Kloosterman 1993; Vercruysse et al. 1994, 1995; Claerebout et al. 1999; Claerebout 2002). Consequently, unexposed animals have heavier worm burdens in their second grazing season (Vercruysse et al. 1994; Satrija et al. 1996; Claerebout et al. 1999). Unexposed animals are susceptible to infection when they are older, and when anthelmintics are not usually being used to control helminth infections (Ploeger and Kloosterman 1993; Dorny et al. 2000). Reducing the number of treatments during the grazing season affords some exposure to parasites, thus allowing immunity to develop (Vercruysse et al. 1995). While one or two treatments a year are sufficient to protect FGS calves against the effects of parasitism, reports show that anthelmintic use tends to be excessive (Ploeger and Kloosterman 1993; Ploeger et al. 2000; Claerebout 2002), and that on average, farmers give up to four treatments a year (Vercruysse and Dorny 1999). Bisset et al. (1987a, b) have shown that, in an all-year grazing management system, fewer curative calf treatments have more positive production effects on herds than regular preventive treatments. As well, fewer treatments cost less, and reduce the selection for anthelmintic resistance (Vercruysse and Claerebout 2001; Claerebout 2002).

While chemoprophylaxis may interfere with the development of immunity, some studies have reported that the effects of reduced immunity are limited, since animals in their second grazing season are exposed to light infections on pasture (Conder et al. 1983;

4

Guldenhaupt and Burger 1983; Jacobs et al. 1987; Claerebout et al. 1998). Under low levels of challenge, previously treated (unexposed) animals rapidly acquire immunity with exposure, and appear not to suffer the loss-producing effects of parasitism (Satrija et al. 1996; Suarez et al. 2001). However, with high larval challenge, untreated animals that have developed resistance to infection show significantly higher live-weight gains than previously treated (unexposed) animals (Suarez et al. 2001). Additional treatment of previously treated (unexposed) animals may be necessary to prevent sub-clinical losses and to limit pasture contamination, as these animals also have higher fecal egg counts (Suarez et al. 2001).

1.3 Control of Gastrointestinal Helminths: Anthelmintics Used in Cattle

Three classes of broad spectrum anthelmintics are used to control gastrointestinal helminth infections in cattle:- the benzimidazoles, the macrocyclic lactones, and the tetrahydropyrimidines and imidazothiazoles. Only the first two groups, which are dealt with in this thesis, will be discussed.

1.4 The Benzimidazoles and their Target Site

The benzimidazoles were introduced in the early 1960s and have been widely used in the human, veterinary and agricultural fields (Campbell 1990; Cook 1990; Horton 1990; Baraldi et al. 2003). They were the first group of anthelmintics on the market to show a broad spectrum of activity, a wide margin of safety, and efficacy against immature stages, and were used in strategic control programs to increase performance and maximize profits (Campbell 1990). Several benzimidazoles are available as oral formulations for use in cattle, including albendazole, oxfendazole and febendazole (Bowman et al. 2003).

The benzimidazoles exert their anti-parasitic effect by binding directly to betatubulin and preventing its polymerization into microtubules, thereby affecting various microtubule-based processes, which include cell division and intracellular transport (Lacey 1988). Initial studies on the mode of action of benzimidazoles suggested various mechanisms, including inhibition of fumarate reductase (Prichard 1973), inhibition of monoamine oxidase activity (Moreno and Barrett 1979), interference with glucose uptake (Van den Bossche and De Nollin 1973) and altered metabolic pathways (Sangster and Prichard 1984). These are thought to result from failure in microtubule function. The inhibitory effect of benzimidazoles on tubulin polymerization was first indicated by studies in fungi (Davidse 1973; Hammerschlag and Sisler 1973), and confirmed by cytological studies in mammalian cell lines and in rats (Styles and Garner 1974), and in the nematode *Ascaris suum* (Borgers and De Nollin 1975). Subsequently, Friedman and Platzer (1980) provided the first evidence of the higher affinity, and hence selective toxicity, of benzimidazoles for nematode tubulin as compared to mammalian tubulin. Six beta-tubulin genes have been identified in the free-living nematode *Caenorhabditis elegans* (Gogonea et al. 1999). One of these, mec-7 (tbb1), is expressed in the touch neurons, and, along with the alpha-tubulin isotype mec-12, is thought to influence the unique microtubule architecture in these neurons (Savage et al. 1989; Savage et al. 1994; Fukushige et al. 1999). In *Drosophila melanogaster*, the beta 2 isotype is expressed only in the testis, and its spermatogenic-specific functions cannot be performed by another *D. melanogaster* isotype, beta-3, or by a beta-2 ortholog from the moth *Heliothis virescens* (Hoyle and Raff 1990; Raff et al. 1997).

Among trichostrongylids, two distinct beta-tubulin isotypes have been described which can be differentiated based on the sequence at the carboxy terminal (Geary et al. 1992; Elard et al. 1996; Grant and Mascord 1996). The expression pattern and functional significance of these two isotypes are unknown. In vertebrates, seven beta-tubulin isotypes have been identified which show tissue-specific distribution and different relative stabilities (Luduena 1993; Roach et al. 1998; Schwarz et al. 1998). This suggests that isotypic differences may be important in regulating microtubule function *in vivo* (Schwarz et al. 1998). The isotype-defining carboxy termini of the differently with cellular factors (Nogales et al. 1998). Tubulin performs a wide variety of functions, and the various isotypes may differ in their abilities to fulfill these functions (Roach et al. 1998; Burkhart et al. 2001).



1.5 The Macrocyclic Lactones and their Target Sites

1.5.1 The Macrocyclic Lactones

The macrocyclic lactones (MLs) are broad spectrum anthelmintics that are widely used to control nematode infections in humans and animals (Campbell et al. 1983). They include the avermectins and milbemycins. The avermectins are natural fermentation products of the soil-dwelling actinomycete *Streptomyces avermectilis* (Campbell et al. 1983). Avermectins produced by this organism are a mixture of eight compounds, designated A_{1a} , A_{1b} , A_{2a} , A_{2b} , B_{1a} , B_{1b} , B_{2a} and B_{2b} (Campbell 1989). The B_1 homologs are the most potent and also have the broadest spectrum of activity among nematodes. The a and b homologs have almost identical activities, although the former are produced in greater amounts (Campbell 1989).

Ivermectin (22, 23-dihydroavermectrin B1) was the first ML to be developed for commercial use. Following its introduction in 1981, ivermectin has been widely used to treat nematode infections in domestic animals, and has been described as 'the most commercially successful veterinary parasiticide ever produced' (Raymond and Sattelle 2002). With its novel mode of action, ivermectin was effective against helminths of sheep that showed resistance to existing compounds (Shoop et al. 1995). In humans, ivermectin is used to control onchocerciasis and lymphatic filariasis (Campbell 1991; Brown et al. 2000). Ivermectin itself is a semi-synthetic derivative of avermectin B1 (Chabala et al. 1980). It is highly effective at a low dose, has a wide safety margin, and a broad spectrum of activity against a range of ecto- and endo-parasites, including nematodes (Table 1) (Campbell et al. 1983). However, these parasites develop at

8

different times, and ML's are used more frequently for effectiveness against endo- and ecto-parasites (Gasbarre et al. 2001). Ivermectin is not effective against trematodes and cestodes.

Table 1. Ivermectin given subcutaneously at 200 μ g kg⁻¹ is effective against a wide range of endo- and ecto-parasites in cattle. Adapted from Shoop et al. (1995).

NEMATODA

Gastrointestinal worms Bunostomum phlebotomum Cooperia oncophora Cooperia pectinata Cooperia punctata Cooperia species Haemonchus placei Mecistocirrus digitatus Nematodirus helvetianus Nematodirus spathiger *Oesophagostomum radiatum* Ostertagia lyrata Ostertagia ostertagi Strongyloides papillosus Toxocara vitulorum Trichostrongylus axei Trichostrongylus colubriformis Trichuris spp.

Lungworms Dictyocaulus viviparous

Skin worms Parafilaria bovicola

Eyeworms Thelazia spp.

ARTHROPODA

Cattle grubs Dermatobia hominis Hypoderma bovis Hypoderma lineatum

Screwworm fly larvae Chrysomya bezziana

Cochliomyia hominivoraz

Sucking lice

Haematopinus eurysternus Linognathus vituli Solenopotes capillatus

Biting lice Damalinia bovis

Mange mites Psoroptes ovis Sarcoptes scabiei var. bovis Chorioptes bovis

Ticks Boophilus microplus Boophilus decoloratus Ornithodorus savignyi Ivermectin is available in topical, injectable (subcutaneous), and oral (liquid, paste, and slow-release bolus) formulations for use in cattle (West et al. 1994; Vermunt et al. 1995). Injectable and oral formulations are given at a dosage level of 0.2 mg/kg body weight, while the topical formulation is given at 0.5 mg/kg body weight (Campbell 1989). Cattle seem to tolerate up to a maximum dose of 1 mg/kg ivermectin, above which animals show signs of central nervous system depression, including ataxia, which progress to paresis, recumbency, decreased lip and tongue tone, drooling, coma and death (Campbell 1989). Administration of propylene glycol, in which ivermectin is diluted, at 30 times the use level (6 mg/kg) causes no ill effects (Campbell 1989).

Although ivermectin has broad spectrum activity against gastrointestinal nematodes, its efficacy varies with target species (Shoop et al. 1996a). Ivermectin has better activity against large intestinal- and abomasal- dwelling parasites than against those in the small intestine of sheep and cattle. The topical formulation has been found to have lower-than-expected efficacy against *Cooperia* species in cattle (Bisset et al. 1990; Eagleson and Allerton 1992; West et al. 1994; McKenna 1995). *C. oncophora* and *Nematodirus helvetianus*, both small intestinal parasites, are considered the dose limiting species for ivermectin in cattle, and the recommended dosage is determined based on efficacy against these species (Egerton et al. 1981; Shoop et al. 1995). Ivermectin's lower efficacy against small intestinal nematodes may not be related to the drug concentration available at the target site, since concentrations of ivermectin in abomasal and in small intestinal mucus are similar (Bogan and McKellar 1988). It is possible that small intestinal nematodes may recover from ivermectin's temporary paralytic effects and

10

re-establish in a more distal location in the small intestine (Bogan and McKellar 1988; Bogan et al. 1988).

Other avermectins that are available for treatment of internal and external parasites in cattle include doramectin and eprinomectin (Shoop et al. 1996b; Bowman et al. 2003). Eprinomectin is unique among avermectins in that it can be used in lactating animals, since it exhibits minimal residue in milk (Shoop et al. 1996a). It is available as a convenient topical formulation, and retains the high potency and wide spectrum of activity that is characteristic to the avermectins (Shoop et al. 1996b). The availability of such anthelmintics with zero withholding time, which can be used in lactating dairy cattle to increase production by alleviating sub-clinical parasitism, has consequences for the development of anthelmintic resistance (Reinemeyer 1995; Morris et al. 2002).

The milbemycins are closely related to the avermectins. Moxidectin, the only milbemycin used in cattle, is available in topical (0.5 mg/kg) and injectable (0.2 mg/kg) formulations (Rock, 2001). It is a chemically altered derivative of F-alpha, a fermentation product of *Streptomyces cyaneogriseus noncyanogenus* (Bowman et al. 2003). While ivermectin and moxidectin are similar in structure, differences include a dissacharide at carbon-13 unique to ivermectin, a methoxine moiety at carbon-23 found only in moxidectin, and different side chains in the two compounds at carbon-25 (Figure 1). Moxidectin's range of activity is similar to that shown by ivermectin. However, compared to ivermectin, moxidectin shows longer persistence and higher efficacy against tissue-dwelling nematodes, while ivermectin is more potent against ectoparasites (Shoop et al. 1995). As with ivermectin, the two dose-limiting species for moxidectin in cattle are *C. oncophora* and *N. helvetianus* (Ranjan et al. 1992; Scholl et al. 1992).



Figure 1. Chemical structures of ivermectin and moxidectin. Adapted from Rock, 2001.

1.5.2 Biological Effects of Macrocyclic Lactones on Nematodes

The effects of MLs have been described in several nematode species. Most experiments have been carried out using ivermectin or avermectin. Avermectin induces irreversible paralysis of somatic musculature in *A. suum*, *C. elegans* and *Haemonchus contortus* (Kass et al. 1980; Geary et al. 1993). In *H. contortus*, the ivermectin-induced paralysis is restricted to the mid-body of the worm, with the head and tail sections showing normal movement (Geary et al. 1993). Ivermectin has also been shown to inhibit motility of the free-living stages of several trichostrongylid species, including *H. contortus*, *Trichostrongylus colubriformis* and *Ostertagia circumcincta* (Ibarra and Jenkins 1984; Gill et al. 1991). Ivermectin's inhibitory effect on motility is the basis of an *in vitro* larval motility assay designed to detect resistance in *H. contortus* (Gill et al. 1991).

Ivermectin also inhibits feeding in adult nematodes, including *T. colubriformis*, *H. contortus* and *C. elegans* (Bottjer and Bone 1985; Avery and Horvitz 1990; Geary et al. 1993). The inhibition of feeding is due to specific inhibition of pharyngeal pumping by ivermectin (Martin 1996). The pharyngeal muscle is far more sensitive to the effects of ivermectin than somatic musculature, and concentrations of ivermectin required to inhibit pharyngeal pumping are 100-fold lower than those required to inhibit motility in *H. contortus* (Geary et al. 1993). This suggests that the pharynx is the main target for the MLs. Inhibition of pharyngeal pumping has also been demonstrated in *H. contortus* larval stages (Kotze 1998). Ivermectin is also a potent inhibitor of larval development (Gill et al. 1995). This occurs at the L1 stage at a concentration of ivermectin comparable to that which inhibits pharyngeal pumping. Therefore, it is possible that inhibition of pharyngeal pumping caused by ivermectin results in starvation, which then leads to inhibition of larval development (Gill et al. 1995).

Several anthelmintics, including ivermectin, levamisole and oxfendazole, suppress oviposition in *T. colubriformis* (Bottjer and Bone 1985). Ivermectin was found to have the most pronounced effect in both *in vivo* and *in vitro* tests (Bottjer and Bone 1985). Ivermectin's effect on fertility has also been observed in *C. elegans*, where it not only inhibits the release of eggs already present in the uterus, but it also suppresses the production of new eggs (Grant 2000). Ivermectin also suppresses the production of new eggs (Grant 2000). Ivermectin also suppresses the production of new eggs (Grant 2000). Ivermectin also suppresses the production of new eggs (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also suppresses the production of new *eggs* (Grant 2000). Ivermectin also been observed in the dog filarid Dirofilaria immitis following treatment with milbemycin D (Sasaki and Kitagawa 1993).

Among filarial nematodes, therefore, the reproductive tract appears to be an important target for MLs.

1.5.3 Ivermectin Sites of Action- the GABACls

compounds, MLs, organophosphates, including the anthelmintic Several imidazothiazoles, tetrahydopyrimidines and piperazine, exert their nematocidal effects by interrupting neuromuscular transmission (Sangster 1996). One of the earliest studies on ivermectin's mode of action suggested that the drug acted on a chloride channel, since it induced paralysis of lobster stretcher muscle by increasing the membrane permeability to chloride ions (Fritz et al. 1979). Ivermectin's effect was blocked by the y-aminobutyric acid (GABA) antagonist picrotoxin, and the authors suggested that the drug was acting on chloride channels gated by GABA. Other electrophysiological studies in the nematodes A. suum and C. elegans showed that avermeetin blocked transmission between interneurons and excitatory motor neurons, leading to paralysis of the somatic musculature (Kass et al. 1980). Similar effects were produced by the GABAergic agonists muscimol and piperazine, and as with avermectin, the effects were reversed by picrotoxin (Kass et al. 1984). These results suggested that avermectin was acting as a GABA agonist at a GABA receptor. The experiments of Holden-Dye and Walker (1990) suggested that ivermectin was acting as a GABA antagonist on somatic muscle cells of A. suum. In all these studies, the high (micromolar) concentrations of ivermectin required to elicit an effect were inconsistent with the high in vivo potency of this drug (Geary et al. 1993). Work by Duce and Scott (1985) showed that, in addition to activating GABAsensitive chloride channels, low ivermectin concentrations increased chloride conductance in GABA-insensitive muscle bundles. The glutamate analog ibotenate produced a dose-dependent increase in chloride conductance in these GABA-insensitive bundles, and demonstrated that ivermectin was interacting with chloride channels that were gated by glutamate. These receptors, which were originally identified as H-(hyperpolarization) receptors (Cull-Candy and Usherwood 1973; Lea and Usherwood 1973a, b), are widely distributed on locust muscle cells (Cull-Candy 1976).

1.5.4 GABA- and Glutamate-gated Chloride Channels Belong to the 'cys loop' Superfamily of Ligand Gated Ion Channels

GABA- and glutamate-gated chloride channels (GABA-Cls and GluCls) are ligand gated ion channels of the 'cys loop' superfamily of receptors, which mediate fast chemical neurotransmission (Raymond and Sattelle 2002). Members of this superfamily include nicotinic acetylcholine (nAch), serotonin (5-HT₃), GABA_A and glycine receptors, and the recently characterized serotonin (MOD-1) and histamine-gated chloride receptors (Ortells and Lunt 1995; Ranganathan et al. 2000; Gisselmann et al. 2002; Zheng et al. 2002). The nAch and 5-HT₃ receptors are excitatory cation channels, and the GluCls, glycine, GABA_A, MOD-1 and histamine receptors are inhibitory chloride channels. GluCls are found only in invertebrates, while glycine-gated chloride channels have been found only in vertebrates (Vassilatis et al. 1997b). MOD-1 and histamine chloride channels have been found only in *C. elegans* and arthropod photoreceptors, respectively (Hardie 1989; Ranganathan et al. 2000; Gisselmann et al. 2002; Zheng et al. 2002). Phylogenetic analysis shows that the GluCls are a distinct group within the ligand-gated ion channel superfamily, and are most closely related to the vertebrate glycine-gated chloride channels (Vassilatis et al. 1997b).

The cys-loop receptors are believed to be structurally similar, and are made up of five subunits that are arranged together around a central pore (Figure 2A). This characteristic image was first derived from electron microscopic images of nAchR from the electric fish, Torpedo (Brisson and Unwin 1985). Native receptors are heteropentameric in structure, though some nAchR subunits like α 7 and α 8 can form homopentameric receptors in vitro which display pharmacological properties resembling those of the native receptor (Karlin et al. 1983; Bechade et al. 1994; Gerzanich et al. 1994; Chang et al. 1996). The different subunits are encoded by closely related genes, and the diversity of native receptors is increased by alternative splicing of these genes (Whiting et al. 1990). Each subunit is made up of a number of domains; an N-terminal region, which is located extracellularly and carries the ligand binding domain, four transmembrane domains, with the second transmembrane domain lining the channel pore, a long intracytoplasmic loop between the third and fourth transmembrane domains, and an extracellular carboxy terminal (Figure 2B) (Unwin 1989; Galzi et al. 1992). Two cysteines, separated by 13 amino acid residues, are found in the extracellular N-terminal domain of all subunits belonging to the cys-loop superfamily (Kao and Karlin 1986; Ortells and Lunt 1995). As well, in GluCls, glycine- and histamine-gated chloride channels, a second pair of cysteine residues, separated by 10 residues, is found before the first transmembrane domain (Vassilatis et al. 1997b; Zheng et al. 2002).

16


B.



Figure 2. Topological organization of cys-loop receptors. **A.** The five subunits that make up the receptor are arranged around a central pore, with the TM2 of each subunit lining the channel pore. **B.** Each subunit is made up of a large N-terminus, four transmembrane domains (TM1-4), an intracytoplasmic loop between TMs 3 and 4, and a short C terminus. Adapted from Cherubini and Conti (2001).

The nAchR is the most extensively characterized member of the ligand gated ion channel superfamily, and much of what is known about the agonist binding sites of the cys-loop receptors comes from studies of this receptor (Miyazawa et al. 1999; Sine 2002). In vertebrates, five distinct subunit types (α , β , γ , δ and ϵ) contribute to the formation of nAchRs (Karlin 1993). Two adjacent cysteine residues found in α subunits, but not in non- α subunits (β , $\gamma \epsilon$ or δ), are essential for acetylcholine binding. Both α and non- α subunits (or only α subunits in the case of the α 7 nAchR) contribute to ligand binding, and the acetylcholine binding sites occupy the boundary between individual subunits (Figure 3) (Changeux and Edelstein 1998). The binding site is formed by several amino acid residues located in distinct regions of the N-terminal extracellular domain known as 'loops' (Changeux and Edelstein 1998; Corringer et al. 2000). Loops A, B and C, termed the principle components, are contributed by the α subunits (Dennis et al. 1988; Galzi et al. 1990; Fu and Sine 1994), while loops D, E and F, the complementary components, are contributed by the neighboring subunit residues (Figure 3) (Czajkowski et al. 1993; Corringer et al. 1995; Martin et al. 1996; Chiara and Cohen 1997; Martin and Karlin 1997). Several residues from loops A, B and C are highly conserved in α subunits (Changeux and Edelstein 1998; Corringer et al. 2000). Opening of the channel occurs when binding of Ach initiates small rotations of the subunits, which then cause a conformational change of the transmembrane domains lining the channel pore, resulting in channel opening (Unwin 1995).



Figure 3. Schematic drawing of the muscle-type $(\alpha 1)_2\beta 1\gamma\delta$ nAchR. This receptor has two Ach-binding sites per receptor complex, corresponding to the two α subunits. The Ach binding sites are located at the interface between α and δ , and α and γ subunits. Loops A, B and C, the 'principal component', are contributed by the α subunits, and loops D, E and F, the 'complementary component', are contributed by the adjacent δ and γ subunits. Adapted from Corringer et al. (2000).

Until recently, knowledge of the three-dimensional structure of LGICs was based on electron microscopy studies on nAchR of *Torpedo californica* (Unwin 1993, 1995). These studies revealed the organization of the ion channel, while mutagenesis and affinity labeling studies were used to identify the location of the ligand binding sites. Recently, Smit et al. (2001) identified an acetylcholine binding protein (AchBP) in the snail *Lymnaea stagnalis* which is produced by glial cells and modulates cholinergic transmission by binding Ach. This protein has a high degree of homology with the extracellular domains of nAchR subunits. Its structure has been solved at high resolution, and provides detailed information on the arrangement of the ligand binding sites (Brejc et al. 2001). With a high resolution structural model of a receptor ligand binding domain now available, the determinants of ligand binding of members of the cys-loop superfamily can now be examined in a three dimensional context. Models for the extracellular domains of chick nAchR (α 7)₅ and the human GABA_A ($\alpha_2\beta_2\gamma$) receptor have been developed on the basis of the AchBP crystal structure (Cromer et al. 2002; Le Novere et al. 2002). These models show that residues implicated in ligand binding by photoaffinity labeling and mutagenesis experiments do indeed form the ligand binding pocket.

1.5.5 The GluCls

The GluCls are important targets for the MLs (Duce and Scott 1985). These channels are distinct from the excitatory glutamate-gated cation channels found in both vertebrates and invertebrates. They are found only in invertebrates, making them ideal drug targets (Duce and Scott 1985; Arena et al. 1992).

C. elegans is highly sensitive to MLs, and has been widely used as a model organism for studying the mode of action of these drugs (Kass et al. 1980; Schaeffer and Haines 1989; Arena et al. 1991; Cully and Paress 1991; Arena et al. 1992; Cully et al. 1994; Arena et al. 1995; Dent et al. 1997; Dent et al. 2000). Specific high-affinity avermectin binding sites have been identified in *C. elegans*, and the binding affinities of different avermectin analogs correlate well with their biological activity (Schaeffer and Haines 1989; Cully and Paress 1991). In experiments to identify the site and mechanism of action of avermectins, Arena et al. (1991) expressed *C. elegans* mRNA in *Xenopus laevis* oocytes, and demonstrated avermectin-sensitive (EC₅₀ = 90 nM), GABA-

insensitive chloride currents. Further experiments showed that the avermectin-sensitive currents were directly activated by glutamate (EC₅₀ = 350 μ M), and that avermectin potentiated the effects of glutamate (Arena et al. 1992). The first GluCl subunits to be cloned were the GluCl α 1 and GluCl β subunits of *C. elegans* (Cully et al. 1994). Other GluCl subunits were subsequently cloned from *C. elegans* (Dent et al. 1997; Laughton et al. 1997a; Vassilatis et al. 1997a; Dent et al. 2000), the parasitic nematodes *H. contortus* (Delany et al. 1998; Forrester et al. 1999; Jagannathan et al. 1999; Cheeseman et al. 2001), *A. suum* (Jagannathan et al. 1999), *D. immitis* and *O. volvulus* (Cully et al. 1996a), and the insect *D. melanogaster* (Cully et al. 1996b).

In *C. elegans*, five genes have been identified that encode GluCl subunits (Cully et al. 1994; Dent et al. 1997; Laughton et al. 1997a; Vassilatis et al. 1997a; Dent et al. 2000; Horoszok et al. 2001). Two of these genes, *avr-14* (*GluCla3*) and *avr-15* (*GluCla2*) are alternatively spliced to yield two different subunits (Dent et al. 1997; Laughton et al. 1997a; Dent et al. 2000). The two *avr-14* splice variants, GluCl α 3A and GluCl α 3B, share the same N-terminal extracellular domain but different membraneassociated domains. Interestingly, the membrane-associated domain of GluCl α 3B lies in the 3' untranslated region (UTR) of the GluCl α 3A cDNA, and may have arisen as a result of partial gene duplication (Laughton et al. 1997a). The *avr-15* gene also encodes two alternatively spliced channel subunits, GluCl α 2A and GluCl α 2B (Dent et al. 1997). The predicted protein sequences of the two subunits are identical, except at the N-terminal, where GluCl α 2A and 2B consist of 202 and 23 amino acids, respectively.

The properties of the different *C. elegans* GluCl subunits have been examined by expression in *Xenopus* oocytes. GluCl α 1 and GluCl β form homomeric channels with

distinct properties. The GluCl β channel responds to glutamate (EC₅₀ = 380 μ M) and is insensitive to ivermectin, while the GluCl α 1 channel responds to ivermectin (EC₅₀ = 140) nM), but shows no response to glutamate (Cully et al. 1994). The glutamate response activates rapidly, and is completely reversible. In contrast, the ivermectin response activates slowly, and is irreversible. Co-expression of GluCl α 1 and GluCl β results in the formation of heteromeric channels that are gated by both glutamate ($EC_{50} = 1.36$ mM) and ivermectin (EC₅₀ = 190 nM). Ivermectin also potentiates the effect of glutamate on the heteromeric channels, indicating that it acts as an allosteric modulator of the channel. The reversal potentials for the ivermectin- and glutamate-sensitive channels are consistent with these receptors being permeable to chloride ions. These channels are also insensitive to the excitatory glutamate receptor agonists NMDA, kainate, quisqualate, AMPA and aspartate. As well, they are insensitive to the chloride channel agonists glycine and GABA. Ibotenate, a glutamate agonist, activates the heteromeric receptor, and the glutamate and ivermectin responses of this channel are blocked by picrotoxin. The ability of several avermectin analogs to activate current in the heteromeric channel correlates with their nematocidal efficacy, suggesting that the effects of the avermectins are mediated through interaction with the GluCls (Cully et al. 1994; Arena et al. 1995).

Other *C. elegans* GluCl subunits have been examined and shown to have pharmacological properties similar to those of GluCl α 1 (Dent et al. 1997; Vassilatis et al. 1997a; Dent et al. 2000; Horoszok et al. 2001). Like GluCl α 1, the *avr-15* splice variants GluCl α 2A and 2B form ivermectin-sensitive channels (EC₅₀ = 108 nM for GluCl α 2B) when expressed in *Xenopus* oocytes (Dent et al. 1997; Vassilatis et al. 1997a). However, unlike GluCl α 1, the GluCl α 2A and 2B channels are also gated by glutamate (EC₅₀ = 2

mM and 208 μ M, respectively). Other *C. elegans* subunits which form ivermectinsensitive, glutamate gated channels include GluCl α 3B and GluCl α 4 (glutamate and ivermectin EC₅₀ = 1.9 mM and 400 nM, respectively) (Dent et al. 2000; Horoszok et al. 2001). Despite its inability to form glutamate-sensitive channels, GluCl α 1 possesses the determinants for glutamate binding on its N-terminal extracellular domain, since a chimera between the GluCl α 1 N-terminal domain and the GluCl β C-terminal domain forms glutamate-sensitive homomeric channels (Etter et al. 1996).

Co-expression of GluCl α 2B with GluCl β results in formation of heteromeric channels that are sensitive to both glutamate (EC₅₀ = 62 µM) and ivermectin (EC₅₀ = 103 nM) (Vassilatis et al. 1997a). Unlike the GluCl α 1 + GluCl β heteromeric channels which are less sensitive to glutamate than the GluCl β homomeric channels, the GluCl α 2B + GluCl β heteromeric channels are more sensitive to glutamate than the respective homomeric channels. It is likely that GluCl α 2B and GluCl β co-assemble *in vivo*, since they are both expressed in the pharyngeal muscle (Dent et al. 1997; Laughton et al. 1997b). The expression sites of GluCl α 1 and GluCl α 4 have not been reported. *avr-14*, which encodes GluCl α 3A and 3B, is expressed in extrapharyngeal neurons, and can inhibit pharyngeal pumping in response to ivermectin (Dent et al. 2000). This action is mediated via gap junctions between pharyngeal and extrapharyngeal neurons.

Like *C. elegans*, the parasitic nematode *H. contortus* is highly sensitive to the effects of the MLs, and the GluCls are also thought to be a major target (Geary et al. 1993; Cheeseman et al. 2001; Forrester et al. 2002; Forrester et al. 2003). Glutamate and ivermectin binding sites have been identified in *H. contortus* crude membrane preparations (Rohrer et al. 1994; Paiement et al. 1999; Hejmadi et al. 2000). Three genes

have been identified in H. contortus that encode GluCl subunits (Delany et al. 1998; Forrester et al. 1999; Jagannathan et al. 1999). Two of these genes, HcGluCla and HcGluCla3, are alternatively spliced to yield two subunits each. Phylogenetic analysis shows that HcGluCla is not orthologous to any of the GluCl subunits that have so far been identified in nematodes (Forrester et al. 1999). However, it is most closely related HcGluCl β and the HcGluCl $\alpha\beta$ splice variants, to the GluCl α 3-type subunits. HcGluCla3A and 3B, are orthologs of C. elegans GluClb, GluCla3A and 3B, respectively (Delany et al. 1998; Jagannathan et al. 1999). When expressed in COS-7 cells, HcGluCla and HcGluCla3B bind ivermectin with high affinity, similar to what is observed in whole membrane preparations. This suggests that GluCls represent the high affinity ivermectin binding site. While direct glutamate binding was not demonstrated in GluCls expressed in COS-7 cells, it was found to potentiate ivermectin and moxidectin binding (Forrester et al. 2002). Recently, Forrester et al. (2003) have demonstrated that HcGluCla forms functional homomeric receptors when expressed in Xenopus oocytes. These receptors are activated by glutamate (EC₅₀ = 8.4 μ M), ivermectin (EC₅₀ = 131 nM) and moxidectin. As in C. elegans, therefore, the GluCls of this parasitic nematode are a target for MLs.

The expression patterns of the four *H. contortus* GluCl subunits have been reported. HcGluCla is expressed in motor neuron commissures, and the receptor it forms is likely to mediate the effects of the MLs on locomotion (Portillo et al. 2003). HcGluCl β , which is orthologous to *C. elegans* GluCl β , is also expressed in motor neuron commissures, and appears to co-localize with HcGluCla, suggesting that in the motor neuron sinnervating the body-wall muscle of *H. contortus*, the GluCl receptors contain

HcGluCla and HcGluCl β (Delany et al. 1998; Portillo et al. 2003). The splice variants HcGluCl α 3A and 3B are also expressed in the motor neuron commissures (Jagannathan et al. 1999; Portillo et al. 2003). In other regions, these two splice variants have unique expression patterns (Portillo et al. 2003). HcGluCl α 3A is found in a pair of lateral neurons in the head of the worm, which are most likely amphidial neurons. HcGluCl α 3B is expressed in nerve cords, as well as three cell bodies on the pharynx, which are most likely pharyngeal neurons M1 and M2. This suggests that pharyngeal receptors containing HcGluCl α 3B are involved in mediating ivermectin's inhibitory effects on pharyngeal pumping. The expression patterns of orthologous subunits of *H. contortus* and *C. elegans* therefore appear to be different, even though the two are closely related phylogenetically (Blaxter et al. 1998). Nonetheless, MLs have similar effects in both species, and the differences in subunit distribution may reflect their adaptation to different life styles (Portillo et al. 2003).

1.5.6 Ivermectin's Interaction with Mammalian Host Targets

Apart from interacting with invertebrate GluCls and GABA-Cls, the MLs have also been shown to interact with vertebrate ion channel receptors. While ivermectin does not directly activate current in heterologously expressed chick or human α 7 nAch receptors, it potentiates the response to Ach in these receptors, causing a 20-fold increase in affinity to Ach (Krause et al. 1998). Ivermectin has also been shown to directly activate mammalian recombinant GABA_A receptors and to potentiate GABA-induced chloride currents in cultured mammalian hippocampal neurons (Krusek and Zemkova 1994; Adelsberger et al. 2000). In the mouse seizure model, ivermectin acts as an anticonvulsant, and this effect is mediated through GABAA receptors (Dawson et al. 2000). As well, ivermectin has a potentiating effect on ATP-gated P2X₄ cation channels (Khakh et al. 1999). Shan et al. (2001) have also demonstrated that ivermectin directly activates glycine-gated chloride channel receptors, and also potentiates the response to glycine. In these receptors, the ivermectin binding site is thought to be different from the glycine binding site, since mutations of known glycine binding sites have little effect on ivermectin sensitivity. Ivermectin therefore interacts with a wide range of host receptors. However, Schaeffer and Haines (1989) showed that the affinity for vertebrate receptors is 100-fold lower than the affinity for nematode receptors. This, and the fact that MLs do not readily cross the blood brain barrier, may account for the low toxicity of these compounds to the mammalian host (Krusek and Zemkova 1994). However, disruption of the mdrla gene, which encodes a drug transporting P-glycoprotein (P-gp) in the blood brain barrier, can interfere with the protective capacity of the blood-brain barrier (Schinkel et al. 1994). P-gp acts as an efflux pump and prevents the accumulation of a variety of chemicals, including ivermectin, in the brain. Mice that are homozygous for a disruption of the mdr1a gene are 50- to 100-fold more sensitive to ivermectin than wildtype mice (Schinkel et al. 1994). Some collie dogs are also extremely sensitive to the effects of ivermectin (Paul et al. 1987). This phenotype is caused by a deletion mutation in the mdr1 gene (Mealey et al. 2001; Roulet et al. 2003). The 4 bp deletion in the fourth coding exon causes a frameshift and introduces a premature stop codon 16 codons downstream of the mutation (Roulet et al. 2003). The mutation is recessive, since only dogs that are homozygous for the deletion mutation show the ivermectin-sensitive phenotype, while dogs that are heterozygous do not (Mealey et al. 2001). In normal

individuals, therefore, functional P-gp-encoding genes limit ivermectin's access to the central nervous system, where the ivermectin-sensitive mammalian receptors are restricted.

1.6 Anthelmintic Resistance

The currently available broad spectrum anthelmintics have provided farmers with safe, easy to use, and effective means of controlling helminth infections in domestic animals (Waller 1994). However, the heavy reliance on anthelmintics has led to the development of resistance, which limits the usefulness of these drugs in domestic animals (Coles et al. 1994; Le Jambre et al. 2000) Although anthelmintic resistance has been reported in horses, cattle, goats and pigs, it is most prevalent in nematode parasites of sheep, and the situation is worst in the sheep-raising countries of the Southern Hemisphere, namely South Africa, New Zealand and Australia (Coles et al. 1994; Waller 1994). While intensive use of anthelmintics is a major factor in resistance development in these countries, the use of inferior generic products has been cited as an important factor in increasing the selection pressure for resistance in poorer developing countries (Waller 1994, 1997).

Worms in natural populations show considerable variability in response to anthelmintics, with some being highly sensitive while others are highly resistant (Le Jambre 1993; Sangster 1996). Alleles conferring resistance are present in an anthelmintic-naïve population, albeit at low frequencies (Kelly et al. 1978; Prichard et al. 1980; Jackson 1993). At this stage, most resistant alleles occur in heterozygous (RS) worms, and homozygous resistant worms (RR) are rare (Barnes et al. 1995). Following

treatment with anthelmintic at the recommended dose rate, susceptible parasites are eliminated, while the small proportion of worms carrying resistance alleles survive and contribute their gene pool to the next generation (Prichard et al. 1980; Martin 1987; Waller 1994). Over time, and with continued selection pressure, the frequency of resistance alleles in the population increases, eventually manifesting as treatment failure. The rate at which resistance develops will be determined by the initial frequency of resistance alleles in the worm population, the degree of dominance and the number of genes involved (Martin 1987; Barnes et al. 1995; Le Jambre et al. 2000). Resistance will evolve faster if the initial frequency of resistance- associated alleles is high, and if it is determined by a single gene. As well, resistance will also develop more rapidly when it is dominant, since homozygous resistant (RR) and heterozygous (RS) worms are able to survive treatment.

Drug resistance is a growing problem in the cattle industry. In recent years, resistance to all three broad spectrum anthelmintic classes has been reported in several countries, including Australia, New Zealand, Argentina, Brazil, the UK and Belgium. Benzimidazole resistance has been found in *Trichostrongylus axei* (Eagleson and Bowie 1986), *C. oncophora* (Jackson et al. 1987; McKenna 1991), *Cooperia punctata* (Mejia et al. 2003), *Ostertagia ostertagi* (Hosking et al. 1996; Mejia et al. 2003), and *Haemonchus* species (McKenna 1991; Mejia et al. 2003). Levamisole and morantel resistance have been reported in *O. ostertagi* (Geerts 1986; Geerts et al. 1987; Borgsteede 1988, 1991) and *Haemonchus placei* (Yadav and Verma 1997), while ivermectin resistance has been reported in *Cooperia* species (Vermunt et al. 1996; Coles et al. 1998; Stafford and Coles 1999; Anziani et al. 2001; Coles et al. 2001; Familton et al. 2001; Fiel et al. 2001).

Multiple resistance to the benzimidazoles and avermectins in *Cooperia* species has also been reported (Vermunt et al. 1995; Mejia et al. 2003).

Anthelmintic resistance has been slower to emerge in cattle than in small ruminants (Waller 1994; Coles 2002a, b). Several factors may account for this. With sufficient exposure, cattle build up a good level of immunity to helminth infections, therefore, as adult animals, they often do not require treatment (Reinemeyer 1995; Coles 2002a). In cattle enterprises, therefore, the practice has been to treat only first-year calves, leaving older animals untreated, so the selection pressure for resistance has been low (Stafford and Coles 1999; Vercruysse and Dorny 1999; Coles 2002a). As well, unlike sheep fecal pellets which disintegrate rapidly, cattle dung pats provide a safe environment in which the free-living stages can develop. Parasites which survive on pasture and are unexposed to anthelmintic will serve to dilute any parasites that survive treatment, thereby slowing down the development of resistance (Armour et al. 1987; Coles 2002a, b). However, the availability of easy-to-use drug formulations, and the treatment of older animals to increase performance will most likely lead to an increase in the incidence of anthelmintic resistance in cattle (Reinemeyer 1995; Vercruysse and Claerebout 2001; Coles 2002a; Morris et al. 2002).

In the field, anthelmintic resistance is usually first suspected when animals that have been treated with the recommended drug dose fail to respond to treatment (Prichard et al. 1980; Taylor and Hunt 1989; Fraser 1991; Coles et al. 1998; Sangster 2001). For broad-spectrum claims, anthelmintics are marketed at dose rates determined by efficacy against the least sensitive (dose-limiting) species (Sangster 1996). The recommended dose rates are therefore much higher than doses required to eliminate the more sensitive

target species. Ivermectin, for example, is >95% effective against susceptible *H. contortus* when administered orally at 0.02 mg/kg body weight, which is ten times less than the recommended dose rate (Shoop et al. 1995). At the early stages of resistance, therefore, the drug will remain effective, since the use level is well above the 95% efficacy level against this species (Sangster 1999). Consequently, by the time resistance becomes apparent, a 10-fold decrease in efficacy will have occurred (Shoop et al. 1995). However, for the dose-limiting species, a smaller decrease in sensitivity will manifest as treatment failure, since this species requires the full dose for complete efficacy (Shoop et al. 1995).

The relative fitness of the resistance genes is an important factor in determining the maintenance of resistance in a parasite population, once it has been selected for (Prichard et al. 1980; Martin 1987). For the benzimidazole class of anthelmintics, several studies have been conducted to determine whether reversion from resistance to susceptibility is likely to occur following removal of the selection pressure. Simpkin and Coles (1978) reported slight reversion to susceptibility after two generations in the absence of drug challenge. However, other studies found that once resistance had developed, reversion to susceptibility did not occur following removal of the anthelmintic selection pressure (Hall et al. 1982; Le Jambre et al. 1982; Herd et al. 1984; Martin 1987; Martin et al. 1988; Jackson and Coop 2000). As well, counter-selecting with a different anthelmintic, levamisole, did not result in reversion to benzimidazole susceptibility (Martin et al. 1988; Borgsteede and Duyn 1989). For reversion to occur, the resistance alleles would have to confer a selective disadvantage in the absence of treatment (Le Jambre et al. 1982). Elard et al. (1998) examined several fitness-related traits in benzimidazole-susceptible and -resistant *Teladorsagia circumcincta*, and found no significant difference between the two groups, suggesting that benzimidazole resistance does not confer a fitness disadvantage, and that once it occurs, it is irreversible. Whether this is the case with ivermectin resistance is not known. The reversion reported by Simpkin and Coles (1978) may have been due to dilution of the resistant parasites following ingestion of susceptible larvae present on pasture (Hall et al. 1982). While the deliberate introduction of susceptible parasites would be an impractical means of slowing down the spread of resistance, reducing the frequency of treatment would reduce the selection pressure for resistance, and prolong the shelf-life of existing anthelmintics (Prichard et al. 1980; Hall et al. 1982; van Wyk and van Schalkwyk 1990; Sangster 1999).

1.7 Benzimidazole Resistance Mechanisms

In nematodes, studies into the mechanisms of benzimidazole resistance have focused on the role of beta-tubulin, which is known to be the target for this class of anthelmintics (Lacey 1990). In *C. elegans*, benzimidazole resistance is associated with the deletion of a benzimidazole-susceptible beta-tubulin gene (Driscoll et al. 1989). Apart from resistance to benzimidazoles, worms lacking this gene show the same phenotype as susceptible worms, suggesting that this gene is not essential, and can be replaced by other members of the beta-tubulin family (Driscoll et al. 1989). In *H. contortus*, restriction fragment length polymorphism (RFLP) analysis on individual worms showed a reduction in diversity of beta-tubulin isotype 1 and 2 genes with benzimidazole resistance (Roos et al. 1990; Kwa et al. 1993b; Kwa et al. 1993a; Beech et al. 1994; Lubega et al. 1994).

Selection of isotype 1 and 2 variants occurs at lower degrees of resistance, while at higher degrees of resistance, individuals carrying isotype 2 genes are eliminated from the population (Kwa et al. 1993b). Analysis of beta-tubulin isotype 1 sequences from susceptible and resistant worms revealed the presence of a phenylalanine-to-tyrosine (Phe-Tyr) mutation at position 200 in all resistant worms examined (Kwa et al. 1994). Other studies found that the same Phe-Tyr mutation at position 200 conferred benzimidazole resistance in fungi (Jung et al. 1992; Koenraadt et al. 1992). The role of this mutation in benzimidazole resistance in H. contortus was confirmed in a heterologous transformation study (Kwa et al. 1995). The H. contortus beta-tubulin isotype 1 gene with a Phe-Tyr mutation at position 200 did not alter the drug phenotype of the benzimidazole-susceptible wild-type C. elegans or the benzimidazole-resistant In contrast, transformation with the H. contortus benzimidazole*ben-1* mutant. susceptible gene (with Phe at position 200) conferred benzimidazole sensitivity in the ben-1 mutant, indicating the functional importance of the codon 200 position in benzimidazole sensitivity and resistance.

Identification of the genetic determinant of benzimidazole resistance in nematodes led to the development of an allele-specific polymerase chain reaction (PCR) test for genotyping individual adult worms based on the amino acid at position 200 (Elard et al. 1999). The test has been adapted for use on L3 larval stages of *T. circumcincta*, *T. colubriformis* and *H. contortus*, which can easily be obtained from fecal cultures, as opposed to sacrificing animals to obtain adult worms (Silvestre and Humbert 2000). These molecular tests are useful for detecting the emergence of resistance, when the frequency of resistant worms in the population is low (Elard et al. 1999).

The mutation of residue 200 of beta-tubulin isotype 1 may not be the only determinant of benzimidazole resistance in nematodes. A Phe-Tyr mutation on isotype 2 has also been found to confer benzimidazole resistance in *H. contortus* (Prichard 2001). As well, mutagenesis studies show that a Phe-Tyr mutation at position 167 reduces binding of benzimidazole to heterologously expressed *H. contortus* beta-tubulin (Prichard et al. 2000). This mutation has also been identified in a benzimidazole-resistant cyathostome population (Kaplan et al. 2000). These studies suggest that a diagnostic test that only identifies the mutation at position 200 of isotype 1 may give false negative results (Prichard 2001). However, the position 167 mutation appears to be rare in benzimidazole-resistant field isolates of *H. contortus*, *T. colubriformis* and *T. circumcincta* (Silvestre and Cabaret 2002). These authors found the position 200 mutation in all benzimidazole-resistant worms examined, suggesting that this is the major determinant of benzimidazole resistance.

1.8 Macrocyclic Lactone Resistance Mechanisms

While the genetic basis of benzimidazole resistance is well established, the mechanisms underlying ivermectin resistance are less well understood. Resistance to the MLs is thought to involve alterations in the target sites, as well as a reduction in the drug concentration at the site of action. *In vitro* assays have been developed to investigate differences between avermectin-susceptible and –resistant parasite isolates (Gill et al. 1998). Avermectin-resistant isolates of *H. contortus* are less sensitive to the effects of avermectins on feeding, larval development and motility (Gill et al. 1998). However, not all avermectin-resistant isolates of *H. contortus* show the same phenotype.

While some resistant isolates are resistant to avermectin's inhibition of larval development and motility, other resistant isolates show no reduction in sensitivity to the effects of avermectins (Gill et al. 1998). These differences may reflect the differences in selection pressures used to produce the different isolates. As well, the differences may reflect the relative contributions of different genes which contribute to the resistance phenotype in different geographic worm populations.

In other studies, Rohrer et al. (1994) compared the binding of ivermectin to membrane preparations from ivermectin-susceptible and -resistant H. contortus L3. The authors found no difference in binding affinity or density between the two strains, suggesting that ivermectin resistance was not due to target site alteration. However, Paiement et al. (1999) demonstrated that ivermectin resistance was associated with an increase in low-affinity glutamate binding sites. These results were confirmed by Hejmadi et al. (2000). Ivermectin was also found to inhibit glutamate binding in membrane preparations from ivermectin-susceptible H. contortus, but had no effect on membrane preparations from ivermectin-resistant worms (Paiement et al. 1999). These authors concluded that ivermectin resistance was associated with changes at a glutamate binding site. Indeed, genetic variability analysis of ivermectin-susceptible and -resistant strains of H. contortus revealed that there was selection on a GluCl gene, suggesting that this gene is involved in ivermectin resistance (Blackhall et al. 1998a). The involvement of GluCls in ivermectin resistance has also been demonstrated in C. elegans, where simultaneous mutation of three GluCl genes, glc1, avr-14 and avr-15, conferred highlevel (4,000-fold) resistance (Dent et al. 2000). In contrast, simultaneous mutation of two of these genes, avr-14 and avr-15, conferred modest (~14-fold) resistance.

In *H. contortus*, the HG1 gene, which encodes a GABA receptor, appears to be involved in ivermectin resistance. Genetic variability analysis of this gene revealed significant differences in allele frequencies between ivermectin-susceptible and –resistant strains of *H. contortus* (Blackhall et al. 2003). One allele, e, increased substantially in frequency in the resistant strain relative to the ivermectin-susceptible strain. The full-length cDNA sequence of this resistance-associated allele was determined, as was the sequence of the most common allele in the susceptible strain, allele A (Feng et al. 2002). These two alleles differ at four amino acid positions, two of which are located in the N-terminal domain. The functional properties of these two alleles were examined by co-expressing them in *Xenopus* oocytes with a *C. elegans* GABA- β subunit (Feng et al. 2002). While ivermectin potentiated the GABA response of the GABA- β /Allele A receptor, it attenuated the response of the GABA- β /allele e receptor. Mutations in the resistance-associated allele (e) therefore seem to affect ivermectin's interaction with the receptor.

Apart from target site alteration, ivermectin resistance may also be mediated by mechanisms that alter the drug concentration at the sites of action. Several studies have provided evidence for the involvement of P-gp in ML resistance in nematodes. A P-gp homolog isolated from *H. contortus* is upregulated in ivermectin-resistant strains of *H. contortus* (Xu et al. 1998). Genetic variability analysis of this same gene in ivermectin-susceptible and –resistant strains of *H. contortus* showed a significant difference in allele frequencies between the two strains (Blackhall et al. 1998b). As well, verapamil, a multidrug-resistance-reversing agent, significantly increased the efficacy of ivermectin against the ivermectin-resistant strain in jirds (Molento and Prichard 1999). These

studies indicate that P-gp-mediated drug efflux plays a role in ML resistance in H. contortus.

1.9 Anthelmintic Resistance Detection Methods

Once resistance has developed to a particular drug, it is no longer useful in protecting animals against parasite infections, and change to a drug with a different mode of action is necessary (McKenna 1990; Prichard 1990). In order to delay the onset of resistance, early detection through regular monitoring is important (Jackson and Coop 2000). Unfortunately, testing for resistance is often carried out when drug inefficacy becomes apparent, by which time the frequency of resistance alleles in the population is high. The presence of anthelmintic resistance can be determined using *in vivo* and *in vitro* assays. The assays include the fecal egg count reduction test (FECRT), the egg hatch assay (EHA), the larval paralysis/motility assays, and the larval development assay (Taylor et al. 2002).

Of the tests available, the FECRT is the one that is routinely used for detecting resistance in the field (Waller 1994, 1997). This test provides an estimate of anthelmintic efficacy by comparing the fecal egg counts of a group of animals before and after treatment, and can be used to evaluate the efficacy of any anthelmintic (McKenna 1990). Guidelines have been recommended by the World Association for the Advancement of Veterinary Parasitology (WAAVP) to standardize the efficacy evaluation process, so that results obtained in different studies can be compared (Coles et al. 1992). Based on the guidelines, resistance is suspected if the percentage reduction in egg count is less than

95%, since the efficacy of modern broad-spectrum anthelmintics is greater than 95% (McKenna 1990).

The FECRT is easy to perform, and does not require sophisticated equipment or highly trained personnel (Waller 1994). It does however, have limitations. Several nematode genera that infect ruminants produce strongyle-type eggs, including *Haemonchus, Ostertagia/Teladorsagia, Trichostrongylus* and *Cooperia* (Fraser 1991). These strongyle-type eggs cannot easily be identified to genus level (McKenna 1996). In a mixed infection, only one species may be resistant, and an egg count cannot determine which species is involved (Prichard et al. 1980). As well, the occurrence of resistance in low-fecundity genera may be masked by reduction in egg counts of susceptible genera that have a high biotic potential (McKenna 1997). Larval cultures of feces collected before and after treatment are therefore recommended for identification of resistant genera (McKenna 1996, 1997).

The major drawback of the FECRT is its lack of sensitivity (Martin et al. 1989). It can only detect resistance when the proportion of resistant worms in the population is at least 25%. Sensitive tests would aid in the early detection of resistance. Such tests have been developed for the benzimidazole class of anthelmintics (Elard et al. 1999; Silvestre and Humbert 2000). No such test exists for the ML anthelmintics. Such a test would allow not just the early detection of ML resistance, but also a survey of its prevalence (Coles et al. 2001). ML resistance in cattle nematodes is still not widespread, and elucidating mechanisms of ML resistance will shed light on possible markers of ML resistance that could be used in designing molecular tests. This thesis focuses on examining several genes that have been implicated in ML resistance.

References

- Adelsberger H., Lepier A. and Dudel J. (2000) Activation of rat recombinant alpha(1)beta(2)gamma(2S) GABA(A) receptor by the insecticide ivermectin. Eur J Pharmacol 394, 163-170.
- Anziani O. S., Zimmermann G., Guglielmone A. A., Vazquez R. and Suarez V. (2001) Avermectin resistance in Cooperia pectinata in cattle in Argentina. *Vet Rec* 149, 58-59.
- Arena J. P., Liu K. K., Paress P. S. and Cully D. F. (1991) Avermectin-sensitive chloride currents induced by Caenorhabditis elegans RNA in Xenopus oocytes. *Mol Pharmacol* **40**, 368-374.
- Arena J. P., Liu K. K., Paress P. S., Schaeffer J. M. and Cully D. F. (1992) Expression of a glutamate-activated chloride current in Xenopus oocytes injected with Caenorhabditis elegans RNA: evidence for modulation by avermectin. *Brain Res Mol Brain Res* 15, 339-348.
- Arena J. P., Liu K. K., Paress P. S., Frazier E. G., Cully D. F., Mrozik H. and Schaeffer J. M. (1995) The mechanism of action of avermectins in Caenorhabditis elegans: correlation between activation of glutamate-sensitive chloride current, membrane binding, and biological activity. J Parasitol 81, 286-294.
- Armour J. (1989) The influence of host immunity on the epidemiology of trichostrongyle infections in cattle. *Vet Parasitol* **32**, 5-19.
- Armour J., Bairden K., Holmes P. H., Parkins J. J., Ploeger H., Salman S. K. and McWilliam P. N. (1987) Pathophysiological and parasitological studies on Cooperia oncophora infections in calves. *Res Vet Sci* 42, 373-381.
- Avery L. and Horvitz H. R. (1990) Effects of starvation and neuroactive drugs on feeding in Caenorhabditis elegans. *J Exp Zool* 253, 263-270.
- Awadzi K., Dadzie K. Y., Shulz-Key H., Haddock D. R., Gilles H. M. and Aziz M. A. (1985) The chemotherapy of onchocerciasis X. An assessment of four single dose treatment regimes of MK-933 (ivermectin) in human onchocerciasis. Ann Trop Med Parasitol 79, 63-78.
- Baraldi E., Mari M., Chierici E., Pondrelli M., Bertolini P. and Pratella G. C. (2003) Studies on thiabendazole resistance of Penicillium expansum of pears: pathogenic fitness and genetic characterization. *Plant Pathol* 52, 362-370.
- Barnes E. H., Dobson R. J. and Barger I. A. (1995) Worm Control and Anthelmintic Resistance -Adventures with a Model. *Parasitol Today* 11, 56-63.
- Bechade C., Sur C. and Triller A. (1994) The inhibitory neuronal glycine receptor. *Bioessays* 16, 735-744.

- Beech R. N., Prichard R. K. and Scott M. E. (1994) Genetic variability of the beta-tubulin genes in benzimidazole-susceptible and -resistant strains of Haemonchus contortus. *Genetics* 138, 103-110.
- Bisset S. A., Marshall E. D. and Morrison L. (1987a) Economics of a dry-cow anthelmintic drenching programme for dairy cows in New Zealand. Part 1. Overall response in 47 herds. *Vet Parasitol* **26**, 107-118.
- Bisset S. A., Marshall E. D. and Morrison L. (1987b) Economics of a dry-cow anthelmintic drenching programme for dairy cows in New Zealand. Part 2. Influence of management factors and other herd characteristics on the level of response. *Vet Parasitol* **26**, 119-129.
- Bisset S. A., Brunsdon R. V. and Forbes S. (1990) Efficacy of a Topical Formulation of Ivermectin against Naturally Acquired Gastrointestinal Nematodes in Weaner Cattle. *New Zeal Vet J* 38, 4-6.
- Blackhall W. J., Prichard R. K. and Beech R. N. (2003) Selection at a gamma-aminobutyric acid receptor gene in Haemonchus contortus resistant to avermeetins/milbemycins. *Mol Biochem Parasitol* 131, 137-145.
- Blackhall W. J., Pouliot J. F., Prichard R. K. and Beech R. N. (1998a) Haemonchus contortus: selection at a glutamate-gated chloride channel gene in ivermectin- and moxidectin-selected strains. *Exp Parasitol* **90**, 42-48.
- Blackhall W. J., Liu H. Y., Xu M., Prichard R. K. and Beech R. N. (1998b) Selection at a Pglycoprotein gene in ivermectin- and moxidectin-selected strains of Haemonchus contortus. *Mol Biochem Parasitol* 95, 193-201.
- Blaxter M. L., De Ley P., Garey J. R., Liu L. X., Scheldeman P., Vierstraete A., Vanfleteren J. R., Mackey L. Y., Dorris M., Frisse L. M., Vida J. T. and Thomas W. K. (1998) A molecular evolutionary framework for the phylum Nematoda. *Nature* 392, 71-75.
- Bogan J. A. and McKellar Q. A. (1988) The pharmacodynamics of ivermectin in sheep and cattle. J Vet Pharmacol Ther 11, 260-268.
- Bogan J. A., McKellar Q. A., Mitchell E. S. and Scott E. W. (1988) Efficacy of ivermectin against Cooperia curticei infection in sheep. Am J Vet Res 49, 99-100.
- Borgers M. and De Nollin S. (1975) Ultrastructural changes in Ascaris suum intestine after mebendazole treatment in vivo. *J Parasitol* **61**, 110-122.
- Borgsteede F. H. (1988) The difference between two strains of Ostertagia ostertagi in resistance to morantel tartrate. *Int J Parasitol* 18, 499-502.
- Borgsteede F. H. (1991) Further studies with a strain of Ostertagia ostertagi resistant to morantel tartrate. Int J Parasitol 21, 867-870.

- Borgsteede F. H. and Duyn S. P. (1989) Lack of reversion of a benzimidazole resistant strain of Haemonchus contortus after six years of levamisole usage. *Res Vet Sci* 47, 270-272.
- Bottjer K. P. and Bone L. W. (1985) Trichostrongylus colubriformis: effect of anthelmintics on ingestion and oviposition. *Int J Parasitol* 15, 501-503.
- Bowman D. D., Lynn R. C., Eberhard M. L. and Georgi J. R. (2003) *Georgis' parasitology for* veterinarians, 8th Edition, pp ix, 422. Saunders, St. Louis, Mo.
- Brejc K., van Dijk W. J., Klaassen R. V., Schuurmans M., van Der Oost J., Smit A. B. and Sixma
 T. K. (2001) Crystal structure of an ACh-binding protein reveals the ligand-binding domain of nicotinic receptors. *Nature* 411, 269-276.
- Brisson A. and Unwin P. N. (1985) Quaternary structure of the acetylcholine receptor. *Nature* **315**, 474-477.
- Brown K. R., Ricci F. M. and Ottesen E. A. (2000) Ivermectin: effectiveness in lymphatic filariasis. *Parasitology* **121 Suppl**, S133-146.
- Burkhart C. A., Kavallaris M. and Band Horwitz S. (2001) The role of beta-tubulin isotypes in resistance to antimitotic drugs. *Biochim Biophys Acta* 1471, O1-9.
- Campbell W. C. (1989) Ivermectin and abamectin, pp xviii, 363. Springer-Verlag, New York.
- Campbell W. C. (1990) Benzimidazoles Veterinary Uses. Parasitol Today 6, 130-133.
- Campbell W. C. (1991) Ivermectin as an antiparasitic agent for use in humans. Annu Rev Microbiol 45, 445-474.
- Campbell W. C., Fisher M. H., Stapley E. O., Albers-Schonberg G. and Jacob T. A. (1983) Ivermectin: a potent new antiparasitic agent. *Science* 221, 823-828.
- Chabala J. C., Mrozik H., Tolman R. L., Eskola P., Lusi A., Peterson L. H., Woods M. F., Fisher M. H., Campbell W. C., Egerton J. R. and Ostlind D. A. (1980) Ivermectin, a new broadspectrum antiparasitic agent. *J Med Chem* 23, 1134-1136.
- Chang Y., Wang R., Barot S. and Weiss D. S. (1996) Stoichiometry of a recombinant GABAA receptor. *J Neurosci* 16, 5415-5424.
- Changeux J. P. and Edelstein S. J. (1998) Allosteric receptors after 30 years. Neuron 21, 959-980.
- Cheeseman C. L., Delany N. S., Woods D. J. and Wolstenholme A. J. (2001) High-affinity ivermectin binding to recombinant subunits of the Haemonchus contortus glutamate-gated chloride channel. *Mol Biochem Parasitol* **114**, 161-168.
- Cherubini E. and Conti F. (2001) Generating diversity at GABAergic synapses. *Trends Neurosci* 24, 155-162.

- Chiara D. C. and Cohen J. B. (1997) Identification of amino acids contributing to high and low affinity d-tubocurarine sites in the Torpedo nicotinic acetylcholine receptor. *J Biol Chem* 272, 32940-32950.
- Claerebout E. (2002) The effects of chemoprophylaxis on acquired immunity against gastrointestinal nematodes in cattle. Verh K Acad Geneeskd Belg 64, 137-149.
- Claerebout E., Dorny P., Vercruysse J., Agneessens J. and Demeulenaere D. (1998) Effects of preventive anthelmintic treatment on acquired resistance to gastrointestinal nematodes in naturally infected cattle. *Vet Parasitol* **76**, 287-303.
- Claerebout E., Dorny P., Agneessens J., Demeulenaere D. and Vercruysse J. (1999) The effect of first season chemoprophylaxis in calves on second season pasture contamination and acquired resistance and resilience to gastrointestinal nematodes. *Vet Parasitol* **80**, 289-301.
- Coles G. C. (2002a) Cattle nematodes resistant to anthelmintics: why so few cases? Vet Res 33, 481-489.
- Coles G. C. (2002b) Sustainable use of anthelmintics in grazing animals. Vet Rec 151, 165-169.
- Coles G. C., Borgsteede F. H. M. and Geerts S. (1994) Anthelmintic-Resistant Nematodes in the Eu. *Parasitol Today* **10**, 288-290.
- Coles G. C., Stafford K. A. and MacKay P. H. (1998) Ivermectin-resistant Cooperia species from calves on a farm in Somerset. Vet Rec 142, 255-256.
- Coles G. C., Watson C. L. and Anziani O. S. (2001) Ivermectin-resistant Cooperia in cattle. Vet Rec 148, 283-284.
- Coles G. C., Bauer C., Borgsteede F. H., Geerts S., Klei T. R., Taylor M. A. and Waller P. J. (1992) World Association for the Advancement of Veterinary Parasitology (W.A.A.V.P.) methods for the detection of anthelmintic resistance in nematodes of veterinary importance. *Vet Parasitol* 44, 35-44.
- Conder D. R., Jones R. M. and Bliss D. H. (1983) The use of the morantel sustained release bolus in second-season grazing cattle. *Vet Parasitol* **12**, 307-312.
- Cook G. C. (1990) Use of Benzimidazole Chemotherapy in Human Helminthiases Indications and Efficacy. *Parasitol Today* 6, 133-136.
- Coop R. L., Sykes A. R. and Angus K. W. (1979) Pathogenicity of Daily Intakes of Cooperia-Oncophora Larvae in Growing Calves. *Veterinary Parasitology* 5, 261-269.
- Corringer P. J., Le Novere N. and Changeux J. P. (2000) Nicotinic receptors at the amino acid level. Annu Rev Pharmacol Toxicol 40, 431-458.



- Corringer P. J., Galzi J. L., Eisele J. L., Bertrand S., Changeux J. P. and Bertrand D. (1995) Identification of a new component of the agonist binding site of the nicotinic alpha 7 homooligomeric receptor. *J Biol Chem* 270, 11749-11752.
- Corwin R. M. (1997) Economics of gastrointestinal parasitism of cattle. Vet Parasitol 72, 451-457; discussion 457-460.
- Cromer B. A., Morton C. J. and Parker M. W. (2002) Anxiety over GABA(A) receptor structure relieved by AChBP. *Trends Biochem Sci* 27, 280-287.
- Cull-Candy S. G. (1976) Two types of extrajunctional L-glutamate receptors in locust muscle fibres. J Physiol 255, 449-464.
- Cull-Candy S. G. and Usherwood P. N. (1973) Two populations of L-glutamate receptors on locust muscle fibres. *Nat New Biol* 246, 62-64.
- Cully D. F. and Paress P. S. (1991) Solubilization and characterization of a high affinity ivermectin binding site from Caenorhabditis elegans. *Mol Pharmacol* 40, 326-332.
- Cully D. F., Wilkinson H., Vassilatis D. K., Etter A. and Arena J. P. (1996a) Molecular biology and electrophysiology of glutamate-gated chloride channels of invertebrates. *Parasitology* 113 Suppl, S191-200.
- Cully D. F., Paress P. S., Liu K. K., Schaeffer J. M. and Arena J. P. (1996b) Identification of a Drosophila melanogaster glutamate-gated chloride channel sensitive to the antiparasitic agent avermectin. J Biol Chem 271, 20187-20191.
- Cully D. F., Vassilatis D. K., Liu K. K., Paress P. S., Van der Ploeg L. H., Schaeffer J. M. and Arena J. P. (1994) Cloning of an avermectin-sensitive glutamate-gated chloride channel from Caenorhabditis elegans. *Nature* 371, 707-711.
- Czajkowski C., Kaufmann C. and Karlin A. (1993) Negatively charged amino acid residues in the nicotinic receptor delta subunit that contribute to the binding of acetylcholine. *Proc Natl Acad Sci U S A* **90**, 6285-6289.
- Davidse L. C. (1973) Antimitotic Activity of Methyl Benzimidazol-2-Yl Carbamate (Mbc) in Aspergillus-Nidulans. *Pestic Biochem Phys* 3, 317-325.
- Dawson G. R., Wafford K. A., Smith A., Marshall G. R., Bayley P. J., Schaeffer J. M., Meinke P. T. and McKernan R. M. (2000) Anticonvulsant and adverse effects of avermeetin analogs in mice are mediated through the gamma-aminobutyric acid(A) receptor. *J Pharmacol Exp Ther* 295, 1051-1060.
- De Graaf D. C., Hilderson H., Berghen P. and Vercruysse J. (1992) Antigenic differences between the life cycle stages of Cooperia oncophora. *Res Vet Sci* 53, 390-392.

- Delany N. S., Laughton D. L. and Wolstenholme A. J. (1998) Cloning and localisation of an avermectin receptor-related subunit from Haemonchus contortus. *Mol Biochem Parasitol* 97, 177-187.
- Dennis M., Giraudat J., Kotzyba-Hibert F., Goeldner M., Hirth C., Chang J. Y., Lazure C., Chretien M. and Changeux J. P. (1988) Amino acids of the Torpedo marmorata acetylcholine receptor alpha subunit labeled by a photoaffinity ligand for the acetylcholine binding site. *Biochemistry* 27, 2346-2357.
- Dent J. A., Davis M. W. and Avery L. (1997) avr-15 encodes a chloride channel subunit that mediates inhibitory glutamatergic neurotransmission and ivermectin sensitivity in Caenorhabditis elegans. *Embo J* 16, 5867-5879.
- Dent J. A., Smith M. M., Vassilatis D. K. and Avery L. (2000) The genetics of ivermectin resistance in Caenorhabditis elegans. *Proc Natl Acad Sci U S A* 97, 2674-2679.
- Dorny P., Demeulenaere D., Smets K. and Vercruysse J. (2000) Control of gastrointestinal nematodes in first season grazing calves by two strategic treatments with eprinomectin. *Vet Parasitol* **89**, 277-286.
- Driscoll M., Dean E., Reilly E., Bergholz E. and Chalfie M. (1989) Genetic and molecular analysis of a Caenorhabditis elegans beta-tubulin that conveys benzimidazole sensitivity. *J Cell Biol* **109**, 2993-3003.
- Duce I. R. and Scott R. H. (1985) Actions of dihydroavermectin B1a on insect muscle. Br J Pharmacol 85, 395-401.
- Duke B. O., Zea-Flores G. and Munoz B. (1991) The embryogenesis of Onchocerca volvulus over the first year after a single dose of ivermectin. *Trop Med Parasitol* **42**, 175-180.
- Eagleson J. S. and Bowie J. Y. (1986) Oxfendazole resistance in Trichostrongylus axei in cattle in Australia. Vet Rec 119, 604.
- Eagleson J. S. and Allerton G. R. (1992) Efficacy and Safety of Ivermectin Applied Topically to Cattle under Field Conditions in Australia. Aust Vet J 69, 133-134.
- Egerton J. R., Eary C. H. and Suhayda D. (1981) The Anthelmintic Efficacy of Ivermectin in Experimentally Infected Cattle. *Veterinary Parasitology* **8**, 59-70.
- Elard L., Comes A. M. and Humbert J. F. (1996) Sequences of beta-tubulin cDNA from benzimidazole-susceptible and -resistant strains of Teladorsagia circumcincta, a nematode parasite of small ruminants. *Mol Biochem Parasitol* **79**, 249-253.
- Elard L., Sauve C. and Humbert J. F. (1998) Fitness of benzimidazole-resistant and -susceptible worms of Teladorsagia circumcincta, a nematode parasite of small ruminants. *Parasitology* 117 (Pt 6), 571-578.

- Elard L., Cabaret J. and Humbert J. F. (1999) PCR diagnosis of benzimidazole-susceptibility or resistance in natural populations of the small ruminant parasite, Teladorsagia circumcincta. *Vet Parasitol* **80**, 231-237.
- Etter A., Cully D. F., Schaeffer J. M., Liu K. K. and Arena J. P. (1996) An amino acid substitution in the pore region of a glutamate-gated chloride channel enables the coupling of ligand binding to channel gating. *J Biol Chem* 271, 16035-16039.
- Familton A. S., Mason P. and Coles G. C. (2001) Anthelmintic-resistant Cooperia species in cattle. Vet Rec 149, 719-720.
- Feng X. P., Hayashi J., Beech R. N. and Prichard R. K. (2002) Study of the nematode putative GABA type-A receptor subunits: evidence for modulation by ivermectin. *J Neurochem* 83, 870-878.
- Fiel C. A., Saumell C. A., Steffan P. E. and Rodriguez E. M. (2001) Resistance of Cooperia to ivermectin treatments in grazing cattle of the Humid Pampa, Argentina. Vet Parasitol 97, 211-217.
- Forrester S. G., Prichard R. K. and Beech R. N. (2002) A glutamate-gated chloride channel subunit from Haemonchus contortus: expression in a mammalian cell line, ligand binding, and modulation of anthelmintic binding by glutamate. *Biochem Pharmacol* 63, 1061-1068.
- Forrester S. G., Hamdan F. F., Prichard R. K. and Beech R. N. (1999) Cloning, sequencing, and developmental expression levels of a novel glutamate-gated chloride channel homologue in the parasitic nematode Haemonchus contortus. *Biochem Biophys Res Commun* **254**, 529-534.
- Forrester S. G., Prichard R. K., Dent J. A. and Beech R. N. (2003) Haemonchus contortus: HcGluCla expressed in Xenopus oocytes forms a glutamate-gated ion channel that is activated by ibotenate and the antiparasitic drug ivermectin. *Mol Biochem Parasitol* **129**, 115-121.
- Fraser C. M. (1991) The Merck veterinary manual : a handbook of diagnosis, therapy, and disease prevention and control for the veterinarian, 7th Edition, pp xxxii, 1832. Merck & Co., Rahway, N.J.
- Friedman P. A. and Platzer E. G. (1980) Interaction of anthelmintic benzimidazoles with Ascaris suum embryonic tubulin. *Biochim Biophys Acta* **630**, 271-278.
- Fritz L. C., Wang C. C. and Gorio A. (1979) Avermeetin B1a irreversibly blocks postsynaptic potentials at the lobster neuromuscular junction by reducing muscle membrane resistance. *Proc Natl Acad Sci U S A* 76, 2062-2066.

- Fu D. X. and Sine S. M. (1994) Competitive antagonists bridge the alpha-gamma subunit interface of the acetylcholine receptor through quaternary ammonium-aromatic interactions. J Biol Chem 269, 26152-26157.
- Fukushige T., Siddiqui Z. K., Chou M., Culotti J. G., Gogonea C. B., Siddiqui S. S. and Hamelin M. (1999) MEC-12, an alpha-tubulin required for touch sensitivity in C. elegans. J Cell Sci 112 (Pt 3), 395-403.
- Galzi J. L., Revah F., Black D., Goeldner M., Hirth C. and Changeux J. P. (1990) Identification of a novel amino acid alpha-tyrosine 93 within the cholinergic ligands-binding sites of the acetylcholine receptor by photoaffinity labeling. Additional evidence for a three-loop model of the cholinergic ligands-binding sites. *J Biol Chem* **265**, 10430-10437.
- Galzi J. L., Devillers-Thiery A., Hussy N., Bertrand S., Changeux J. P. and Bertrand D. (1992) Mutations in the channel domain of a neuronal nicotinic receptor convert ion selectivity from cationic to anionic. *Nature* 359, 500-505.
- Gasbarre L. C., Leighton E. A. and Sonstegard T. (2001) Role of the bovine immune system and genome in resistance to gastrointestinal nematodes. *Vet Parasitol* **98**, 51-64.
- Geary T. G., Nulf S. C., Favreau M. A., Tang L., Prichard R. K., Hatzenbuhler N. T., Shea M. H., Alexander S. J. and Klein R. D. (1992) Three beta-tubulin cDNAs from the parasitic nematode Haemonchus contortus. *Mol Biochem Parasitol* 50, 295-306.
- Geary T. G., Sims S. M., Thomas E. M., Vanover L., Davis J. P., Winterrowd C. A., Klein R. D., Ho N. F. and Thompson D. P. (1993) Haemonchus contortus: ivermectin-induced paralysis of the pharynx. *Exp Parasitol* 77, 88-96.
- Geerts S. (1986) Levamisole-resistant nematodes. Vet Rec 118, 283.
- Geerts S., Brandt J., Kumar V. and Biesemans L. (1987) Suspected resistance of Ostertagia ostertagi in cattle to levamisole. *Vet Parasitol* 23, 77-82.
- Gerzanich V., Anand R. and Lindstrom J. (1994) Homomers of alpha 8 and alpha 7 subunits of nicotinic receptors exhibit similar channel but contrasting binding site properties. *Mol Pharmacol* 45, 212-220.
- Gill J. H., Redwin J. M., van Wyk J. A. and Lacey E. (1991) Detection of resistance to ivermectin in Haemonchus contortus. *Int J Parasitol* 21, 771-776.
- Gill J. H., Redwin J. M., van Wyk J. A. and Lacey E. (1995) Avermectin inhibition of larval development in Haemonchus contortus--effects of ivermectin resistance. *Int J Parasitol* 25, 463-470.

- Gill J. H., Kerr C. A., Shoop W. L. and Lacey E. (1998) Evidence of multiple mechanisms of avermectin resistance in haemonchus contortus--comparison of selection protocols. Int J Parasitol 28, 783-789.
- Gisselmann G., Pusch H., Hovemann B. T. and Hatt H. (2002) Two cDNAs coding for histaminegated ion channels in D. melanogaster. *Nat Neurosci* 5, 11-12.
- Gogonea C. B., Gogonea V., Ali Y. M., Merz K. M., Jr. and Siddiqui S. S. (1999) Computational prediction of the three-dimensional structures for the Caenorhabditis elegans tubulin family. J Mol Graph Model 17, 90-100, 126-130.
- Grant W. (2000) What is the real target for ivermectin resistance selection in Onchocerca volvulus? *Parasitol Today* **16**, 458-459.
- Grant W. N. and Mascord L. J. (1996) Beta-tubulin gene polymorphism and benzimidazole resistance in trichostrongylus colubriformis. *Int J Parasitol* 26, 71-77.
- Gross S. J., Ryan W. G. and Ploeger H. W. (1999) Anthelmintic treatment of dairy cows and its effect on milk production. *Vet Rec* 144, 581-587.
- Guldenhaupt G. and Burger H. J. (1983) The use of a morantel sustained release bolus in the seasonal control of parasitic gastroenteritis in second-season cattle. *Vet Parasitol* **12**, 313-320.
- Hall C. A., Ritchie L. and Kelly J. D. (1982) Effect of removing anthelmintic selection pressure on the benzimidazole resistance status of Haemonchus contortus and Trichostrongylus colubriformis in sheep. *Res Vet Sci* **33**, 54-57.
- Hammerschlag R. S. and Sisler H. D. (1973) Benomyl and Methyl-2-Benzimidazolecarbamate (Mbc) - Biochemical, Cytological and Chemical Aspects of Toxicity to Ustilago-Maydis and Saccharomyces-Cerevesiae. *Pestic Biochem Phys* 3, 42-54.
- Hardie R. C. (1989) A histamine-activated chloride channel involved in neurotransmission at a photoreceptor synapse. *Nature* **339**, 704-706.
- Hawkins J. A. (1993) Economic benefits of parasite control in cattle. Vet Parasitol 46, 159-173.
- Hejmadi M. V., Jagannathan S., Delany N. S., Coles G. C. and Wolstenholme A. J. (2000) Lglutamate binding sites of parasitic nematodes: an association with ivermectin resistance? *Parasitology* 120 (Pt 5), 535-545.
- Herd R. P., Streitel R. H., McClure K. E. and Parker C. F. (1984) Control of hypobiotic and benzimidazole-resistant nematodes of sheep. *J Am Vet Med Assoc* 184, 726-730.
- Herlich H. (1965) The effects of the intestinal worms, Cooperia pectinata and Cooperia oncophora, on experimentally infected calves. Am J Vet Res 26, 1032-1036.

- Holden-Dye L. and Walker R. J. (1990) Avermectin and avermectin derivatives are antagonists at the 4-aminobutyric acid (GABA) receptor on the somatic muscle cells of Ascaris; is this the site of anthelmintic action? *Parasitology* **101 Pt 2**, 265-271.
- Horoszok L., Raymond V., Sattelle D. B. and Wolstenholme A. J. (2001) GLC-3: a novel fipronil and BIDN-sensitive, but picrotoxinin-insensitive, L-glutamate-gated chloride channel subunit from Caenorhabditis elegans. *Br J Pharmacol* **132**, 1247-1254.
- Horton R. J. (1990) Benzimidazoles in a Wormy World. Parasitol Today 6, 106-106.
- Hosking B. C., Watson T. G. and Leathwick D. M. (1996) Multigeneric resistance to oxfendazole by nematodes in cattle. *Vet Rec* **138**, 67-68.
- Hoyle H. D. and Raff E. C. (1990) Two Drosophila beta tubulin isoforms are not functionally equivalent. *J Cell Biol* **111**, 1009-1026.
- Ibarra O. F. and Jenkins D. C. (1984) The relevance of in vitro anthelmintic screening tests employing the free-living stages of trichostrongylid nematodes. *J Helminthol* 58, 107-112.
- Isenstein R. S. (1963) The life history of Cooperia oncephora (Railliet, 1898) Ransom, 1907, a nematodes parasite of cattle. *J Parasitol* **49**, 235-240.
- Jackson F. (1993) Anthelmintic resistance--the state of play. Br Vet J 149, 123-138.
- Jackson F. and Coop R. L. (2000) The development of anthelmintic resistance in sheep nematodes. *Parasitology* **120 Suppl**, S95-107.
- Jackson R. A., Townsend K. G., Pyke C. and Lance D. M. (1987) Isolation of Oxfendazole Resistant Cooperia-Oncophora in Cattle. *New Zeal Vet J* 35, 187-189.
- Jacobs D. E., Pitt S. R., Foster J. and Fox M. T. (1987) Interactions between chemoprophylaxis and immunity to bovine parasitic gastroenteritis and bronchitis: pilot studies using an oxfendazole pulse release bolus. *Res Vet Sci* **43**, 273-275.
- Jagannathan S., Laughton D. L., Critten C. L., Skinner T. M., Horoszok L. and Wolstenholme A. J. (1999) Ligand-gated chloride channel subunits encoded by the Haemonchus contortus and Ascaris suum orthologues of the Caenorhabditis elegans gbr-2 (avr-14) gene. *Mol Biochem Parasitol* 103, 129-140.
- Jung M. K., Wilder I. B. and Oakley B. R. (1992) Amino acid alterations in the benA (betatubulin) gene of Aspergillus nidulans that confer benomyl resistance. *Cell Motil Cytoskeleton* 22, 170-174.
- Kao P. N. and Karlin A. (1986) Acetylcholine receptor binding site contains a disulfide cross-link between adjacent half-cystinyl residues. *J Biol Chem* **261**, 8085-8088.
- Kaplan, R.M., Chapman, M.R., Tolliver, S.C., Lyons, E.T., Klei, T.R. Characterization of β tubulin genes from cyathostome populations with differing sensitivities to benzimidazole

anthelmintics. American Association of Veterinary Parasitologists, Abstract # 109, Fortyfifth Annual Meeting 2000

Karlin A. (1993) Structure of nicotinic acetylcholine receptors. Curr Opin Neurobiol 3, 299-309.

- Karlin A., Holtzman E., Yodh N., Lobel P., Wall J. and Hainfeld J. (1983) The arrangement of the subunits of the acetylcholine receptor of Torpedo californica. J Biol Chem 258, 6678-6681.
- Kass I. S., Stretton A. O. and Wang C. C. (1984) The effects of avermectin and drugs related to acetylcholine and 4-aminobutyric acid on neurotransmission in Ascaris suum. *Mol Biochem Parasitol* 13, 213-225.
- Kass I. S., Wang C. C., Walrond J. P. and Stretton A. O. (1980) Avermectin B1a, a paralyzing anthelmintic that affects interneurons and inhibitory motoneurons in Ascaris. *Proc Natl Acad Sci U S A* 77, 6211-6215.
- Kelly J. D., Whitlock H. V., Thompson H. G., Hall C. A., Martin I. C. and Le Jambre L. F. (1978) Physiological characteristics of free-living and parasitic stages of strains of Haemonchus contortus, susceptible or resistant to benzimidazole anthelmintics. *Res Vet Sci* 25, 376-385.
- Khakh B. S., Proctor W. R., Dunwiddie T. V., Labarca C. and Lester H. A. (1999) Allosteric control of gating and kinetics at P2X(4) receptor channels. *J Neurosci* **19**, 7289-7299.
- Kloosterman A., Albers G. A. and van den Brink R. (1984) Negative interactions between Ostertagia ostertagi and Cooperia oncophora in calves. *Vet Parasitol* **15**, 135-150.
- Kloosterman A., Ploeger H. W. and Frankena K. (1991) Age resistance in calves to Ostertagia ostertagi and Cooperia oncophora. Vet Parasitol 39, 101-113.
- Koenraadt H., Somerville S. C. and Jones A. L. (1992) Characterization of Mutations in the Beta-Tubulin Gene of Benomyl-Resistant Field Strains of Venturia-Inaequalis and Other Plant Pathogenic Fungi. *Phytopathology* **82**, 1348-1354.
- Kotze A. C. (1998) Effects of macrocyclic lactones on ingestion in susceptible and resistant Haemonchus contortus larvae. *J Parasitol* 84, 631-635.
- Krause R. M., Buisson B., Bertrand S., Corringer P. J., Galzi J. L., Changeux J. P. and Bertrand D. (1998) Ivermectin: a positive allosteric effector of the alpha7 neuronal nicotinic acetylcholine receptor. *Mol Pharmacol* 53, 283-294.
- Krusek J. and Zemkova H. (1994) Effect of ivermectin on gamma-aminobutyric acid-induced chloride currents in mouse hippocampal embryonic neurones. *Eur J Pharmacol* 259, 121-128.
- Kwa M. S., Veenstra J. G. and Roos M. H. (1993a) Molecular characterisation of beta-tubulin genes present in benzimidazole-resistant populations of Haemonchus contortus. *Mol Biochem Parasitol* **60**, 133-143.

- Kwa M. S., Veenstra J. G. and Roos M. H. (1994) Benzimidazole resistance in Haemonchus contortus is correlated with a conserved mutation at amino acid 200 in beta-tubulin isotype 1.
 Mol Biochem Parasitol 63, 299-303.
- Kwa M. S., Kooyman F. N., Boersema J. H. and Roos M. H. (1993b) Effect of selection for benzimidazole resistance in Haemonchus contortus on beta-tubulin isotype 1 and isotype 2 genes. *Biochem Biophys Res Commun* 191, 413-419.
- Kwa M. S., Veenstra J. G., Van Dijk M. and Roos M. H. (1995) Beta-tubulin genes from the parasitic nematode Haemonchus contortus modulate drug resistance in Caenorhabditis elegans. *J Mol Biol* **246**, 500-510.
- Lacey E. (1988) The role of the cytoskeletal protein, tubulin, in the mode of action and mechanism of drug resistance to benzimidazoles. *Int J Parasitol* **18**, 885-936.
- Lacey E. (1990) Mode of Action of Benzimidazoles. Parasitol Today 6, 112-115.
- Laughton D. L., Lunt G. G. and Wolstenholme A. J. (1997a) Alternative splicing of a Caenorhabditis elegans gene produces two novel inhibitory amino acid receptor subunits with identical ligand binding domains but different ion channels. *Gene* **201**, 119-125.
- Laughton D. L., Lunt G. G. and Wolstenholme A. J. (1997b) Reporter gene constructs suggest that the Caenorhabditis elegans avermectin receptor beta-subunit is expressed solely in the pharynx. *J Exp Biol* 200 (Pt 10), 1509-1514.
- Le Jambre L. F. (1993) Molecular variation in trichostrongylid nematodes from sheep and cattle. *Acta Trop* **53**, 331-343.
- Le Jambre L. F., Martin P. J. and Jarrett R. G. (1982) Comparison of changes in resistance of Haemonchus contortus eggs following withdrawal of thiabendazole selection. *Res Vet Sci* 32, 39-43.
- Le Jambre L. F., Gill J. H., Lenane I. J. and Baker P. (2000) Inheritance of avermectin resistance in Haemonchus contortus. *Int J Parasitol* **30**, 105-111.
- Le Novere N., Grutter T. and Changeux J. P. (2002) Models of the extracellular domain of the nicotinic receptors and of agonist- and Ca2+-binding sites. *Proc Natl Acad Sci U S A* **99**, 3210-3215.
- Lea T. J. and Usherwood P. N. (1973a) Effect of ibotenic acid on chloride permeability of insect muscle-fibres. *Comp Gen Pharmacol* **4**, 351-363.
- Lea T. J. and Usherwood P. N. (1973b) The site of action of ibotenic acid and the identification of two populations of glutamate receptors on insect muscle-fibres. *Comp Gen Pharmacol* 4, 333-350.

Lubega G. W., Klein R. D., Geary T. G. and Prichard R. K. (1994) Haemonchus contortus: the role of two beta-tubulin gene subfamilies in the resistance to benzimidazole anthelmintics. *Biochem Pharmacol* 47, 1705-1715.

Luduena R. F. (1993) Are tubulin isotypes functionally significant. Mol Biol Cell 4, 445-457.

- Martin M., Czajkowski C. and Karlin A. (1996) The contributions of aspartyl residues in the acetylcholine receptor gamma and delta subunits to the binding of agonists and competitive antagonists. *J Biol Chem* 271, 13497-13503.
- Martin M. D. and Karlin A. (1997) Functional effects on the acetylcholine receptor of multiple mutations of gamma Asp174 and delta Asp180. *Biochemistry* **36**, 10742-10750.
- Martin P. J. (1987) Development and control of resistance to anthelmintics. Int J Parasitol 17, 493-501.
- Martin P. J., Anderson N. and Jarrett R. G. (1989) Detecting Benzimidazole Resistance with Fecal Egg Count Reduction Tests and Invitro Assays. *Aust Vet J* 66, 236-240.
- Martin P. J., Anderson N., Brown T. H. and Miller D. W. (1988) Changes in resistance of Ostertagia spp. to thiabendazole following natural selection or treatment with levamisole. Int J Parasitol 18, 333-340.
- Martin R. J. (1996) An electrophysiological preparation of Ascaris suum pharyngeal muscle reveals a glutamate-gated chloride channel sensitive to the avermectin analogue, milbemycin D. *Parasitology* 112 (Pt 2), 247-252.
- McKenna P. B. (1990) The Detection of Anthelmintic Resistance by the Fecal Egg Count Reduction Test - an Examination of Some of the Factors Affecting Performance and Interpretation. *New Zeal Vet J* 38, 142-147.
- McKenna P. B. (1991) Resistance to Benzimidazole Anthelmintics in Cattle in New-Zealand. New Zeal Vet J 39, 154-155.
- McKenna P. B. (1995) Topically Applied Ivermectin and Cooperia Infections in Cattle. New Zeal Vet J 43, 44-44.
- McKenna P. B. (1996) Potential limitations of the undifferentiated faecal egg count reduction test for the detection of anthelmintic resistance in sheep. *New Zeal Vet J* 44, 73-75.
- McKenna P. B. (1997) Further potential limitations of the undifferentiated faecal egg count reduction test for the detection of anthelmintic resistance in sheep. *New Zeal Vet J* **45**, 244-246.
- Mealey K. L., Bentjen S. A., Gay J. M. and Cantor G. H. (2001) Ivermectin sensitivity in collies is associated with a deletion mutation of the mdr1 gene. *Pharmacogenetics* **11**, 727-733.

- Mejia M. E., Fernandez Igartua B. M., Schmidt E. E. and Cabaret J. (2003) Multispecies and multiple anthelmintic resistance on cattle nematodes in a farm in Argentina: the beginning of high resistance? *Vet Res* 34, 461-467.
- Miyazawa A., Fujiyoshi Y., Stowell M. and Unwin N. (1999) Nicotinic acetylcholine receptor at 4.6 A resolution: transverse tunnels in the channel wall. *J Mol Biol* **288**, 765-786.
- Molento M. B. and Prichard R. K. (1999) Effects of the multidrug-resistance-reversing agents verapamil and CL 347,099 on the efficacy of ivermectin or moxidectin against unselected and drug-selected strains of Haemonchus contortus in jirds (Meriones unguiculatus). *Parasitol Res* 85, 1007-1011.
- Moreno M. S. and Barrett J. (1979) Monoamine oxidase in adult Hymenolepis diminuta (Cestoda). *Parasitology* 78, 1-5.
- Morris C. A., Cullen N. G., Green R. S. and Hickey S. M. (2002) Sire effects on antibodies to nematode parasites in grazing dairy cows. *New Zeal J Agr Res* **45**, 179-185.
- Nogales E., Wolf S. G. and Downing K. H. (1998) Structure of the alpha beta tubulin dimer by electron crystallography. *Nature* **391**, 199-203.
- Ortells M. O. and Lunt G. G. (1995) Evolutionary history of the ligand-gated ion-channel superfamily of receptors. *Trends Neurosci* 18, 121-127.
- Paiement J., Prichard R. K. and Ribeiro P. (1999) Haemonchus contortus: characterization of a glutamate binding site in unselected and ivermectin-selected larvae and adults. *Exp Parasitol* 92, 32-39.
- Parkins J. J., Taylor L. M., Holmes P. H., Bairden K., Salman S. K. and Armour J. (1990) Pathophysiological and parasitological studies on a concurrent infection of Ostertagia ostertagi and Cooperia oncophora in calves. *Res Vet Sci* 48, 201-208.
- Parmentier H. K., Ploeger H. W., Nieuwland M. G., Souren P. J., Van Pinxteren L. A., Rietveld F. W., De Vries Reilingh G. and Kloosterman A. (1995) Low molecular weight Cooperia oncophora antigens: characterization and humoral immune responses in calves mono-infected with 100,000 infective larvae. *Vet Parasitol* 59, 219-230.
- Paul A. J., Tranquilli W. J., Seward R. L., Todd K. S., Jr. and DiPietro J. A. (1987) Clinical observations in collies given ivermectin orally. *Am J Vet Res* 48, 684-685.
- Ploeger H. W. and Kloosterman A. (1993) Gastrointestinal nematode infections and weight gain in dairy replacement stock: first-year calves. *Vet Parasitol* 46, 223-241.
- Ploeger H. W., Kloosterman A., Borgsteede F. H. and Eysker M. (1990) Effect of naturally occurring nematode infections in the first and second grazing season on the growth performance of second-year cattle. *Vet Parasitol* 36, 57-70.

- Ploeger H. W., Borgsteede F. H., Sol J., Mirck M. H., Huyben M. W., Kooyman F. N. and Eysker M. (2000) Cross-sectional serological survey on gastrointestinal and lung nematode infections in first and second-year replacement stock in the netherlands: relation with management practices and use of anthelmintics. *Vet Parasitol* 90, 285-304.
- Portillo V., Jagannathan S. and Wolstenholme A. J. (2003) Distribution of glutamate-gated chloride channel subunits in the parasitic nematode Haemonchus contortus. *J Comp Neurol* **462**, 213-222.
- Prichard R. (2001) Genetic variability following selection of Haemonchus contortus with anthelmintics. *Trends Parasitol* 17, 445-453.
- Prichard R. K. (1973) The fumarate reductase reaction of Haemonchus contortus and the mode of action of some anthelmintics. *Int J Parasitol* **3**, 409-417.
- Prichard R. K. (1990) Anthelmintic resistance in nematodes: extent, recent understanding and future directions for control and research. *Int J Parasitol* **20**, 515-523.
- Prichard R. K., Hall C. A., Kelly J. D., Martin I. C. A. and Donald A. D. (1980) The Problem of Anthelmintic Resistance in Nematodes. *Aust Vet J* 56, 239-251.
- Raff E. C., Fackenthal J. D., Hutchens J. A., Hoyle H. D. and Turner F. R. (1997) Microtubule architecture specified by a beta-tubulin isoform. *Science* 275, 70-73.
- Prichard, R., Oxberry, M., Bounhas, Y., Sharma, S., Lubega, G., Geary, T. Polymerization and BZ binding assays with recombinant α- and β- tubulins from *Haemonchus contortus*. American Association of Veterinary Parasitologists, Abstract # 108, Forty-fifth Annual Meeting 2000.
- Ranganathan R., Cannon S. C. and Horvitz H. R. (2000) MOD-1 is a serotonin-gated chloride channel that modulates locomotory behaviour in C. elegans. *Nature* **408**, 470-475.
- Ranjan S., Trudeau C., Prichard R. K., von Kutzleben R. and Carrier D. (1992) Efficacy of moxidectin against naturally acquired nematode infections in cattle. Vet Parasitol 41, 227-231.
- Raymond V. and Sattelle D. B. (2002) Novel animal-health drug targets from ligand-gated chloride channels. *Nat Rev Drug Discov* 1, 427-436.
- Reinemeyer C. R. (1995) Should You Deworm Your Clients Dairy Cattle? Vet Med-Us 90, 496-502.
- Roach M. C., Boucher V. L., Walss C., Ravdin P. M. and Luduena R. F. (1998) Preparation of a monoclonal antibody specific for the class I isotype of beta-tubulin: the beta isotypes of tubulin differ in their cellular distributions within human tissues. *Cell Motil Cytoskeleton* 39, 273-285.
- Rohrer S. P., Birzin E. T., Eary C. H., Schaeffer J. M. and Shoop W. L. (1994) Ivermectin binding sites in sensitive and resistant Haemonchus contortus. *J Parasitol* 80, 493-497.
- Roos M. H., Boersema J. H., Borgsteede F. H., Cornelissen J., Taylor M. and Ruitenberg E. J. (1990) Molecular analysis of selection for benzimidazole resistance in the sheep parasite Haemonchus contortus. *Mol Biochem Parasitol* 43, 77-88.
- Roulet A., Puel O., Gesta S., Lepage J. F., Drag M., Soll M., Alvinerie M. and Pineau T. (2003) MDR1-deficient genotype in Collie dogs hypersensitive to the P-glycoprotein substrate ivermectin. *Eur J Pharmacol* 460, 85-91.
- Sangster N. (1996) Pharmacology of anthelmintic resistance. Parasitology 113 Suppl, S201-216.
- Sangster N. C. (1999) Anthelmintic resistance: past, present and future. *Int J Parasitol* 29, 115-124; discussion 137-118.
- Sangster N. C. (2001) Managing parasiticide resistance. Vet Parasitol 98, 89-109.
- Sangster N. C. and Prichard R. K. (1984) Uptake of thiabendazole and its effects on glucose uptake and carbohydrate levels in the thiabendazole-resistant and susceptible Trichostrongylus colubriformis. *Int J Parasitol* 14, 121-126.
- Sasaki Y. and Kitagawa H. (1993) Effects of milberrycin D on microfilarial number and reproduction of Dirofilaria immitis in dogs. *J Vet Med Sci* 55, 763-769.
- Satrija F. and Nansen P. (1993) Experimental concurrent infections with Ostertagia ostertagi and Cooperia oncophora in the calf. *Res Vet Sci* 55, 92-97.
- Satrija F., Nansen P., Jorgensen R. J., Monrad J. and Esfandiari A. (1996) The effects of firstseason strategic and tactical ivermectin treatments on trichostrongylosis in the first- and second-season grazing. *Vet Parasitol* **64**, 219-237.
- Savage C., Hamelin M., Culotti J. G., Coulson A., Albertson D. G. and Chalfie M. (1989) mec-7 is a beta-tubulin gene required for the production of 15-protofilament microtubules in Caenorhabditis elegans. *Genes Dev* **3**, 870-881.
- Savage C., Xue Y., Mitani S., Hall D., Zakhary R. and Chalfie M. (1994) Mutations in the Caenorhabditis elegans beta-tubulin gene mec-7: effects on microtubule assembly and stability and on tubulin autoregulation. *J Cell Sci* **107** (**Pt 8**), 2165-2175.
- Schaeffer J. M. and Haines H. W. (1989) Avermectin binding in Caenorhabditis elegans. A twostate model for the avermectin binding site. *Biochem Pharmacol* 38, 2329-2338.
- Schinkel A. H., Smit J. J., van Tellingen O., Beijnen J. H., Wagenaar E., van Deemter L., Mol C. A., van der Valk M. A., Robanus-Maandag E. C., te Riele H. P. and et al. (1994) Disruption of the mouse mdr1a P-glycoprotein gene leads to a deficiency in the blood-brain barrier and to increased sensitivity to drugs. *Cell* 77, 491-502.

- Scholl P. J., Guillot F. S. and Wang G. T. (1992) Moxidectin: systemic activity against common cattle grubs (Hypoderma lineatum) (Diptera: Oestridae) and trichostrongyle nematodes in cattle. *Vet Parasitol* 41, 203-209.
- Schwarz P. M., Liggins J. R. and Luduena R. F. (1998) Beta-tubulin isotypes purified from bovine brain have different relative stabilities. *Biochemistry* 37, 4687-4692.
- Shan Q., Haddrill J. L. and Lynch J. W. (2001) Ivermectin, an unconventional agonist of the glycine receptor chloride channel. *J Biol Chem* 276, 12556-12564.
- Shoop W. L., Mrozik H. and Fisher M. H. (1995) Structure and activity of avermeetins and milberrycins in animal health. *Vet Parasitol* **59**, 139-156.
- Shoop W. L., DeMontigny P., Fink D. W., Williams J. B., Egerton J. R., Mrozik H., Fisher M. H., Skelly B. J. and Turner M. J. (1996a) Efficacy in sheep and pharmacokinetics in cattle that led to the selection of eprinomectin as a topical endectocide for cattle. *Int J Parasitol* 26, 1227-1235.
- Shoop W. L., Egerton J. R., Eary C. H., Haines H. W., Michael B. F., Mrozik H., Eskola P., Fisher M. H., Slayton L., Ostlind D. A., Skelly B. J., Fulton R. K., Barth D., Costa S., Gregory L. M., Campbell W. C., Seward R. L. and Turner M. J. (1996b) Eprinomectin: a novel avermectin for use as a topical endectocide for cattle. *Int J Parasitol* 26, 1237-1242.
- Silvestre A. and Humbert J. F. (2000) A molecular tool for species identification and benzimidazole resistance diagnosis in larval communities of small ruminant parasites. *Exp Parasitol* **95**, 271-276.
- Silvestre A. and Cabaret J. (2002) Mutation in position 167 of isotype 1 beta-tubulin gene of Trichostrongylid nematodes: role in benzimidazole resistance? *Mol Biochem Parasitol* **120**, 297-300.
- Simpkin K. G. and Coles G. C. (1978) Instability of benzimidazole resistance in nematode eggs. *Res Vet Sci* 25, 249-250.
- Sine S. M. (2002) The nicotinic receptor ligand binding domain. J Neurobiol 53, 431-446.
- Smit A. B., Syed N. I., Schaap D., van Minnen J., Klumperman J., Kits K. S., Lodder H., van der Schors R. C., van Elk R., Sorgedrager B., Brejc K., Sixma T. K. and Geraerts W. P. (2001) A glia-derived acetylcholine-binding protein that modulates synaptic transmission. *Nature* 411, 261-268.
- Stafford K. and Coles G. C. (1999) Nematode control practices and anthelmintic resistance in dairy calves in the south west of England. *Vet Rec* 144, 659-661.
- Styles J. A. and Garner R. (1974) Benzimidazolecarbamate Methyl-Ester Evaluation of Its Effects in-Vivo and in-Vitro. *Mutat Res* 26, 177-187.

- Suarez V. H., Busetti M. R. and Babinec F. J. (2001) Effects of previous suppressive anthelmintic treatments on subsequent nematode infection in fattening cattle in Argentina. *Vet Parasitol* 96, 221-231.
- Taylor M. A. and Hunt K. R. (1989) Anthelmintic drug resistance in the UK. Vet Rec 125, 143-147.
- Taylor M. A., Hunt K. R. and Goodyear K. L. (2002) Anthelmintic resistance detection methods. *Vet Parasitol* **103**, 183-194.
- Thomas G. W., Barton N. J., McGowan A. A., Mickan F. J., Miller G. and James R. E. (1984) Effect of whole herd anthelmintic treatment on milk production of dairy cows. *Res Vet Sci* 36, 266-269.
- Unwin N. (1989) The structure of ion channels in membranes of excitable cells. *Neuron* **3**, 665-676.
- Unwin N. (1993) Nicotinic acetylcholine receptor at 9 A resolution. J Mol Biol 229, 1101-1124.
- Unwin N. (1995) Acetylcholine receptor channel imaged in the open state. Nature 373, 37-43.
- Van den Bossche H. and De Nollin S. (1973) Effects of mebendazole on the absorption of low molecular weight nutrients by Ascaris suum. *Int J Parasitol* **3**, 401-407.
- van Wyk J. A. and van Schalkwyk P. C. (1990) A novel approach to the control of anthelminticresistant Haemonchus contortus in sheep. *Vet Parasitol* **35**, 61-69.
- Vassilatis D. K., Arena J. P., Plasterk R. H., Wilkinson H. A., Schaeffer J. M., Cully D. F. and Van der Ploeg L. H. (1997a) Genetic and biochemical evidence for a novel avermectinsensitive chloride channel in Caenorhabditis elegans. Isolation and characterization. J Biol Chem 272, 33167-33174.
- Vassilatis D. K., Elliston K. O., Paress P. S., Hamelin M., Arena J. P., Schaeffer J. M., Van der Ploeg L. H. and Cully D. F. (1997b) Evolutionary relationship of the ligand-gated ion channels and the avermectin-sensitive, glutamate-gated chloride channels. J Mol Evol 44, 501-508.
- Vercruysse J. and Dorny P. (1999) Integrated control of nematode infections in cattle: a reality? A need? A future? *Int J Parasitol* **29**, 165-175; discussion 183-164.
- Vercruysse J. and Claerebout E. (2001) Treatment vs non-treatment of helminth infections in cattle: defining the threshold. *Vet Parasitol* **98**, 195-214.
- Vercruysse J., Hilderson H. and Claerebout E. (1994) Effect of Chemoprophylaxis on Immunity to Gastrointestinal Nematodes in Cattle. *Parasitol Today* **10**, 129-132.

- Vercruysse J., Hilderson H. and Claerebout E. (1995) Effect of chemoprophylaxis with avermectins on the immune response to gastrointestinal nematodes in first-season grazing calves. *Vet Parasitol* 58, 35-48.
- Vermunt J. J., West D. M. and Pomroy W. E. (1995) Multiple resistance to ivermectin and oxfendazole in Cooperia species of cattle in New Zealand. Vet Rec 137, 43-45.
- Vermunt J. J., West D. M. and Pomroy W. E. (1996) Inefficacy of moxidectin and doramectin against ivermectin-resistant Cooperia spp of cattle in New Zealand. New Zeal Vet J 44, 188-193.
- Waller P. J. (1994) The development of anthelmintic resistance in ruminant livestock. Acta Trop 56, 233-243.
- Waller P. J. (1997) Anthelmintic resistance. Vet Parasitol 72, 391-405; discussion 405-312.
- West D. M., Vermunt J. J., Pomroy W. E. and Bentall H. P. (1994) Inefficacy of Ivermectin against Cooperia Spp Infection in Cattle. *New Zeal Vet J* 42, 192-193.
- Whiting P., McKernan R. M. and Iversen L. L. (1990) Another mechanism for creating diversity in gamma-aminobutyrate type A receptors: RNA splicing directs expression of two forms of gamma 2 phosphorylation site. *Proc Natl Acad Sci U S A* **87**, 9966-9970.
- Xu M., Molento M., Blackhall W., Ribeiro P., Beech R. and Prichard R. (1998) Ivermectin resistance in nematodes may be caused by alteration of P-glycoprotein homolog. *Mol Biochem Parasitol* **91**, 327-335.
- Yadav C. L. and Verma S. P. (1997) Morantel resistance by Haemonchus placei in cattle. *Vet Rec* 141, 499-500.
- Zheng Y., Hirschberg B., Yuan J., Wang A. P., Hunt D. C., Ludmerer S. W., Schmatz D. M. and Cully D. F. (2002) Identification of two novel Drosophila melanogaster histamine-gated chloride channel subunits expressed in the eye. J Biol Chem 277, 2000-2005.

CHAPTER 2

Inefficacy of ivermectin against a resistant Cooperia oncophora field isolate

Annete I. Njue and Roger K. Prichard

Submitted, Parasitology Research

Abstract

Since 1999, two *Cooperia oncophora* isolates, originally obtained from the UK, were maintained by regular passage through calves at the Macdonald Campus, McGill University farm. One isolate, IVS, was susceptible to ivermectin, while the IVR isolate was resistant to ivermectin. These two isolates were used to study the mechanisms of ivermectin resistance. To confirm the susceptible/ resistant status after four years of passaging through calves, a controlled study was performed in which calves infected with IVS and IVR infective larvae were treated with ivermectin. Ivermectin at the recommended dose was 100% effective at eliminating the IVS isolate, since no eggs were found in feces, and no adult worms were found in the small intestine of the treated IVS-infected calf. In contrast, the IVR-infected calf continued to pass eggs in feces even after treatment with ivermectin, and adult worms were found in the small intestine at necropsy.





2.1 Introduction

Ivermectin is a highly potent broad-spectrum anthelmintic that is widely used in cattle. It is available in injectable, oral and topical formulations for use in cattle (Vermunt et al. 1995). *Cooperia* is one of the target species for ivermectin, though the topical and injectable formulations have been shown to have lower than expected efficacy against this species (Benz et al. 1989; Bisset et al. 1990; West et al. 1994; McKenna 1995). *Cooperia* is the dose-limiting species for ivermectin, and the lower efficacy against this species may exert a high selection pressure for resistance (Vermunt et al. 1995).

Cooperia species resistant to ivermectin have been identified under field conditions in New Zealand (Vermunt et al. 1995, 1996; Familton et al. 2001), the UK (Coles et al. 1998, 2001), and Argentina (Anziani et al. 2001; Fiel et al. 2001). Our work focuses on understanding the mechanisms of ivermectin resistance in *Cooperia oncophora*. An ivermectin-resistant isolate (IVR), originally identified by Coles et al. (1998), has been central to our work. A second isolate, designated as ivermectin-susceptible (IVS), has also been used. The parasites have been maintained at the McGill University Macdonald campus farm by regular passage through male Holstein calves. To establish the susceptible/ resistant status of the two isolates after four years of passaging, they were subjected to a controlled efficacy test.

2.2 Materials and Methods

2.2.1 Parasites

Two *C. oncophora* isolates (IVS and IVR) were kindly provided by Dr. G. C. Coles (University of Bristol, UK). The IVS isolate had been maintained without anthelmintic

pressure at Weybridge Experimental Station, UK, while IVR represents a field isolate originally obtained from a farm in Somerset, UK, where ivermectin resistance was reported (Coles et al. 1998).

2.2.2 Animals

Four three-month old calves were used in the study. They were housed indoors in individual pens at the Macdonald Campus, McGill University farm. The animals were fed hay and water *ad libitum*, along with restricted grain supplement. Fourteen days prior to the start of the experiment, the calves were drenched with fenbendazole at a dose rate of 10 mg per kg body weight to clear any existing helminth infections. Prior to infection with IVS or IVR infective larvae, animals were checked for general health, and feces collected to test the worm-free status of the animals.

2.2.3 Fecal egg count reduction test (FECRT)

To determine the efficacy of ivermectin against IVS and IVR worms, two calves were infected with 10,000 IVS infective larvae (L3), and two others were infected with 10,000 IVR L3. In each group, one animal was treated with ivermectin (Ivomec®, Merial) at the recommended dose rate of 200 μ g per kg body weight in cattle, while the second animal was left untreated to act as control. The ivermectin dose was calculated based on the weight of the animal. Treatments were administered as subcutaneous injections.

Fecal samples were collected *per rectum* from all animals at the time of treatment, and seven and 14 days post-treatment. Egg counts were performed on fecal samples using a modified McMaster technique (Ministry of Agriculture Fisheries and Food. 1986). 4.5 grams of feces were thoroughly mixed in 40.5 ml of water. The suspension was sieved through a piece of gauze. The strained fluid was poured into a centrifuge tube, and centrifuged for 2 min at 1500 r.p.m. The supernatant was discarded, and the sediment loosened by stirring. The tube was then filled with saturated sodium chloride solution to the same level as before. The tube contents were thoroughly mixed by inverting the tube five or six times, and a McMaster slide chamber filled using a Pasteur pipette. After mixing again, the second chamber on the slide was filled. All the eggs in both chambers were counted, and the total number multiplied by 10 and expressed as eggs per gram of feces.

The anthelmintic efficacy was calculated based on the percent reduction in fecal egg counts using the formula:

 $(C - T) / C \times 100 = \%$ efficacy,

where C and T are the fecal egg counts of the control (untreated) and treated animals, respectively. Ivermectin resistance was defined as less than 95% reduction in fecal egg counts post-treatment (Coles et al. 1992).

2.2.4 Calculating Reduction in Worm Burden

Necropsy of all four animals was performed on days 22 and 23 post-treatment. The contents of the small intestine were emptied into a sampling bucket, and the small intestine split open longitudinally using scissors. The mucosal surface was washed thoroughly with warm water (38°C). 1 liter aliquots were then processed for worm

recovery and enumeration by a modification of the agar gel technique of Slotved et al. (1996). A 2% agar solution (Bacto-Agar) was prepared by mixing the agar with tap water. This was heated to boiling point and cooled to 45°C. One liter of intestinal washings was mixed with 1 liter of the 2% agar solution at 45°C. The mixture was then poured into eight trays in which disposable J-cloths (22 x 28 cm) had been placed. The agar gel solidified after a few minutes, and the cloths with the adhering gel were clipped to a stick and placed in a plastic container, with the sticks being used to suspend the cloths in the container. The container was then filled with RPMI medium at 38°C, and incubated at 38°C for three hours. The worms migrated out of the gel and collected at the bottom of the container. These were picked and counted. Worms remaining in the agar gel were also picked and included in the total count. Percent reduction in adult worm burden was calculated using the formula listed above.

2.3 Results

To determine the efficacy of ivermectin against IVS and IVR worms, two calves were infected with 10,000 IVS L3, and two others infected with 10,000 IVR L3. Calves infected with the IVS isolate were in good body condition, and did not show any clinical signs of parasitic gastroenteritis (PGE). Following treatment of one of the IVS-infected animals with ivermectin, no eggs were observed in the feces (Table 1). In contrast, the two calves infected with the IVR isolate had diarrhea, rough hair coats, and were unthrifty. Even after one of the IVR-infected calves was treated with ivermectin, it continued to pass eggs in feces (Table 1), and remained in poor condition.



Table 1. Pre- and post-treatment fecal egg counts from untreated calves and calves treated with injectable ivermectin. Day 0 = day of treatment, and Day 7 and 14 = days post-treatment.

Animal	Day 0	Day 7	Day 14
IVS treated	170	0	0
IVS untreated	110	110	90
% reduction	-	100	100
IVR treated	190	30	40
IVR untreated	180	130	180
% reduction	-	76.9	77.8

The efficacy of ivermectin was measured by the percentage reduction in post-treatment fecal egg counts in treated animals compared with untreated animals. The fecal egg counts were determined on days 0, 7 and 14 post-treatment, and are shown in Table 1. The values for the treated animals were compared with those for the untreated animal in the same group. Treatment with the recommended dose of ivermectin ($200 \mu g / kg$ body weight) was 100% effective against the IVS isolate, since no eggs were seen in the treated IVS animal on days 7 and 14 post-treatment. For the IVR isolate, ivermectin reduced fecal egg counts by 76.9% and 77.8% at days 7 and 14 post-treatment, respectively.

Table 2 shows adult worm counts for the four calves. Adult worms were found in untreated IVS- and IVR-infected animals. No adult worms were found in the small intestine of the IVS-infected animals following treatment with ivermectin. In contrast, ivermectin reduced the worm burden of the IVR-infected animal by only 70.5%.

Animal	Post-treatment Adult Worm Burden
IVS treated	0
IVS untreated	1330
% reduction	100
IVR treated	250
IVR untreated	848
% reduction	70.5

Table 2. Post-treatment worm burdens in treated and untreated calves

2.4 Discussion

The present study compared the efficacy of ivermectin against IVS and IVR isolates of *C. oncophora*, and confirmed the ivermectin-resistant status of the IVR isolate, which was previously reported by Coles et al. (1998). While ivermectin was 100% effective against the susceptible IVS isolate, efficacy against the IVR isolate was 77.8% on day 14 post-treatment. FECRT results below 95% are indicative of anthelmintic resistance (Coles et al. 1992). These findings were supported by the adult worm counts, where treatment of

IVR-infected animals resulted in a 70.5% reduction in adult worms, while no worms were found in the treated IVS-infected animal.

Consistent with the results of Coles et al. (2001), the IVR-infected animal remained in poor body condition even after treatment, though the egg count was low (40 EPG on day 14 post-treatment). At slaughter, adult worms were found in the small intestine of this animal, suggesting that ivermectin was ineffective at eliminating the adult IVR worms. Of interest is the observation that the untreated IVS-infected animal had a higher worm burden that the IVR-infected animals, yet the IVS-infected animal did not show any clinical signs of PGE. As stated by Coles et al. (2001), ivermectin-susceptible isolates appear to be of low pathogenicity compared to ivermectin-resistant isolates. Calves infected with 10,000 to 20,000 ivermectin-susceptible L3 did not show any clinical signs of disease (Borgsteede and Hendriks 1979; Coop et al. 1979), while animals infected with 15,000 or 20,000 ivermectin-resistant L3 showed edema, weight loss and diarrhea (Coles et al. 2001; Anziani et al. 2001). Cooperia is generally considered a parasite of low pathogenicity, and young calves develop immunity to this parasite following exposure in the first grazing season (Coop et al. 1979; Vermunt et al. 1995; Claerebout et al. 1999). However, at high infection levels (a single dose of 200,000 L3, or 10 000 larvae per day for six weeks), Cooperia can cause clinical disease (Borgsteede and Hendriks 1979; Armour et al. 1987). The ivermectin-resistant isolate appears to be more pathogenic than the susceptible isolate, with infected animals showing clinical signs of parasitic gastroenteritis (Anziani et al. 2001; Coles et al. 2001), including failure to thrive, that could impact negatively on productivity (Coop et al. 1979; Armour et al. 1987; Vermunt et al. 1995).

Doramectin and moxidectin were found to be ineffective against ivermectinresistant *C. oncophora* (Vermunt et al. 1995, 1996; Anziani et al. 2001), and this suggests that ivermectin-resistant *C. oncophora* show side resistance to doramectin and moxidectin, both members of the macrocyclic lactone class of anthelmintics. Levamisole, fenbendazole and oxfendazole were used successfully against ivermectinresistant *C. oncophora* (Anziani et al. 2001; Coles et al. 1998, 2001). Levamisole belongs to the imidazothiazole class of anthelmintics, while fenbendazole and oxfendazole are benzimidazoles. These anthelmintics act via mechanisms that are distinct from those of the macrocyclic lactones. The imidazothiazoles act on worm acetylcholine receptors, while the benzimidazoles act on tubulin (Kwa et al. 1993, 1995; Sangster 1999). Chloride channels gated by the neurotransmitters glutamate and gammaaminobutyric acid (GABA) have been identified as targets for the macrocyclic lactones (Holden-Dye and Walker 1990; Cully et al. 1994; Arena et al. 1995; Martin 1996; Dent et al. 2000; Feng et al. 2002).

The FECRT is the most widely used test for detecting the presence of anthelmintic resistance (Taylor et al. 2002). It compares egg counts in treated and non-treated animals at the time of treatment and 10-14 days later, which is less than the minimum prepatent periods of important nematode species infecting ruminants (Bowman et al. 2003). While the FECRT is easy to perform, its major drawback is that it can only detect resistance if more than 25% of the worms in the population are resistant (Martin et al. 1989). No other technique is used routinely for detecting anthelmintic resistance, and there is a need for developing sensitive techniques if low-level ivermectin resistance is to be detected. This would allow action to be taken before a high proportion of worms in

the population are resistant. Knowledge of the molecular mechanism of benzimidazole resistance has made possible the development of a sensitive molecular tool for detecting benzimidazole resistance in individual worms (Kwa et al. 1994; Elard and Humbert 1999; Silvestre and Humbert 2000). Studying the differences in the genes thought to be involved in ivermectin resistance in the IVS and IVR worms will help identify possible markers for ivermectin resistance, which could be used in the development of sensitive diagnostic tests.

Acknowledgements

The authors thank G. Bingham and D. Eshelby for technical assistance. This study was funded by Fort Dodge Animal Health and NSERC, Canada.

- Anziani O. S., Zimmermann G., Guglielmone A. A., Vazquez R. and Suarez V. 2001.
 Avermectin resistance in Cooperia pectinata in cattle in Argentina. *Vet Rec* 149, 58-59.
- Arena J. P., Liu K. K., Paress P. S., Frazier E. G., Cully D. F., Mrozik H. and Schaeffer J.
 M. 1995. The mechanism of action of avermectins in Caenorhabditis elegans: correlation between activation of glutamate-sensitive chloride current, membrane binding, and biological activity. *J Parasitol* 81, 286-294.
- Armour J., Bairden K., Holmes P. H., Parkins J. J., Ploeger H., Salman S. K. and McWilliam P. N. 1987. Pathophysiological and parasitological studies on Cooperia oncophora infections in calves. *Res Vet Sci* 42, 373-381.
- Benz G. W., Roncalli R. A. and Gross S. J. 1989. Ivermectin and Abamectin. Ed WC Campbell. p 215. New York, Springer-Verlag.
- Bisset S. A., Brunsdon R. V., Forbes S. 1990. Efficacy of a topical formulation of ivermectin against naturally acquired gastrointestinal nematodes in weaner cattle. *New Zealand Vet J* 38, 4-6.
- Borgsteede F. H. and Hendriks J. 1979. Experimental infections with Cooperia oncophora (Railliet, 1918) in calves. Results of single infections with two graded dose levels of larvae. *Parasitology* **78**, 331-342.
- Bowman D. D., Lynn R. C., Eberhard M. L. and Georgi J. R. (2003) Georgis' parasitology for veterinarians, 8th Edition, pp ix, 422. Saunders, St. Louis, Mo.

- Claerebout E., Dorny P., Agneessens J., Demeulenaere D. and Vercruysse J. 1999. The effect of first season chemoprophylaxis in calves on second season pasture contamination and acquired resistance and resilience to gastrointestinal nematodes. *Vet Parasitol* **80**, 289-301.
- Coles G. C., Stafford K. A. and MacKay P. H. 1998. Ivermectin-resistant Cooperia species from calves on a farm in Somerset. *Vet Rec* 142, 255-256.
- Coles G. C., Watson C. L. and Anziani O. S. 2001. Ivermectin-resistant Cooperia in cattle. *Vet Rec* 148, 283-284.
- Coles G. C., Bauer C., Borgsteede F. H., Geerts S., Klei T. R., Taylor M. A. and Waller
 P. J. 1992. World Association for the Advancement of Veterinary Parasitology (W.A.A.V.P.) methods for the detection of anthelmintic resistance in nematodes of veterinary importance. *Vet Parasitol* 44, 35-44.
- Coop R. L., Sykes A. R. and Angus K. W. 1979. The pathogenicity of daily intakes of *Cooperia oncophora* larvae in growing calves. *Vet Parasitol* **5**, 261-269.
- Cully D. F., Vassilatis D. K., Liu K. K., Paress P. S., Van der Ploeg L. H., Schaeffer J. M. and Arena J. P. 1994. Cloning of an avermectin-sensitive glutamate-gated chloride channel from Caenorhabditis elegans. *Nature* **371**, 707-711.
- Dent J. A., Smith M. M., Vassilatis D. K. and Avery L. 2000. The genetics of ivermectin resistance in Caenorhabditis elegans. *Proc Natl Acad Sci U S A* **97**, 2674-2679.
- Elard L. and Humbert J. F. 1999. Importance of the mutation of amino acid 200 of the isotype 1 beta-tubulin gene in the benzimidazole resistance of the small-ruminant parasite Teladorsagia circumcincta. *Parasitol Res* **85**, 452-456.

- Familton A. S., Mason P. and Coles G. C. 2001. Anthelmintic-resistant Cooperia species in cattle. *Vet Rec* 149, 719-720.
- Feng X. P., Hayashi J., Beech R. N. and Prichard R. K. 2002. Study of the nematode putative GABA type-A receptor subunits: evidence for modulation by ivermectin. J Neurochem 83, 870-878.
- Fiel C. A., Saumell C. A., Steffan P. E. and Rodriguez E. M. 2001. Resistance of Cooperia to ivermectin treatments in grazing cattle of the Humid Pampa, Argentina. *Vet Parasitol* 97, 211-217.
- Holden-Dye L. and Walker R. J. 1990. Avermectin and avermectin derivatives are antagonists at the 4-aminobutyric acid (GABA) receptor on the somatic muscle cells of Ascaris; is this the site of anthelmintic action? *Parasitology* **101 Pt 2**, 265-271.
- Kwa M. S., Veenstra J. G. and Roos M. H. 1994. Benzimidazole resistance in Haemonchus contortus is correlated with a conserved mutation at amino acid 200 in beta-tubulin isotype 1. *Mol Biochem Parasitol* **63**, 299-303.
- Kwa M. S., Kooyman F. N., Boersema J. H. and Roos M. H. 1993. Effect of selection for benzimidazole resistance in Haemonchus contortus on beta-tubulin isotype 1 and isotype 2 genes. *Biochem Biophys Res Commun* 191, 413-419.
- Kwa M. S., Veenstra J. G., Van Dijk M. and Roos M. H. 1995. Beta-tubulin genes from the parasitic nematode Haemonchus contortus modulate drug resistance in Caenorhabditis elegans. *J Mol Biol* **246**, 500-510.
- Martin P. J., Anderson N. and Jarrett R. G. 1989. Detecting benzimidazole resistance with faecal egg count reduction tests and in vitro assays. *Aust Vet J* 66, 236-240.

- Martin R. J. 1996. An electrophysiological preparation of Ascaris suum pharyngeal muscle reveals a glutamate-gated chloride channel sensitive to the avermectin analogue, milbemycin D. *Parasitology* **112** (**Pt 2**), 247-252.
- McKenna P. B. 1995. Topically applied ivermectin and *Cooperia* infections in cattle. New Zealand Vet J 43, 44.
- Ministry of Agriculture Fisheries and Food. Great Britain. 1986. Manual of veterinary parasitological laboratory techniques. p 160. H.M. Stationery Off., London.
- Sangster N. C. 1999. Anthelmintic resistance: past, present and future. *Int J Parasitol* **29**, 115-124; discussion 137-118.
- Silvestre A. and Humbert J. F. 2000. A molecular tool for species identification and benzimidazole resistance diagnosis in larval communities of small ruminant parasites. *Exp Parasitol* **95**, 271-276.
- Slotved H. C., Barnes E. H., Bjorn H., Christensen C. M., Eriksen L., Roepstorff A. and Nansen P. 1996. Recovery of Oesophagostomum dentatum from pigs by isolation of parasites migrating from large intestinal contents embedded in agar-gel. *Vet Parasitol* 63, 237-245.
- Taylor M. A., Hunt K. R. and Goodyear K. L. 2002. Anthelmintic resistance detection methods. *Vet Parasitol* **103**, 183-194.
- Vermunt J. J., West D. M. and Pomroy W. E. 1995. Multiple resistance to ivermectin and oxfendazole in Cooperia species of cattle in New Zealand. *Vet Rec* 137, 43-45.
- Vermunt J. J., West D. M. and Pomroy W. E. 1996. Inefficacy of moxidectin and doramectin against ivermectin-resistant *Cooperia* spp. of cattle in New Zealand. *New Zealand Vet J* 44, 188-193.

West D. M., Vermunt J. J., Pomroy W. E. and Bentall H. P. 1994. Inefficacy of ivermectin against *Cooperia spp*. infection in cattle. *New Zealand Vet J* 42, 192-193.

In the previous chapter, we confirmed that the two isolates used in this study, IVS and IVR, differ in their sensitivity to ivermectin. Based on studies in *C. elegans* and *H. contortus*, the glutamate-gated chloride channels are thought to be major targets for ivermectin. To determine their role in ivermectin resistance in *C. oncophora*, we first cloned two full-length GluCl cDNAs from this nematode, since these sequences have not previously been reported. We then analyzed the genetic variability of the GluCl genes in IVS and IVR worms. The results are presented in Chapter 3.

CHAPTER 3

Genetic variability of glutamate gated chloride channel genes in ivermectinsusceptible and -resistant strains of *Cooperia oncophora*

Annete I. Njue and Roger K. Prichard

Submitted, Parasitology

Abstract

The glutamate-gated chloride channels (GluCls) are members of the ligand-gated ion channel (LGIC) superfamily that are thought to be involved in ivermectin's mode of action and mechanism of resistance. Using reverse-transcriptase PCR techniques, two full-length GluCl cDNAs, encoding GluCl α 3 and GluCl β subunits, were cloned from *Cooperia oncophora*, a nematode parasite of cattle. The two sequences show a high degree of identity to similar subunits from other nematodes. The *C. oncophora* GluCl α 3 subunit is most closely related to the *Haemonchus contortus* GluCl α 3B subunit, while *C. oncophora* GluCl β subunit shares high sequence identity with the *H. contortus* GluCl β subunit. Using single-strand conformation polymorphism, the genetic variability of these two genes was analyzed in ivermectin-susceptible and -resistant isolates of *C. oncophora*. Statistical analysis revealed an association between the *C. oncophora* GluCl α 3 gene and ivermectin resistance. No such association was seen with the *GluCl\beta* gene.

3.1 Introduction

Gastrointestinal helminth infections can severely limit the productivity of domestic ruminants, causing serious economic losses even at sub-clinical infection levels (Le Jambre 1993). Control relies heavily on the use of anthelmintics, among them ivermectin. Unfortunately, the development of resistance threatens the usefulness of these drugs. While resistance is primarily a problem in sheep and goats, reports of its occurrence in cattle, though not as common, are emerging (Williams 1997; Coles 2002a, b). Because they are genetically diverse, parasitic helminths are able to respond to selective pressure (Grant 1994; Otsen et al. 2001). Parasites carrying a resistance gene or genes are able to survive drug treatment and pass on their resistance genes to their offspring (Prichard 1990; Jackson 1993; Jackson and Coop 2000). Over time, the frequency of resistant individuals in the population increases, resulting in treatment failure.

Ivermectin resistance involving *Cooperia* species has been reported by several authors (Vermunt et al. 1995; Coles et al. 1998; Anziani et al. 2001; Coles et al. 2001; Familton et al. 2001; Fiel et al. 2001), highlighting the need for research into the mechanisms of resistance with a view to monitoring and limiting its spread. Studies on the mode of action of ivermectin have focused largely on the free- living *Caenorhabditis elegans*, where the GluCls have been identified as major targets (Arena et al. 1992; Cully et al. 1994; Arena et al. 1995; Dent et al. 1997; Dent et al. 2000). These channels have also been identified in parasitic nematodes, including *Haemonchus contortus* (Delany et al. 1998; Forrester et al. 1999; Jagannathan et al. 1999) and *Ascaris suum* (Martin 1996; Jagannathan et al. 1999). As members of the LGIC superfamily, GluCls are thought to be

hetero-pentameric in structure, with the five membrane-spanning subunits assembled around a central pore (Barnard 1996). Binding of ivermectin opens the channel, which is permeable to chloride ions, leading to inhibitory action in the nervous system (Arena et al. 1995; Martin 1996; Dent et al. 1997). Ivermectin also exerts its effects on other members of the LGIC superfamily, including GABA- gated chloride channels (Holden-Dye and Walker 1990; Feng et al. 2002) and nicotinic acetylcholine receptors (Krause et al. 1998). As well, ivermectin is a substrate for P-glycoproteins (P-gps), transmembrane proteins involved in transporting compounds across membranes (Schinkel et al. 1994; Pouliot et al. 1997).

Ivermectin resistance is thought to involve multiple mechanisms (Gill et al. 1998; Prichard 2001). Genetic variability studies suggest a role for P-gp in ivermectin resistance in *Haemonchus contortus* (Blackhall et al. 1998b; Sangster et al. 1999). Xu et al. (1998) have also shown that P-gp is over-expressed in ivermectin-selected strains of *H. contortus*. As well, selection at an α subunit GluCl gene is associated with ivermectin resistance in *H. contortus* (Blackhall et al. 1998a). In *C. elegans*, simultaneous mutations in three α -subunit GluCl genes results in high-level resistance (Dent et al. 2000). Whether the GluCls also play a role in ivermectin resistance in *C. oncophora* is not known. To determine this, we have cloned two GluCl subunit cDNAs (Co GluCl α 3 and Co GluCl β) from this nematode, and examined their genetic variability in ivermectinsusceptible (IVS) and -resistant (IVR) worms. The results suggest a role for Co GluCl α 3, but not GluCl β , in ivermectin resistance.

3.2 Materials and Methods

3.2.1 Parasites

IVS and IVR *C. oncophora* isolates were provided by Dr. Coles, University of Bristol, UK. The IVS isolate was passaged through calves without drug treatment at the Weybridge Experimental Station, UK, while IVR is an ivermectin-resistant field isolate (Coles et al. 1998). The parasites were maintained at the Macdonald Campus (McGill University) farm by passaging through male Holstein calves every four months. Adult *C. oncophora* were collected live at necropsy from the small intestine, and stored in liquid nitrogen until used for RNA and DNA extraction.

3.2.2 Isolation of two full-length C. oncophora GluCl cDNAs using RT-PCR

Using an oligo-dT primer (GibcoBRL) and murine Moloney leukemia virus reverse transcriptase (GibcoBRL), first strand cDNA was synthesized from total RNA isolated from bulk adult worms. The cDNA served as template for the initial amplification of a GluCl fragment using degenerate primers and the Advantage 2 PCR kit (ClonTech). For a nested PCR approach, four degenerate primers were designed based on an alignment of known GluCl sequences from *H. contortus* and *C. elegans*. In the first-round PCR reaction, the primers used were: outer sense primer 5' TGGATGCCNGAYACNTT 3' and the outer antisense primer 5' CCARTANRRDATRTTRA 3'. The PCR product from the first round was then used as template for a second amplification using the nested primers 5' ATHGAYAARCCNAAYGT 3' (sense) and 5' GCNCCRAADATRAANGC 3' (antisense). For both rounds, amplification conditions were 94°C for 30 s to denature, and 30 cycles of 94°C for 20s, 50°C for 30s, and 72°C for 1 min, followed by a final

extension at 72°C for 5min. PCR products from the second round were examined on a 1% agarose gel stained with 0.5 μ g/ml ethidium bromide, and the fragment of the expected size (~550 bp) was purified using the Nucleospin Gel Extraction Kit (ClonTech). The purified PCR products were cloned into a TA vector (Invitrogen) and sequenced. Using the sequence information of this fragment, gene-specific primers were designed for 5' and 3' RACE reactions.

For the 5' RACE reaction, the Advantage 2 cDNA kit (ClonTech) was used. Two (5' 5' RACE2 primers, 5' RACE1 and 5' RACE 5' CGCCGATCCACACGTCCACCGCCTTTAT 3' and CGAGCCAGAATGAAACCCAAGAGACGAC 3', respectively) were designed and used in a semi-nested PCR reaction with the nematode splice leader sequence SL1 (5' GGTTTAATTACCCAAGTTTGAG 3'). Amplification conditions were 94°C for 1 min, followed by 30 cycles of 94°C for 30s, 54°C for 30s, and 72°C for 1 min, followed by a final extension at 72°C for 5min. The fragment obtained after the second round of PCR was purified, ligated into a TA cloning vector (Invitrogen) and subsequently sequenced from both directions with M13 forward and reverse vector primers.

To isolate the 3' end of the *C. oncophora* GluCl cDNA, the Marathon cDNA Amplification kit (ClonTech) was used. Adaptor-ligated double-stranded cDNA was prepared according to the manufacturer's recommendations and used as template. Two gene-specific sense primers 3' RACE1 (5' CTGGTGTTGTCCTGCCCGATGTCGTTGG 3') and 3' RACE2 (5' CCAGCAAAACCAATACGGGAGAATACAG 3') were used in a nested PCR reaction with the antisense adaptor primers AP1 and AP2, respectively. A

touchdown PCR was employed as recommended by the manufacturer. The resulting product was purified, cloned and sequenced from both directions.

A second full-length GluCl cDNA, encoding a β subunit, was also isolated from *C. oncophora* using the same procedures outlined above, except for the primers. The degenerate primers used for the initial isolation were designed based upon an alignment of the amino acid sequences of GluCl β subunits from *H. contortus* (Delany et al. 1998) and *C. elegans* (Cully et al. 1994). Four degenerate primers were used in a nested PCR approach. In the first round of PCR, the sense and antisense primers used were 5' GTYCGKGTYAAYATYATGAT 3' and 5' GACRAAYGCGTATTCSAGMA 3', respectively. The first-round reaction was then used as template for further amplification using the nested primers 5' GAYGTMGTYAAYATGGARTA 3' (sense) and 5' TTGRCASGCTCCRAGCCASA 3' (antisense). The 736-bp product from the second round was cloned and sequenced, and gene-specific primers were then designed for 5' and 3' RACE reactions.

To amplify the 5' end of the GluCl β cDNA, two antisense primers, 5' RACE1 (5' GGTACAGTGAGAAACTTTGGAGGG 3') and 5' RACE2 (5' CATATGCCAGACGCGAATCGAGCC 3'), were used in a semi-nested PCR reaction with SL1 and Advantage 2 cDNA polymerase (ClonTech). For the 3' RACE reactions, two sense primers, 3' RACE1 (5' AGCAGCCGAATTAGTATCACGAGC 3') and 3' RACE2 (5' CCAGTGCAGCTCAAACCGGGCGTC 3') were used with the antisense adaptor primers AP1 and AP2 in a nested PCR reaction using the Marathon cDNA amplification kit (ClonTech).

3.2.3 PCR amplification of genomic DNA from individual male worms for single-strand conformation polymorphism (SSCP) analysis

Genomic DNA was isolated from individual adult male worms as described by (Beech et al. 1994). To ensure that the genomic DNA sample obtained represented one worm, only male worms were used. For the first GluCl subunit, 150 genomic DNA samples were examined (75 from each group), while 160 samples (80 from each group) were tested for GluClß subunit. The DNA samples were amplified in a standard PCR reaction using Taq polymerase (GibcoBRL) with the gene specific primers listed in Table 1. All primers were exonic, spanning a predicted intron in both genes (Table 1). The primers were designed using the OLIGO® primer analysis software (Version 4.06). The intron position was predicted based on the C. elegans GluCl cosmid sequence (Accession Number U97196). All PCR amplifications were carried out in a 25-µl reaction volume using 2 µl (~2 ng) genomic DNA from individual worms as template, 3 mM MgCl₂, 0.2 mM dNTPs, 0.5 U Taq polymerase and 0.4 µM of each primer. Negative (no DNA) controls were included. PCR conditions were optimized individually to each region to ensure specificity. For the first GluCl subunit gene, the thermal cycling conditions included an initial incubation at 94°C for 4 min, followed by 35 cycles of 94°C for 30 sec, 48°C for 30sec, and 72°C for 1 min, with a final extension at 72°C for 5 min. The same conditions were used to amplify a fragment of the C. oncophora GluClß gene from individual worms, except that the annealing temperature was 53°C. Prior to SSCP, a 5 µl aliquot of each PCR product was checked by agarose gel electrophoresis to confirm the size and specificity of the products.

Table 1. Primers used to amplify fragments of Co $GluCl\alpha$ and Co $GluCl\beta$ genes from individual male worms for SSCP. The primers were exonic, and the position of the intron was predicted based on a *C. elegans* GluCl cosmid sequence (Accesssion Number U97196).

Gene	Primer	Sequence	Intron	Fragment
		(5' -3')	size (bp)	size (bp)
Co GluCl a	Sense	TGGATCGACAAAATTGCCT	93	228
	Antisense	ACTTGGTAACGACATCTTG		
Co GluCl _β	Sense	AGCTCGTGCCACATGCAGC	68	278
	Antisense	ATTGGTGTGACTAGTGCAG		

3.2.4 SSCP analysis

For SSCP screening of the first GluCl subunit gene samples, 1 μ l of each PCR product was mixed with 15 μ l of loading buffer containing 95% formamide, 10 mM NaOH, 0.25% bromophenol blue, and 0.25% xylene cyanol. The samples were then denatured at 95°C for 5 min, and immediately placed on ice to stabilize the single strands of DNA. 10 μ l of each sample were loaded onto a 15% nondenaturing polyacrylamide gel and electrophoresed in 1XTBE for 18 hours at room temperature and 110V. The conditions for SSCP electrophoresis were standardized for the optimal resolution of bands. This involved testing gels of different polyacrylamide concentrations (8%, 12%, 15%, 20%), as well as voltage settings, product:dye ratios, and gel running times. For the GluClβ gene, the PCR samples were mixed with the loading dye in the same product: dye ratio of 1:15, and electrophoresed on a 12% polyacrylamide gel for 24 hrs at 90V and room temperature. Following electrophoresis, the gels were stained with ethidium bromide and visualized using the Biorad Molecular Imager FX with its corresponding Quantity One (Version 4.2.1) software. Alleles were identified based on their unique banding patterns. A chi-square analysis was performed to test for differences in allele frequencies between the two groups, IVS and IVR. As well, genotypic frequencies for both groups were analyzed for Hardy-Weinberg equilibrium using a chi-square test.

3.2.5 Sequence analysis

All alleles were investigated by direct sequencing. PCR products amplified from genomic DNA of individual worms were purified using the Nucleospin Gel Extraction kit (ClonTech) and sequenced using PCR primers. Nucleic acid sequences were analyzed using MacVector 7.0 software (Oxford Molecular Groups, England). The exact size and position of the intron was determined by comparative alignment of the alleles with the corresponding full-length *C. oncophora* GluCl cDNA sequences.

C. oncophora GluCl predicted proteins were aligned with other GluCl sequences using CLUSTALW on the SDSC Biology Workbench (Version 3.2). Phylogenetic analyses were then carried out using MEGA Version 2.1 (Kumar et al. 2001). The strength of the tree nodes was assessed by bootstrap resampling.

3.3 Results

Using degenerate primers and RT-PCR, two partial cDNA fragments encoding different GluCls were obtained from *C. oncophora*. 5' and 3' RACE techniques were then used to amplify the full-length sequences. The spliced leader sequence SL1, which is found on

many nematode mRNAs, facilitated isolation of the 5' end. The two cDNAs were 1818 and 1480 nucleotides long, with short 5' untranslated regions of 97 and 12 nucleotides, respectively. The predicted protein sequences were 438 and 432 amino acids (aa) long, respectively. A database search showed the 438-aa sequence to have highest identity to the H. contortus and C. elegans GluCla3B subunits (88% and 82% identity, respectively), while the 432-aa long predicted protein sequence showed 90% and 76% identity to H. contortus and C. elegans GluClß subunits (Table 2). Because of the high homology to GluCla3 and GluCl β subunit classes, the two cDNAs cloned from C. oncophora were assigned to these groups, and are referred to as C. oncophora (Co) GluCla3 and Co GluClB. These two sequences shared 54% identity at the amino acid level. A phylogenetic tree constructed by the Neighbor-Joining method placed Co GluCla3 predicted protein with GluCla3B sequences from H. contortus and C. elegans, while CoGluClß grouped with H. contortus and C. elegans GluClß sequences (Figure 1). These groupings were highly supported by bootstrap analysis. The nucleotide sequence data for Co GluCla3 and GluClB have been submitted to GenBank under Accession Numbers AY372756 and AY372757, respectively (Appendices B.1 and B.2, respectively).

Table 2. Amino acid sequence identities of the Co GluCl α 3 and GluCl β subunits with other GluCl sequences reported in GenBank. Partial sequences are indicated by an asterisk. Highest sequence identity to a different full-length sequence is shown in bold type.

	C. oncophora GluCla3	C. oncophora GluClβ
	(438 aa)	(432 aa)
C. oncophora GluClβ	54	-
H. contortus GluCla3B	88	51
C. elegans GluCla3B	82	51
*O. volvulus GluCIX	88	52
*D. immitis GluClX	85	56
*A. suum gbr-2	76	44
H. contortus GluClo3A	74	44
C. elegans GluCla3A	71	44
C. elegans glc-3	69	55
H. contortus GluCla	63	42
D. melanogaster GluCl	59	40
H. contortus GluClβ	55	90
C. elegans GluClβ	54	76
C. elegans GluCla2A	53	46
C. elegans GluCla2B	53	46
C. elegans GluCla1	52	45



Figure 1. Neighbor-joining tree showing relationships of the *C. oncophora* GluCl subunits with full-length GluCls from *H. contortus* and *C. elegans*. The two *C. oncophora* sequences are highlighted in bold. The Co GluCl α 3 sequence is most closely related to GluCl α 3B sequences from *H. contortus* and *C. elegans* (100% bootstrap value). The Co GluCl β subunit groups with the two β subunits from *H. contortus* and *C. elegans* (100% bootstrap value). The Co GluCl β subunit groups with the two β subunits from *H. contortus* and *C. elegans*.



An alignment of the C. oncophora GluCla3 polypeptide with GluCla3B sequences from H. contortus and C. elegans is shown in Figure 2A, while an alignment of GluCl β polypeptide sequences from C. oncophora, H. contortus and C. elegans is shown in Figure 2B. The C. oncophora GluCla3 and GluClβ predicted protein sequences contain features common to ligand-gated ion channel subunits- a long N-terminal extracellular domain containing four cysteine residues, four predicted membrane-spanning domains (TM1-4), a long intracellular loop between TM3 and TM4, and a very short C-terminal extracellular domain. The estimated molecular weights of the Co GluCl α 3 and GluCl β predicted proteins were 50,667 and 49,651 respectively. Putative signal peptides were identified at the amino termini of both sequences, and these were predicted to be cleaved between amino acids 30 and 31 for Co GluCla3 (Figure 2A), and 16 and 17 for Co GluClß (Figure 2B). Co GluCla3 contains one predicted N-linked glycosylation site at amino-acid residue 57, as well as consensus sites for phosphorylation by casein kinase II and protein kinase C in the intracellular loop between TM's 3 and 4 (Figure 2A). These are conserved in H. contortus and C. elegans GluCla3B. The Co GluClß sequence contains two conserved glycosylation sites at asparagines residues 52 and 209, and phosphorylation sites for protein kinase C and casein kinase II at amino acid residues 337 and 392 respectively (Figure 2B).

Figure 2. Alignment of GluClo3 (A) and GluCl β (B) deduced amino acid sequences of *C. oncophora, H. contortus*, and *C. elegans*. Completely conserved residues are shaded grey. Transmembrane domains are indicated (TM1- TM4). Predicted phosphorylation (protein kinase C (•) and casein kinase 2 (\circ)) and N- linked glycosylation sites (\diamond) are shown. The glycosylation and phosphorylation sites are conserved among similar subunits. The putative signal peptide cleavage sites are indicated by (\mathbf{v}). The four extracellular cysteine residues are indicated by ($\mathbf{\bullet}$). The first two cysteines, separated by 13 residues, are common to all members of the ligand-gated ion channel superfamily, and this region forms the neurotransmitter signature sequence. The second cysteine pair, separated by 10 residues, is unique to glutamate-, glycine- and histamine- gated chloride channel subunits.
Figure	2A
--------	----

C. ONCOPHORA GluClα3	1	MATSVPLATRI GPI LALI CI VITI ISTVEGKRKLKEQEI I QRI LNNYDWRVRPRGLNAS	SW 60
H. CONTORTUS GIUCIQ3B	1	MANSVPLATRI GPMLALI OTVSTI MSAVEAKRKLKEQEI I ORI LNNYDWRVRPRGLNA	SW 60
C. ELEGANS GIUCIO3B	1	MWHYRLTTLLLIISLIHSIRAKRKLKEQEIIQRILKDYDWRVRPRGMNA	FW 51
	~		
H CONTORTUS Glucio3B	61	PDT GGPVL VTVNT YL HST SKT DDVNMEYSAGF I FHEEWVDAHLAYGHFEDES- I EVPPI	- V 119
C ELEGANS GluCla3B	50	PDT GGPVLVTVNTVLRST SKI DDVNMEVSALIFI FREEWVDARLAVGRFEDES-JIEVPP	- V 119
	52	PUIGGPVLVIVNI TLRSISKI DUVNMETSAUFIFHEEWIUGHLATEHTEESGUIEVPPI	<u>- V</u>
			.
C. ONCOPHORA GluCla3	120	VI ATSENADOSOOI WMPDTEEONEKEABBHI I DKPNVLI BLHKDGSLLVSVBLSLVLS	P 179
H. CONTORTUS GluCla3B	120	VLATSENADOSOOI WMPDTFFONEKEABBHLI DKPNVLI BI HKDGSI I YSVBI SI VI SI	.P 179
C. ELEGANS GluClo3B	112	VLATSENADOSOOI WMPDTFFONEKEARRHLI DKPNVLI BI HKNGOI LYSVBLSLVLS	CP 171
		•	۲
C. ONCOPHORA GluClα3	180	MSLEFYPLDRQNCLI DLASYAYTTQDI KYEWKEQNPVQQKDGLRQSLPSFELQDVVTK	YC 239
H. CONTORTUS GluClα3B	180	MSLEFYPLDRQNCLIDLASYAYTTQDIKYEWKEQNPVQQKDGLRQSLPSFELQDVVTK	Y C 239
C. ELEGANS GluCla3B	172	MSI FEYPI DRONCLI DI ASVAVITODI KYEWKEKKIPI DOKOGI ROSI PSEFI ODVVID	Y C 231
		more in condition benoticit in the medical and the second of the second	
		TM1 TM2	 !
	940		
C. ONCOPHORA GluCla3	240	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLIQLYIPCIMLVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLIQLYIPCIMLUVSWVSFWLDKDAVPARVSLG	VT 299
C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B	240 240 232	TM1 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLUVVSWVSFWLDKDAVPARVSLG TSUTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVSWVSFWLDKDAVPARVSLG	VT 299 VT 299 VT 299 VT 291
C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B	240 240 232	TM1 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLLVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG	VT 299 VT 299 VT 299 VT 291
C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B	240 240 232	TM1 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLLVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3	VT 299 VT 299 VT 299 VT 291
C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B C. ONCOPHORA GluCla3	240 240 232 300	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLUVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3 TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE	VT 299 VT 299 VT 291 VT 291
C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B	240 240 232 300 300	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSL TSLTNTGEYSCARVVLRLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3 TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE	VT 299 VT 299 VT 291 VT 291 KK 359 KK 359
C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B	240 240 232 300 300 292	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3 TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE	VT 299 VT 299 VT 291 VT 291 KK 359 KK 359 KK 351
C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B	240 240 232 300 300 292	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3 TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NTKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE	VT 299 VT 299 VT 291 KK 359 KK 359 KK 351
C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B	240 240 232 300 300 292	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3 TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE	VT 299 VT 299 VT 291 KK 359 KK 359 KK 351
C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B C. ONCOPHORA GluCla3 H. CONTORTUS GluCla3B C. ELEGANS GluCla3B C. ONCOPHORA GluCla3	240 240 232 300 300 292 360	TM1 TM2 TSKTNTGEYSCARVKLLL RREYSYYLI QLYI PCI ML VVVSWVSFWL DKDAVPARVSLG TSKTNTGEYSCARVKLLL RREYSYYLI QLYI PCI ML VVVSWVSFWL DKDAVPARVSLG TSLTNTGEYSCARVKLLL RREYSYYLI QLYI PCI ML VVVSWVSFWL DKDAVPARVSLG TSLTNTGEYSCARVKLLL RREYSYYLI QLYI PCI ML VVVSWVSFWL DKDAVPARVSLG TM3 TM3 TM3 TM4 O TM4 M4 KKTRL DDCVCPSERPAL RL DL ST FRR GWTPLN- RL DVL GRNADL SRRVDL MSRI TF	VT 299 VT 299 VT 291 KK 359 KK 359 KK 351
 C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B C. ONCOPHORA GluClα3B C. ONCOPHORA GluClα3B H. CONTORTUS GluClα3B H. CONTORTUS GluClα3B 	240 240 232 300 300 292 360 360	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM2 TM3 TLTMTT QASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TM3 TM4 TM4 TM4 TM4 TM4 TM4 TM4 TM4 TM4 C TM4 KKTRLDDCVCPSERPALRLDLSTFRRRGWTPLN- RLLDVLGRNADLSRRVDLMSRITF KKTRLDDCVCPSERPALRLDLSTFRRRGWTPLN- RLLDMLGRNADLSRRVDLMSRITF KKTRLDDCVCPSERPALRLDLSTFRRRGWTPLN- RLLDMLGRNADLSRRVDLMSRITF	VT 299 VT 299 VT 291 VT 291 KK 359 KK 359 KK 351 PT 418 PS 418
C. ONCOPHORA GIUCIa3 H. CONTORTUS GIUCIa3B C. ELEGANS GIUCIa3B C. ONCOPHORA GIUCIa3 H. CONTORTUS GIUCIa3B C. ELEGANS GIUCIa3B C. ONCOPHORA GIUCIa3 H. CONTORTUS GIUCIa3B C. ELEGANS GIUCIa3B	240 240 232 300 300 292 360 360 352	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3 TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TLLTMTTQASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TM4 KKTRLDDCVCPSERPALRLDLSTFRRRGWTPLN- RLLDVLGRNADLSRVDLMSRITF KKTRLDDCVCPSERPALRLDLSNYRRGWTPLN- RLLDVLGRNADLSRVDLMSRITF KKTRI DDCVCPSDRPPLRLDLSAYRSVKRLPI I KHI SEI LSTNI DI SRRVDLMSRITF	VT 299 VT 299 VT 291 VT 291 KK 359 KK 359 KK 351 PT 418 PS 418 PL 411
C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B	240 240 232 300 300 292 360 360 352	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVVLRLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3 TLTMTTOASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TM3 TLTMTTOASGI NSKLPPVSYI KAVDVWI GVCLAFI FGALLEYAVVNYYGRKEFLRKE TM4 TM4 TM4 C TM4 KKTRLDDCVCPSERPALRLDLSTFRRRGWTPLN- RLLDVLGRNADLSRVDLMSRITF KKTRLDDCVCPSERPALRLDLSNYRRGWTPLN- RLLDMLGRNADLSRVDLMSRITF KKTRLDCVCPSERPALRLDLSNYRRGWTPLN- RLLDMLGRNADLSRVDLMSRITF KKTRLDCVCPSERPALRLDLSNYRRGWTPLN- RLLDMLGRNADLSRVDLMSRITF KKTRLDISAYRSVKRLPITKREWTPLN-	VT 299 VT 299 VT 291 VT 291 KK 359 KK 359 KK 351 PT 418 PS 418 PL 411
 C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B C. ONCOPHORA GluClα3 H. CONTORTUS GluClα3B C. ELEGANS GluClα3B C. ONCOPHORA GluClα3B C. ELEGANS GluClα3B C. ELEGANS GluClα3B C. ONCOPHORA GluClα33 	240 240 232 300 300 292 360 360 352 419	Image: The second of the se	VT 299 VT 299 VT 291 VT 291 KK 359 KK 359 KK 351 PT 418 PS 418 PL 411
C. ONCOPHORA GIUCIA3 H. CONTORTUS GIUCIA3B C. ELEGANS GIUCIA3B C. ONCOPHORA GIUCIA3 H. CONTORTUS GIUCIA3B C. ELEGANS GIUCIA3B C. ONCOPHORA GIUCIA3 H. CONTORTUS GIUCIA3B C. ELEGANS GIUCIA3B	240 240 232 300 300 292 360 360 352 419 419	TM1 TM2 TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSKTNTGEYSCARVKLLLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TSLTNTGEYSCARVMLRLRREYSYYLI QLYI PCI MLVVVSWVSFWLDKDAVPARVSLG TM3 TM3 TM3 TM3 TM3 TM3 TM3 TM4 TM3 TM3 TM4 TM3 TM3 TM3 TM3 TM4 C TM3 TM3 TM3 TM3 TM3 TM3 TM4 O TM4 KAVDVW GVCLAFI FGALLEYAVVNYYGRKEFL RKE TM4 KKTRLDDCVCPSERPALRLDLSTFRRGWTPLN- RLLDVLGRNADLSREVDLMSRITF KKTRLDDCVCPSERPALRLDLSNYRRGWTPLN- RLLDVLGRNADLSREVDLMSRITF KKTRLDCVCPSDRPPLRLDLSAYRSVKRLPI I KRI SEI LSTNI DI SREVDLMSRLTF <th>VT 299 VT 299 VT 291 VT 291 KK 359 KK 359 KK 351 PT 418 PS 418 PL 411</th>	VT 299 VT 299 VT 291 VT 291 KK 359 KK 359 KK 351 PT 418 PS 418 PL 411
C. ONCOPHORA GIUCIA3 H. CONTORTUS GIUCIA3B C. ELEGANS GIUCIA3B C. ONCOPHORA GIUCIA3 H. CONTORTUS GIUCIA3B C. ELEGANS GIUCIA3B C. ONCOPHORA GIUCIA3 H. CONTORTUS GIUCIA3B C. ONCOPHORA GIUCIA3 H. CONTORTUS GIUCIA3B C. ELEGANS GIUCIA3B	240 240 232 300 300 292 360 360 352 419 419 419	Image: The second se	VT 299 VT 299 VT 291 VT 291 KK 359 KK 359 KK 359 KK 351 PT 418 PS 418 PL 411

Figure 2B

	V	
C. ONCOPHORA GIUCIβ 1 H. CONTORTUS GIUCIβ 1 Π. CONTORTUS GIUCIβ 1	MTFALVETVATVAVMADSSHVSRRSSGGTQEQEILNE MSQYMMVAVAAVVAVAGSSQISRRSTGGTQEQEILNE	LLSNYDMRVRPPPTNYSDPTGPV 60 LLSNYDMRVRPPPSNYSDPMGPV 60
C. ELEGANS GIUCIB 1	MATPSSESILLLLLMPVVTNGEYSMQSEQEILN/	VLUKINYDMRVRPPP-ANSSTEGAV 57
C. ONCOPHORA GIUCIβ 61 H. CONTORTUS GIUCIβ 61	TVRVNI MIRMLSKI DVVNMEYSMQLTFREQWLDSRL	AYARLGYHNPPKFLTVPHIKSNLW 120 AYAHLGYHNPPKFLTVPHIKSNLW 120
C. ELEGANS GluClβ 58	NVRVNI MI RMLSKI DVVNMEYSI QLTFREQWIDARL	YENLGFYNPPAFLTVPHVKKSLW 117
C. ONCOPHORA GIUCIB 121	I PDTEEPTEKAAHBHLI DTDNMEL BI HPDGKVI YSSI	ALSI TSSCHMOLOLYPLDLOFCDF 180
H. CONTORTUS GluClβ 121	I PDTFFPTEKAAHRHLI DTDNMFLRI HPDGKVLYSSI	N SITSSCHMQLOLYPLDLQFCDF 180
C. ELEGANS GluClβ 118	I PDTFFPTEKAAHRHLI DMENMFL RI YPDGKI LYSSI	TI SLITSSOPMALQLYPLDYQSONE 177
H CONTORTUS GIUCIS 181	IDLVSYAHTMKDI VYQWDPTAPVQLKPGVGSDLPNFQI	TNI TINDDCISHINIGSYACL RM 240
C. ELEGANS GluCIβ 178	DL VSYAHTMNDI MYEWDPSTPVQL KPGVGSDL PNF	RNYTTNADCTSHINTGSYGCL RM 237
	TM1	TM2
C. ONCOPHORA GluCIß 241	QLTLKRQFSYYLVQLYGPTTMI VIVSWSFW DMHS	TAGRVAL GVTTLLTMTTMQAAI NA 300
H. CONTORTUS GluClβ 241 C. ELEGANS GluClβ 238	QLTLKRQFSYYLVQLYGPTTMIVIVSWSFWIDMHS	TAGRVALGVTTLLTMTTMQAAINA 300 TAGRVALGVTTLLTMTTMQSAINA 297
	TM3	D
C. ONCOPHORA GluClβ 301	KLPPVSYVKVVDVWLGACQTFVFGALLEYAFVSYQD	SOROT DOAKS CAT RKAOKRRACME 360
H. CONTORTUS GluCiβ 301	KL PPVSYVKVVDVWL GACQTFVFGALLEYAFVSYQD	SOROTEOAKSRAARKAOKRRAKME 360
C. ELEGANS GIUCIP 298	KLPPVSYVKVVDVWLGACQIFVFGALLEYAFVSTQD	
	0	E 1V/14
	LISERDHYOPPCTCHLYODYEPTLIRDRLRRYFTKPDY	LPAKI DYYARFCVPLGFLAFNALY 420 LPAKI DYYARFCVPLGFLAFNALY 420
C. ELEGANS GluClβ 358	MUDAEVYQPPCTCHTFEARE-TFRDKVRRYFTKPDY	PAKI DEVARENVPLAFLAFNULY 416
C. ONCOPHORA GluClβ 421	WTSCLVMVSRLI 432	
H. CONTORTUS GluClβ 421 C. ELEGANS GluClβ 417	WTSCLVMVSRLV WMSCLI MSANASTPESLV 434	

To examine genetic variation of the Co $GluCl\alpha3$ and $GluCl\beta$ genes, genomic DNA was prepared from individual IVS and IVR male worms. Co $GluCl\alpha3$ and $GluCl\beta$ gene fragments were amplified separately and subjected to agarose gel electrophoresis. While no size variations were detected on agarose gels, SSCP revealed differences among the samples (Fig. 3).



Single-strand conformation polymorphism analysis of Co $GluCl\beta$ PCR products. (A). Agarose gel showing the quality of the PCR products derived from 10 individual male worms. M= molecular weight marker, C= No-DNA control. (B). SSCP patterns of 11 individual IVS worms at the Co $GluCl\beta$ gene. Worms shown here are homozygous BB (2 bands) or heterozygous AB (three bands). For the Co *GluClc3* gene, 75 IVS and 75 IVR individual male worms were analyzed. A total of nine different alleles (A-J) were identified (Fig. 4). Allele A was the most common allele in the IVS group, with a frequency of 0.406. The frequency of allele A in the IVR group was 0.21. Allele E was the most common allele in the IVR group, with a frequency of 0.27. This allele was also found in the IVS group, but at a lower frequency of 0.18. As with allele A, alleles C and G were found at a higher frequency in the IVS group, while alleles D, F and H were found at higher frequencies in the IVR group. Allele J was found only in the IVR group. Chi-square analysis revealed a significant difference in allele frequencies between IVS and IVR groups (p < 0.001). Genotypic frequencies did not differ significantly from Hardy-Weinberg equilibrium.



Figure 4. Co GluClo3 allele frequencies in IVS and IVR worms

For the Co $GluCl\beta$ gene, 160 worms (80 IVS and 80 IVR) were examined. Only two alleles, A and B, were detected by SSCP (Figs. 3 and 5). Allele A had frequencies of 0.137 and 0.1 in the IVS and IVR groups, respectively, while allele B had frequencies of 0.863 and 0.9 in the same respective groups. Allele frequencies were not significantly different between IVS and IVR groups, and genotypic frequencies did not differ significantly from Hardy-Weinberg equilibrium.



Figure 5. Co $GluCl\beta$ allele frequencies in IVS and IVR worms

To determine the nucleotide differences among the different alleles, amplicons representing all Co $GluCl\alpha\beta$ and $GluCl\beta$ alleles were subjected to direct sequencing. For the Co $GluCl\alpha\beta$ gene, 228 bp of sequence was obtained for each allele, with a 93-bp

intron flanked by exons. No length polymorphisms were detected. Alignment of the alleles revealed 18 polymorphic sites, distributed across the length of the sequence (Fig. 6A). Of these, 13 were singletons, while five were found two or more times among the nine allele sequences. All were single base substitutions, with 14 (77.8%) being transitions (A \leftrightarrow G, n=9; C \leftrightarrow T, n=5) and 4 (22.2%) being transversions (C \leftrightarrow G, n=2; A \leftrightarrow T, n=1; A \leftrightarrow C, n=1). 13 of the polymorphic sites were found in the intron region. Of the five nucleotide changes found in the exon, four were at the third codon position, and one was at the first codon position. All exon variants were silent mutations, consisting of single nucleotide substitutions with no amino acid changes.



Positions of polymorphic sites in the region of the Co $GluCl\alpha$ and Co $GluCl\beta$ genes analyzed. Polymorphic sites are highlighted as vertical bars. (A). 18 positions were found to be polymorphic for the Co $GluCl\alpha$ gene. Five of these were in the exon, and are marked by arrows. All were silent mutations. (B). Only one polymorphic site was identified in the Co $GluCl\beta$ gene in an exon region. The nucleotide substitution at this position did not result in amino acid change.

For the Co $GluCl\beta$ gene, 278 bp of sequence was obtained for both alleles A and B. The intron was 68 bp long. Figure 6B shows that only one polymorphic site was identified following alignment of the alleles. This nucleotide difference, a C \leftrightarrow T transition, was found in an exon region, and represented a silent mutation at the third codon position.

3.4 Discussion

GluCls are members of the 'cys-loop' class of LGICs that are found only in invertebrates (Cleland 1996). Among nematodes, full-length GluCl cDNAs have been cloned from *C. elegans* (Cully et al. 1994; Cully et al. 1996; Dent et al. 1997; Vassilatis et al. 1997; Dent et al. 2000; Horoszok et al. 2001), and *H. contortus* (Delany et al. 1998; Forrester et al. 1999; Jagannathan et al. 1999). Partial GluCl cDNA sequences have also been obtained from *A. suum* (Jagannathan et al. 1999), *Dirofilaria immitis* and *Onchocerca volvulus* (Cully et al. 1996). In the present study, two full-length GluCl cDNAs, encoding GluClc3 and GluCl β subunits, were cloned from *C. oncophora*. The Co GluClc3 predicted protein sequence showed high identity to GluClc3 sequences from *H. contortus* (88%), *O. volvulus* (88%), *C. elegans* 82%), *A. suum* (76%) and *D. immitis* (85%) (Table 2). The conservation of sequence among GluClc3 subunits from different nematodes, which was previously highlighted by Jagannathan et al. (1999), suggests that these subunits may play an important role in the function of nematode nerve cells.

The Co GluCl α 3 and GluCl β subunits exhibit features that are characteristic of cys-loop LGICs- a long hydrophobic N-terminal extracellular domain containing a pair of cysteine residues separated by 13 amino acids, and four transmembrane domains. A second pair of cysteine residues, separated by 10 amino acids, is also present in the N-

terminal extracellular region. This is conserved in GluCls, vertebrate glycine-gated chloride channels (Vassilatis et al. 1997), and the recently characterized histamine-gated chloride channels (Zheng et al. 2002).

Studies of *C. elegans* GluCls have shown that these receptors are targets of the avermectin class of endecocides (Arena et al. 1992, 1995; Cully et al. 1994, 1996; Vassilatis et al. 1997: Dent et al. 1997, 2000: Horoszok et al. 2001). When expressed in *Xenopus* oocytes, the *C. elegans* GluCl α subunit forms homomeric channels that are activated by ivermectin (Cully et al. 1994). Other *C. elegans* GluCls that are sensitive to ivermectin include AVR-15/ GluCl α 2 (Dent et al. 1997), AVR-14B/ gbr-2B/ GluCl α 3B (Dent et al. 2000) and glc-3 (Horoszok et al. 2001). While ivermectin sensitivity appears to be a feature of α -like GluCl subunits, the *C. elegans* AVR-14A subunit, which is a splice variant of the *avr-14/ GluCl\alpha3* gene, does not respond to ivermectin (Dent et al. 2000). As well, the *C. elegans* GluCl β subunit is sensitive to glutamate, but not ivermectin (Cully et al. 1994).

GluCls have also been implicated in the mechanism of resistance to ivermectin. In *C. elegans*, simultaneous mutation of three GluCl genes, *glc-1*, *avr-15* and *avr-14*, confers high-level ivermectin resistance (Dent et al. 2000). In *H. contortus*, changes in allele frequencies as a result of ivermectin selection have been observed in a GluCl gene encoding a putative α -subunit (Blackhall et al. 1998a). In the present study, genetic variability of two *C. oncophora* GluCl genes, Co *GluCl* α and Co *GluCl*, was analyzed in IVS and IVR worms using SSCP in combination with direct sequencing. Amplicons of Co *GluCl* α and Co *GluCl* β genes analyzed were ~228 and 278 bp in length, respectively, well within the high sensitivity range of 100-350 bp for SSCP (Benkwitz et

al. 1999). A significant difference in allele frequencies was observed between IVS and IVR worms at the Co *GluClc3* gene locus, suggesting that this gene may be involved in ivermectin resistance. Two alleles, E and H, were found at higher frequencies in the IVR group, while allele A had a higher frequency in the IVS group. Differences in allele frequencies between the two groups at the less variable Co *GluCl3* gene locus were not significant, and this gene may not be involved in ivermectin resistance. Similar analyses were performed on the gene encoding actin in the same group of worms. All 32 IVS and 32 IVR worms examined were heterozygous for two alleles, A and B. At the variable beta tubulin isotypes 1 and 2 loci, where ten alleles were identified, allele frequencies were not significantly different between the two groups (Njue and Prichard, in press). These findings suggest that the differences seen at the Co *GluCl3* locus were most likely due to ivermectin selection, and not population bottlenecking, which would have affected the other genes.

While a significant difference in Co $GluCl\alpha3$ allele frequencies was evident, there was no reduction in genetic variability in the IVR group. Similar results were also obtained by Blackhall et al. (1998a), who found five alleles of a putative GluCl α gene in ivermectin-selected and unselected *H. contortus*. In other studies, low levels of benzimidazole resistance did not result in loss of genetic diversity of beta tubulin isotype 1 and 2 genes (Kwa et al. 1993). However, at higher levels of benzimidazole resistance, significant reductions in genetic variability have been reported in *H. contortus*, *Trichostrongylus colubriformis* and *Teladorsagia circumcincta* (Kwa et al. 1993; Beech et al. 1994; Grant and Mascord 1996; Elard and Humbert 1999). With no detectable loss of genetic variability, selection at the Co *GluClo3* gene may be at the early stages.

For both Co *GluClc3* and *GluClβ* genes, the region analyzed by SSCP lay between the two dicysteine loops. Sequencing of the two *GluClβ* alleles revealed that they were identical at the amino acid level. As well, all nine Co *GluClc3* alleles had identical amino acid sequences, suggesting that this region may not directly be involved in ivermectin resistance. For both *GluClc3* and *GluClβ* genes, a second region, the intracytoplasmic loop, was also analyzed by SSCP and sequencing of alleles, and no amino acid changes were seen among *GluClc3* alleles or the *GluClβ* alleles identified (Appendix B). SSCP cannot detect mutations located outside the region being analyzed, and it is possible that mutations in other regions of the gene may be involved in ivermectin resistance.

The expression sites of the Co GluCl α 3 and GluCl β subunits remain to be determined. In *C. elegans*, the avr-14/ GluCl α 3 subunits, which are homologous to Co GluCl α 3, are expressed in extrapharyngeal neurons (Dent et al. 2000). Activation of extrapharyngeal neurons by ivermectin can inhibit pharyngeal pumping, an action mediated via gap junctions (Dent et al. 2000). The *C. elegans* GluCl β subunit is expressed in the pharynx (Laughton et al. 1997). In *H. contortus*, the GluCl β subunit is expressed in motor neuron commissures, and may therefore have some effect on body wall muscle (Delany et al. 1998; Portillo et al. 2003). The splice variants HcGluCl α 3A and 3B are also expressed in the motor neuron commissures (Jagannathan et al. 1999; Portillo et al. 2003). In other regions, these two splice variants have unique expression patterns (Portillo et al. 2003). HcGluCl α 3A is found in a pair of lateral neurons in the head of the worm, which are most likely amphidial neurons. HcGluCl α 3B is expressed in nerve cords, as well as three cell bodies on the pharynx. This suggests that pharyngeal

receptors containing HcGluCla3B are involved in mediating ivermectin's inhibitory effects on pharyngeal pumping.

The two isolates used in this study, IVS and IVR, were originally obtained from different locations in the UK, and their geographical separation, albeit not too great, may have contributed to the difference in allele frequencies at the Co *GluClo3* gene. However, no significant differences in allele frequencies were seen at actin and beta tubulin genes. These two genes have no known association with ivermectin's mode of action. As well, the same Co *GluClo3* alleles were found in the IVS and IVR groups, suggesting that the significant differences observed in Co *GluClo3* allele frequencies were most likely due to ivermectin selection. In *H. contortus*, selection with benzimidazoles has been shown to select for the same allelic types in independently derived field isolates, suggesting that geographical separation does not preclude relevant comparison of the allele frequencies of such isolates (Kwa et al. 1993; Beech et al. 1994). However, further work is required to determine whether there are any functional differences between IVS and IVR Co GluClo3 and GluClβ subunits. Studies are underway to characterize these subunits by expressing them in *Xenopus* oocytes.

Acknowledgements

The authors thank Dr. G. Coles for providing the parasite isolates used in this study. The technical assistance of D. Eshelby and G. Bingham is greatly appreciated. This study was supported by grants from Fort Dodge Animal Health and NSERC, Canada.

- Anziani O. S., Zimmermann G., Guglielmone A. A., Vazquez R. and Suarez V. (2001)
 Avermectin resistance in *Cooperia pectinata* in cattle in Argentina. *Vet Rec* 149, 58-59.
- Arena J. P., Liu K. K., Paress P. S., Schaeffer J. M. and Cully D. F. (1992) Expression of a glutamate-activated chloride current in *Xenopus* oocytes injected with *Caenorhabditis elegans* RNA: evidence for modulation by avermectin. *Brain Res Mol Brain Res* 15, 339-348.
- Arena J. P., Liu K. K., Paress P. S., Frazier E. G., Cully D. F., Mrozik H. and Schaeffer J.
 M. (1995) The mechanism of action of avermectins in *Caenorhabditis elegans*: correlation between activation of glutamate-sensitive chloride current, membrane binding, and biological activity. *J Parasitol* 81, 286-294.
- Barnard E. A. (1996) The transmitter-gated channels: a range of receptor types and structures. *Trends Pharmacol Sci* 17, 305-309.
- Beech R. N., Prichard R. K. and Scott M. E. (1994) Genetic variability of the beta-tubulin genes in benzimidazole-susceptible and -resistant strains of *Haemonchus contortus*. *Genetics* 138, 103-110.
- Benkwitz C., Oberdorf-Maass S. and Neyses L. (1999) Combined SSCP and heteroduplex analysis of the human plasma membrane Ca(2+)-ATPase isoform 1 in patients with essential hypertension. *Biochem Biophys Res Commun* **261**, 515-520.

- Blackhall W. J., Pouliot J. F., Prichard R. K. and Beech R. N. (1998a) *Haemonchus contortus*: selection at a glutamate-gated chloride channel gene in ivermectin- and moxidectin-selected strains. *Exp Parasitol* **90**, 42-48.
- Blackhall W. J., Liu H. Y., Xu M., Prichard R. K. and Beech R. N. (1998b) Selection at a P-glycoprotein gene in ivermectin- and moxidectin-selected strains of *Haemonchus contortus*. *Mol Biochem Parasitol* **95**, 193-201.
- Blouin M. S., Dame J. B., Tarrant A. and Courtney C. H. (1992) Unusual population genetics of a parasite nematode: mtDNA variation with and among populations. *Evolution* **46**, 470-476.
- Cleland T. A. (1996) Inhibitory glutamate receptor channels. Mol Neurobiol 13, 97-136.
- Coles G. C. (2002a) Cattle nematodes resistant to anthelmintics: why so few cases? Vet Res 33, 481-489.
- Coles G. C. (2002b) Sustainable use of anthelmintics in grazing animals. Vet Rec 151, 165-169.
- Coles G. C., Stafford K. A. and MacKay P. H. (1998) Ivermectin-resistant *Cooperia* species from calves on a farm in Somerset. *Vet Rec* 142, 255-256.
- Coles G. C., Watson C. L. and Anziani O. S. (2001) Ivermectin-resistant *Cooperia* in cattle. *Vet Rec* 148, 283-284.
- Cully D. F., Wilkinson H., Vassilatis D. K., Etter A. and Arena J. P. (1996) Molecular biology and electrophysiology of glutamate-gated chloride channels of invertebrates. *Parasitology* 113 Suppl, S191-200.



- Cully D. F., Vassilatis D. K., Liu K. K., Paress P. S., Van der Ploeg L. H., Schaeffer J. M. and Arena J. P. (1994) Cloning of an avermectin-sensitive glutamate-gated chloride channel from *Caenorhabditis elegans*. *Nature* **371**, 707-711.
- Delany N. S., Laughton D. L. and Wolstenholme A. J. (1998) Cloning and localisation of an avermectin receptor-related subunit from *Haemonchus contortus*. *Mol Biochem Parasitol* 97, 177-187.
- Dent J. A., Davis M. W. and Avery L. (1997) avr-15 encodes a chloride channel subunit that mediates inhibitory glutamatergic neurotransmission and ivermectin sensitivity in *Caenorhabditis elegans. Embo J* 16, 5867-5879.
- Dent J. A., Smith M. M., Vassilatis D. K. and Avery L. (2000) The genetics of ivermectin resistance in *Caenorhabditis elegans*. *Proc Natl Acad Sci U S A* **97**, 2674-2679.
- Elard L. and Humbert J. F. (1999) Importance of the mutation of amino acid 200 of the isotype 1 beta-tubulin gene in the benzimidazole resistance of the small-ruminant parasite *Teladorsagia circumcincta*. *Parasitol Res* **85**, 452-456.
- Familton A. S., Mason P. and Coles G. C. (2001) Anthelmintic-resistant *Cooperia* species in cattle. *Vet Rec* 149, 719-720.
- Feng X. P., Hayashi J., Beech R. N. and Prichard R. K. (2002) Study of the nematode putative GABA type-A receptor subunits: evidence for modulation by ivermectin. J Neurochem 83, 870-878.
- Fiel C. A., Saumell C. A., Steffan P. E. and Rodriguez E. M. (2001) Resistance of *Cooperia* to ivermectin treatments in grazing cattle of the Humid Pampa, Argentina. *Vet Parasitol* 97, 211-217.

- Forrester S. G., Hamdan F. F., Prichard R. K. and Beech R. N. (1999) Cloning, sequencing, and developmental expression levels of a novel glutamate-gated chloride channel homologue in the parasitic nematode *Haemonchus contortus*. *Biochem Biophys Res Commun* 254, 529-534.
- Gill J. H., Kerr C. A., Shoop W. L. and Lacey E. (1998) Evidence of multiple mechanisms of avermectin resistance in *Haemonchus contortus*- comparison of selection protocols. *Int J Parasitol* 28, 783-789.
- Grant W. N. (1994) Genetic variation in parasitic nematodes and its implications. Int J Parasitol 24, 821-830.
- Grant W. N. and Mascord L. J. (1996) Beta-tubulin gene polymorphism and benzimidazole resistance in *Trichostrongylus colubriformis*. Int J Parasitol 26, 71-77.
- Holden-Dye L. and Walker R. J. (1990) Avermectin and avermectin derivatives are antagonists at the 4-aminobutyric acid (GABA) receptor on the somatic muscle cells of Ascaris; is this the site of anthelmintic action? *Parasitology* **101 Pt 2**, 265-271.
- Horoszok L., Raymond V., Sattelle D. B. and Wolstenholme A. J. (2001) GLC-3: a novel fipronil and BIDN-sensitive, but picrotoxinin-insensitive, L-glutamate-gated chloride channel subunit from *Caenorhabditis elegans*. *Br J Pharmacol* **132**, 1247-1254.

Jackson F. (1993) Anthelmintic resistance- the state of play. Br Vet J 149, 123-138.

- Jackson F. and Coop R. L. (2000) The development of anthelmintic resistance in sheep nematodes. *Parasitology* **120 Suppl**, S95-107.
- Jagannathan S., Laughton D. L., Critten C. L., Skinner T. M., Horoszok L. and Wolstenholme A. J. (1999) Ligand-gated chloride channel subunits encoded by the

Haemonchus contortus and Ascaris suum orthologues of the Caenorhabditis elegans gbr-2 (avr-14) gene. Mol Biochem Parasitol 103, 129-140.

- Krause R. M., Buisson B., Bertrand S., Corringer P. J., Galzi J. L., Changeux J. P. and Bertrand D. (1998) Ivermectin: a positive allosteric effector of the alpha7 neuronal nicotinic acetylcholine receptor. *Mol Pharmacol* 53, 283-294.
- Kumar S., Tamura K., Jakobsen I. B. and Nei M. (2001) MEGA2: molecular evolutionary genetics analysis software. *Bioinformatics* 17, 1244-1245.
- Kwa M. S., Veenstra J. G. and Roos M. H. (1993) Molecular characterisation of betatubulin genes present in benzimidazole-resistant populations of Haemonchus contortus. *Mol Biochem Parasitol* **60**, 133-143.
- Laughton D. L., Lunt G. G. and Wolstenholme A. J. (1997) Reporter gene constructs suggest that the Caenorhabditis elegans avermectin receptor beta-subunit is expressed solely in the pharynx. *J Exp Biol* **200** (**Pt 10**), 1509-1514.
- Le Jambre L. F. (1993) Molecular variation in trichostrongylid nematodes from sheep and cattle. *Acta Trop* **53**, 331-343.
- Martin R. J. (1996) An electrophysiological preparation of *Ascaris suum* pharyngeal muscle reveals a glutamate-gated chloride channel sensitive to the avermectin analogue, milbemycin D. *Parasitology* **112** (**Pt 2**), 247-252.
- McKenna P. B. (1996) Anthelmintic resistance in cattle nematodes in New Zealand: is it increasing? *New Zealand Vet J* 44, 76.
- Njue A. I. and Prichard R. K. (2003) Cloning two full-length beta-tubulin isotype cDNAs from *Cooperia oncophora*, and screening for benzimidazole resistance-associated mutations in two field populations. *Parasitol* (in press).

- Otsen M., Hoekstra R., Plas M. E., Buntjer J. B., Lenstra J. A. and Roos M. H. (2001) Amplified fragment length polymorphism analysis of genetic diversity of *Haemonchus contortus* during selection for drug resistance. *Int J Parasitol* **31**, 1138-1143.
- Portillo V., Jagannathan S. and Wolstenholme A. J. (2003) Distribution of glutamategated chloride channel subunits in the parasitic nematode Haemonchus contortus. *J Comp Neurol* **462**, 213-222.
- Pouliot J. F., L'Heureux F., Liu Z., Prichard R. K. and Georges E. (1997) Reversal of Pglycoprotein-associated multidrug resistance by ivermectin. *Biochem Pharmacol* 53, 17-25.
- Prichard R. (2001) Genetic variability following selection of Haemonchus contortus with anthelmintics. *Trends Parasitol* 17, 445-453.
- Prichard R. K. (1990) Anthelmintic resistance in nematodes: extent, recent understanding and future directions for control and research. *Int J Parasitol* **20**, 515-523.
- Sangster N. C., Bannan S. C., Weiss A. S., Nulf S. C., Klein R. D. and Geary T. G. (1999) Haemonchus contortus: sequence heterogeneity of internucleotide binding domains from P-glycoproteins. Exp Parasitol 91, 250-257.
- Schinkel A. H., Smit J. J., van Tellingen O., Beijnen J. H., Wagenaar E., van Deemter L.,
 Mol C. A., van der Valk M. A., Robanus-Maandag E. C., te Riele H. P. and et al.
 (1994) Disruption of the mouse mdr1a P-glycoprotein gene leads to a deficiency in
 the blood-brain barrier and to increased sensitivity to drugs. *Cell* 77, 491-502.
- Vassilatis D. K., Arena J. P., Plasterk R. H., Wilkinson H. A., Schaeffer J. M., Cully D.F. and Van der Ploeg L. H. (1997) Genetic and biochemical evidence for a novel

avermectin-sensitive chloride channel in *Caenorhabditis elegans*. Isolation and characterization. *J Biol Chem* 272, 33167-33174.

- Vermunt J. J., West D. M. and Pomroy W. E. (1995) Multiple resistance to ivermectin and oxfendazole in *Cooperia* species of cattle in New Zealand. *Vet Rec* 137, 43-45.
- Vermunt J. J., West D. M. and Pomroy W. E. (1996) Inefficacy of moxidectin and doramectin against ivermectin-resistant *Cooperia* spp. of cattle in New Zealand. *New Zealand Vet J* 44, 188-193.
- Williams J. C. (1997) Anthelmintic treatment strategies: current status and future. Vet Parasitol 72, 461-470; discussion 470-467.
- Xu M., Molento M., Blackhall W., Ribeiro P., Beech R. and Prichard R. (1998) Ivermectin resistance in nematodes may be caused by alteration of P-glycoprotein homolog. *Mol Biochem Parasitol* **91**, 327-335.
- Zheng Y., Hirschberg B., Yuan J., Wang A. P., Hunt D. C., Ludmerer S. W., Schmatz D. M. and Cully D. F. (2002) Identification of two novel *Drosophila melanogaster* histamine-gated chloride channel subunits expressed in the eye. J Biol Chem 277, 2000-2005.

Results from the previous chapter demonstrate an association between the *C. oncophora* $GluCl\alpha 3$ gene and ivermectin resistance. However, no amino acid changes were found in the short exon regions examined. Full-length GluCl cDNA sequences amplified from IVS and IVR worms revealed amino acid differences. The following chapter describes the experiments carried out to determine the effects of these mutations on channel function.

Mutations in the extracellular domains of glutamate-gated chloride channel α 3 and β subunits from ivermectin-resistant *Cooperia oncophora* affect agonist sensitivity

Annete I. Njue,* Jon Hayashi,† Lyle Kinne,† Xiao-Peng Feng* and Roger K. Prichard*

*Institute of Parasitology, McGill University, 21,111 Lakeshore Road, Ste-Anne-de-Bellevue, Quebec, Canada, H9X 3V9

†FMC Corporation, Neurobiology, Highway 1, Plainsboro Road, Princeton, New Jersey, USA, 08540

Journal of Neurochemistry- In press

Abstract

Two full-length glutamate-gated chloride channel (GluCl) cDNAs, encoding GluCla3 and GluClß subunits, were cloned from ivermectin-susceptible (IVS) and -resistant (IVR) Cooperia oncophora adult worms. The IVS and IVR GluCla3 subunits differ at three amino acid positions, while the IVS and IVR GluClß subunits differ at two amino acid positions. The aim of this study was to determine whether mutations in the IVR subunits affect agonist sensitivity. The subunits were expressed singly and in combination in Electrophysiological whole-cell voltage-clamp recordings Xenopus laevis oocytes. showed that mutations in the IVR GluClo3 caused a modest but significant threefold loss of sensitivity to glutamate, the natural ligand for GluCl receptors. As well, a significant decrease in sensitivity to the anthelmintics ivermectin and moxidectin was observed in the IVR GluCla3 receptor. Mutations in the IVR GluClß subunit abolished glutamate sensitivity. Co-expressing the IVS GluClo3 and GluCl_β subunits resulted in heteromeric channels that were more sensitive to glutamate than the respective homomeric channels, demonstrating co-assembly of the subunits. In contrast, the heteromeric IVR channels were less sensitive to glutamate than the homomeric IVR GluCl α 3 channels. The heteromeric IVS channels were significantly more sensitive to glutamate than the heteromeric IVR channels. Of the three amino acids distinguishing the IVS and IVR GluCla3 subunits, only one of them, L256F, accounted for the differences in response between the IVS and IVR GluCla3 homomeric channels.

Ivermectin (22, 23- dihydroavermectin B1a) and moxidectin are potent endectocides that are widely used to control parasite infections in domestic animals (Campbell 1989; McKellar and Benchaoui, 1996). Ivermectin is believed to exert its antiparasitic effects by activating glutamate-gated chloride channels (GluCls) (Cully et al. 1994; Cully et al. 1996; Dent et al. 1997; Vassilatis et al. 1997; Dent et al. 2000). These inhibitory receptors are found only in invertebrates (Cleland 1996), and belong to the superfamily of ligand-gated ion channels. The GluCls of Caenorhabditis elegans have been studied extensively. Four GluCl subunits, GluCla, GLC-3, AVR-14 (GluCla3) and AVR-15, form ivermectin-sensitive channels when expressed in Xenopus oocytes (Cully et al. 1994; Dent et al. 1997; Laughton et al. 1997a; Dent et al. 2000; Horoszok et al. 2001). A fifth subunit, GluCl\beta, forms glutamate-sensitive channels that do not respond to ivermectin (Cully et al. 1994). A GluCla subunit cloned from Drosophila melanogaster forms ivermectin-sensitive channels in Xenopus oocytes (Cully et al. 1996). A strong correlation between ivermectin's nematocidal effect and the potentiation and activation of an ivermectin-sensitive chloride current in Xenopus oocytes has been demonstrated (Arena et al. 1995). Ivermectin also acts on nematode y-aminobutyric acid (GABA)gated chloride channels, and this may also contribute to the antiparasitic effects of macrocyclic endectocides (Holden-Dye and Walker 1990; Feng et al. 2002).

Cooperia oncophora is a nematode that belongs to the superfamily Trichostrogyloidea, and is a common parasite of cattle in temperate climates (Parmentier et al. 1995). It is one of the dose-limiting species for ivermectin in cattle (Shoop et al. 1995). Ivermectin resistance involving *Cooperia* species has been reported (Vermunt et

al. 1996; Coles et al. 1998; Anziani et al. 2001; Coles et al. 2001; Familton et al. 2001; Fiel et al. 2001), and highlights the need to understand ivermectin's mode of action and mechanisms of resistance. Milbemycins, a class of compounds structurally related to avermectins, also activate GluCls (Arena et al. 1995). While some reports suggest that moxidectin, a milbemycin, is effective against ivermectin-resistant nematodes (Craig et al. 1992; Pankavich et al. 1992), other reports suggest that ivermectin resistance leads to loss of sensitivity to moxidectin (Conder et al. 1993; Shoop et al. 1993). Moxidectin has been shown to be ineffective against ivermectin-resistant *C. oncophora* (Vermunt et al. 1996).

In an attempt to understand the role of GluCls in ivermectin resistance in C. oncophora, we have cloned two full-length cDNAs, encoding GluCla3 and β subunits These subunits show high homology to related (Njue and Prichard, submitted). sequences in C. elegans; at the amino acid level, C. oncophora GluCla3 shares 80% identity with the C. elegans AVR-14B/GluCla3B, while C. oncophora GluCl\beta shares 76% identity with C. elegans GluClß subunit. Genetic variation analysis of short fragments of the C. oncophora GluCl genes in ivermectin-susceptible (IVS) and resistant (IVR) worms showed the GluClas gene to be polymorphic, and the GluCl β gene less so (Njue and Prichard, submitted). Initial isolation of the IVR C. oncophora was reported by Coles et al. (1998). Here, we report the cloning of full-length GluCl α 3 and β subunit alleles from IVS and IVR worms. Three non-synonymous mutations were identified in the N-terminal extracellular domains of IVR GluCla3. As well, two mutations were identified in the N-terminal domain of IVR GluClB. To determine whether mutations in the IVR GluCl subunits affect receptor function, the subunits were expressed in *Xenopus* oocytes and responses examined. We demonstrate that *C*. *oncophora* GluCls form functional homomeric and heteromeric channels when expressed in *Xenopus* oocytes, and show that mutations in the two GluCl subunits affect agonist sensitivity.

4.2 Materials and methods

4.2.1 Generation of full-length cDNAs, and cRNA synthesis

Total RNA extracted from bulk IVS and IVR adult worms using Trizol ReagentTM was reverse-transcribed using an oligo-dT primer (GibcoBRL). This was used as template for PCR amplification of the *C. oncophora* GluCl α 3 and GluCl β cDNAs. Subunit cDNAs generated were subcloned into the pT7TS vector, which contains 5' and 3' untranslated *Xenopus* β -globin regions (Dent et al. 1997). The clones were sequenced from both directions using the vector primers T7 and SP6.

The IVS and IVR GluCl α 3 subunits differ at three amino acid positions (Fig. 1). To determine the effect of each mutation on channel properties, site-directed mutagenesis was used to construct three mutants (MUT 1-3) each bearing one of the amino acid substitutions that distinguish IVS and IVR GluCl α 3 (Table 1). All three mutations required a single base substitution. Site-directed mutagenesis was performed on pT7TS-IVS GluCl α 3 clones using the QuickChange mutagenesis kit (Stratagene), and the successful incorporation of mutations was confirmed by sequencing.

All cDNA/pT7TS plasmids were linearized using Xba 1 immediately downstream of the 3' untranslated β -globin sequence, and cRNA was transcribed using the T7 mMessage mMachine Kit (Ambion). Quality and quantity of cRNA were checked by

agarose gel electrophoresis and absorption spectroscopy. Samples were stored at -80°C until use.

Table 1. *C. oncophora* GluCl α 3 mutants tested. To determine the effect of each amino acid substitution on channel properties, three mutants, each representing one of the amino acid substitutions, were constructed. The letter preceding the position number refers to the amino acid in the IVS GluCl α 3 subunit, and the letter following the number refers to the amino acid in the IVR GluCl α 3 subunit. For each mutant, the remainder of the sequence was identical to that of the IVS GluCl α 3 subunit.

GluCla3 MUTANT	NUCLEOTIDE MUTATION	AMINO ACID CHANGE		
MUT 1	GAG to GGG	E114G		
MUT 2	GTT to GCT	V235A		
MUT 3	CTT to TTT	L256F		

4.2.2 Oocyte preparation and injection

Ovarian tissue was isolated from mature female *Xenopus laevis* by a simple surgical procedure. The frogs were anaesthetized by immersion in neutralized ethyl m-aminobenzoate solution (tricaine, methanesulphonate salt, 0.2% solution w/v; Sigma). The isolated ovarian lobes were placed in a dish containing OR2 calcium-free buffer (82.5 mM NaCl, 2 mM KCl, 1mM MgCl₂, 5 mM HEPES, pH adjusted to 7.5 with NaOH) and dissected into smaller clumps of oocytes. These were rinsed several times, and gently shaken in OR2 containing 4 mg/ml collagenase type 1A (Sigma) for 1 hour at

17°C. The oocytes were rinsed thoroughly in OR2 and manually defolliculated before being placed in ND96 buffer (96 mM NaCl, 2 mM KCl, 1mM MgCl₂, 1.8 mM CaCl₂, 5 mM HEPES, pH adjusted to 7.4 with NaOH) supplemented with 100 units/ml penicillin and 100 μ g/ml streptomycin. 9.2 nl of cRNA (~1ng/subunit) in RNAse-free water were injected cytoplasmically. To express the heteromeric GluClo3/GluCl β receptors, the two subunit cRNAs were mixed at a ratio of 1:1 before injection into oocytes. Injected oocytes were maintained at 17°C in ND96 supplemented with penicillin and streptomycin, and electrophysiological recordings were performed at room temperature (22-25°C) 2-3 days after injection.

4.2.3 Electrophysiological recordings

Whole-cell currents were recorded from oocytes using the two-electrode voltage-clamp technique. Oocytes were voltage-clamped at a holding potential of -80 mV using Axoclamp 2B voltage clamp amplifier (Axon Instruments). Recordings were made in ND96. Recording microelectrodes were filled with 3M potassium acetate, and had tip resistances of 1-3 M Ω . Oocytes were held in a 0.13 ml bath, and continually superfused with ND96 at a rate of 0.8 ml/min. Drugs were applied by local microperfusion using an electronic digital pipette, and the duration of application was 4 sec. For the dose-response experiments, recordings were made sequentially from the lowest to the highest drug concentration, with an interval of 2 min between applications. For the glutamate responses, a second application of the same concentration was applied every third response to ensure reproducibility. Glutamate was dissolved in ND96. Ivermectin and

moxidectin were prepared as 10 mM stock solutions in dimethyl sulfoxide (DMSO) and diluted in ND96.

Data acquisition and analysis were conducted using the pCLAMP suite of programs (Axon Instruments). The current-voltage relationship of the glutamate-induced response was determined over a voltage range of -80 to +10 mV. 250 μ M glutamate was first applied at -80 mV. This was followed by repeated glutamate application at holding potentials of -40, -20, 0 and 10 mV. The peak amplitude of the glutamate-induced currents at various potentials was normalized by assigning 100% to the value at -80 mV. The glutamate dose-response relationships were measured by applying a series of glutamate dose-response relationships were measured by applying a series of sturate dose-response curves were normalized by assigning 100% to current activated by saturating concentrations of glutamate in the same oocyte. Because the effects of ivermectin and moxidectin were irreversible, the responses to successive drug administration were taken to be additive (Arena et al. 1995). The ivermectin and moxidectin, respectively. Dose-response data were analyzed using GraphPad Prism (GraphPad Software). Dose-response curves were fitted to the equation:

$$I / I_{max} = 1 / [1 + (EC_{50} / [D])^{h}],$$

where I_{max} is the maximal response, [D] is the drug concentration, EC₅₀ is the concentration of agonist necessary to elicit half the maximum response, and h is the Hill

coefficient. Data are given as means \pm S.E. Statistical analyses were performed using Student's unpaired two-tailed *t* test, with a significance level of *P* < 0.05.

4.2.4 Drugs

Except for moxidectin, all drugs were purchased from Sigma. Moxidectin was a gift from Fort Dodge Animal Health.

4.3 Results

Full-length cDNAs encoding *C. oncophora* GluClo3 and GluCl β subunits were obtained from ivermectin-susceptible (IVS) and ivermectin-resistant (IVR) worms. The predicted protein sequences were aligned using Clustal W and are shown in Figure 1. Comparison of the IVS and IVR GluClo3 subunits showed three amino acid differences at positions 114, 235 and 256. The first two mutations (E114G and V235A) were a result of transitional substitutions at the second codon position (Table 1), while the position 256 mutation (L256F) was caused by a substitution at the first codon position. All three mutations were located in the N-terminal extracellular domain. Alignment of the IVS and IVR GluCl β subunits also revealed two mutations (V60A and R101H) in the Nterminal extracellular domain, both resulting from transitional substitutions at the second codon position (Fig. 1B). **Figure 1.** Comparing the predicted amino acid sequences of the *C. oncophora* GluCl α 3 (A) and GluCl β (B) subunits from IVS and IVR worms. Conserved residues are shaded grey. The putative transmembrane domains, TM 1 – 4, are overlined. A. The IVS and IVR GluCl α 3 sequences differ at three amino acid positions (highlighted by $\mathbf{\nabla}$), all located in the N-terminal extracellular domain. B. The IVS and IVR GluCl β sequences differ at two amino acid positions, both found in the N-terminal extracellular domain.

Figure 1A.

		10	20	30	40	50	60
IVS GluClα3	MRTSVPLA	TRIGPIL	ALICIVI	TIISTVEGKRI	LKEQEIIQRI	LNNYDWRVRPRG	LNASW
IVR GluCla3	MRTSVPLA	TRI GPIL	ALICIVI	TIISTVEGKRI	LKEQEIIQRI	LNNYDWRVRPRG	LNASW
		70					
IVS GluClo3	DICCOVI		80	90 DVNNEVOAOET	TOFENOLOADI		120
IVB GluCia3	POTGGPVL		Delekin	DVNMETSAUF	FREEWVUARL	ATGHFEDESTEV	PPFVV
in a contraction		<u>v i v ivi i 1</u>	<u>noroki u</u>	DVINICISAULI	CHEEWVUANL	ATURFEDESIUS	PPFVV
	1	30	140	150	160	170	180
IVS GluClα3	LATSENAD	QSQQI WM	PDTFFQN	EKEARRHLI DI	PNVLIRIHKD	GSILYSVALSLV	LSCPM
IVR GluCla3	LATSENAD	QS QQ1 WM	PDTFFQN	EKEARRHLI DI	(PNVLI RI HKD	GSILYSVALSLV	LSCPM
							,
IVS GluClas	CIECVOID	BONCLID					
IVB GluCla3	SIFEVDIN	PONCLID	L A S I A I I I A S V A V T	TODIKYEWKE	IN FVQQADGL H		TEVOT
		<u>nonoli d</u>	LAJIAII		<u>arvaandarn</u>	<u>WOLFOFLLUDV</u> A	<u>li kron</u>
		_		T 88	4		788.0
	2	150	260 -	1 141		290	1 W Z
IVS GluCla3	SKTNTGEY	SCARVKL	LLRREYS	YYLI QLYI PCI	MLVVVSWVSF	WLDKDAVPARVS	LGVTT
IVR GluCla3	SKTNTGEY	SCARVKL	FLRREYS	YYLI QLYI PCI	MLVVVSWVSF	WLDKDAVPARVS	LGVTT
			200		TM 3	04////////////////////////////////////	250
WS GluCla3	LITMITOA	CLNCKI	DDVEVIK	AVDVWLCVCL	ELEGALLEVA	VVNVVQDKELE	INENNY
IVB GluCio3	LITMTTOA	SGINSKI	PPVSVIK	AVDVWI GVCL	FIFGALLETA	VVNVVGRKFFLF	KEKKK
		<u>oor woke</u>	<u></u>	<u></u>			
							TAA A
	8	370	380	390	400	410	1 101 49
IVS GluCla3	KTRLDDCV	CPSERPA	LRLDLST	FRRRGWTPLN	RLLDVLGRNAD	LSRRVDLMSRIT	FPTLF
IVR GluCla3	KTRLDDCV	CPSERPA	LRLDLST	FRRRGWTPLN	RLLDVLGRNAD	LSRRVDLMSRIT	FPTLF

IVS GluCla3	TVFLVFYYSVYVKQSNLE
IVR GluCla3	TVFLVFYYSVYVKQSNLE

Figure 1B.

		10	20	30		40	50	60
IVS GluClβ	MTFALVFT	VATVAV	MADSSH	VSRRSSGGTC	EQEILNEL	LSNYDMRVI	RPPPTNYSD	PTGPV
IVR GluClβ	MTFALVFT	VATVAV	MADSSH	VSRRSSGGTO	EQEILNEL	LSNYDMRVI	RPPPTNYSD	PTGPA
		70						
IVS GluCia	TVDVALL		80		WI DODLAW		110	120
IVB GluCIB	TVDVNIMI	DMICKI	DVVNME	I SMULIFHEU Venniteden	WLUSHLAY	ARLGYHNPI	PKFLIVPHI	KSNLW
	[] 0 13 0 13 1 MI	<u>naconi</u>	DVVI	ISMULITALU	WLDONLAI		PAPLIVPAI	NONLW
		130	140	150		160	170	180
IVS GluClβ	I PDTFFP1	EKAAHF	HLIDTD	NMFLRIHPDO	KVLYSSRI	SITSSCHM	QLQLYPLDL	QFCDF
IVR GluClβ	I PDTFFP1	EKAAHF	HLI DTD	NMFLRI HPDC	KVLYSSRI	SITSSCHM	<u>QLQLYPLDL</u>	QFCDF
		100	200					040
IVS GluCIB	DIVSVAHT	MKDLVV		DVOL KBGVCS	DIPNEOLT	NITTNDDC	TCHTNTCCV	ACL D SS
IVR GluCIB	DLVSYAHT	MKDIVY	OWDPTA	PVQLKPGVGS	DLPNFQLT	NITTNDDC	TSHTNTGSY	ACLRM
	and the second				and the second secon			
			TM	1			TM 2	
			TM	1			TM 2	
IVS GluClβ	QLTL KROP	SYYLVC	TM	1 MIVIVSWVSF	WI DMHSTA	GRVALGVT	TM 2	300 A A I N A
IVS GluClβ IVR GluClβ	QLTLKRQF QLTLKRQF		TM QLYGPTT QLYGPTT	1 Mivivswvsf Mivivswvsf	WIDMHSTA WIDMHSTA	GRVAL GVT GRVAL GVT	TM 2 TLLTMTTMG TLLTMTTMG	300 IAAINA IAAINA
IVS GluClβ IVR GluClβ	QLTLKROF QLTLKROF		TM QLYGPTT QLYGPTT	1 MIVIVSWVSF MIVIVSWVSF	WIDMHSTA WIDMHSTA	GRVALGVT GRVALGVT	TM 2 TLLTMTTMC TLLTMTTMC	300 IAAINA IAAINA
IVS GluClβ IVR GluClβ	QLTLKRQF QLTLKRQF	SYYLVC SYYLVC 310	TM RLYGPTTI RLYGPTTI	1 MIVIVSWVSF MIVIVSWVSF TM3	WIDMHSTA WIDMHSTA	GRVALGVT GRVALGVT	TM 2 TLLTMTTMC TLLTMTTMC 350	300 IAAINA IAAINA 360
IVS GluClβ IVR GluClβ IVS GluClβ	QLTLKRQF QLTLKRQF KLPPVSYV	SYYLVC SYYLVC 310 /KVVDVV	TM · RLYGPTTI RLYGPTTI	1 MIVIVSWVSF MIVIVSWVSF TM3 FVFGALLEYA	WIDMHSTA WIDMHSTA	GRVALGVT GRVALGVT 340 RQTDQAKS	TM 2 T L L T M T T M G T L L T M T T M G 350 G A T R K A Q K R	300 IAAINA IAAINA 360 RAQME
IVS GluClβ IVR GluClβ IVS GluClβ IVS GluClβ	QLTLKRQF QLTLKRQF KLPPVSYV KLPPVSYV	SYYLVC SYYLVC 310 /KVVDVV /KVVDVV	TM · RLYGPTT RLYGPTT VLGACQT VLGACQT	1 MIVIVSWVSF MIVIVSWVSF TM3 FVFGALLEYA FVFGALLEYA	WIDMHSTA WIDMHSTA	GRVALGVT GRVALGVT 340 RQTDQAKS RQTDQAKS	TM 2 T L L T M T T M G T L L T M T T M G 350 350 Q A T R K A Q K R Q A T R K A Q K R	300 AAINA AAINA 360 RAQME RAQME
IVS GluClβ IVR GluClβ IVS GluClβ IVR GluClβ	QLTLKRQF QLTLKRQF KLPPVSYV KLPPVSYV	SYYLVC SYYLVC 310 /KVVDVV /KVVDVV	TM · RLYGPTT RLYGPTT VLGACQT VLGACQT	1 MIVIVSWVSF TM3 FVFGALLEYA FVFGALLEYA	WIDMHSTA WIDMHSTA WIDMHSTA	GRVAL GVT GRVAL GVT 340 RQT DQAKS RQT DQAKS	TM 2 TLLTMTTMC TLLTMTTMC 350 QATRKAQKR QATRKAQKR	300 AAINA AAINA 360 RAQME RAQME
IVS GluClβ IVR GluClβ IVS GluClβ IVR GluClβ	QLTLKRQF QLTLKRQF KLPPVSYV KLPPVSYV	SYYLVC SYYLVC 310 /KVVDVV /KVVDVV	TM · RLYGPTTI RLYGPTTI VLGACQT	1 MIVIVSWVSF MIVIVSWVSF TM3 FVFGALLEYA FVFGALLEYA	WIDMHSTA WIDMHSTA 	GRVAL GVT GRVAL GVT 340 RQT DQAKS RQT DQAKS	TM 2 TLLTMTTMC 1LLTMTTMC 350 QATRKAQKR QATRKAQKR	300 A A I N A A A I N A 360 R A Q M E R A Q M E TM 4
IVS GluClβ IVR GluClβ IVS GluClβ IVR GluClβ	ALTLKRAF ALTLKRAF KLPPVSYV KLPPVSYV	310 (KVVDVV (KVVDVV 370 370	TM · RLYGPTTI RLYGPTTI VLGACQT VLGACQT	1 MIVIVSWVSF MIVIVSWVSF TM3 FVFGALLEYA FVFGALLEYA 390	WI DMHSTA WI DMHSTA 	GRVAL GVT GRVAL GVT 340 RQT DQAKS RQT DQAKS 400	TM 2 TLLTMTTMC 1LLTMTTMC 350 QATRKAQKR QATRKAQKR	300 AAINA AAINA 360 RAQME RAQME TM4
IVS GIUCIβ IVR GIUCIβ IVS GIUCIβ IVR GIUCIβ IVS GIUCIβ	GLTLKRQF GLTLKRQF KLPPVSYV KLPPVSYV	310 (KVVDVV (KVVDVV 370 370 370 370 370 370 370 370	TM - RLYGPTTI RLYGPTTI VLGACQT VLGACQT 380 ILYQDYE	1 MIVIVSWVSF TM3 FVFGALLEYA FVFGALLEYA 390 PTLRDRLRR)	WIDMHSTA WIDMHSTA FVSYQDSQ FVSYQDSQ FVSYQDSQ FVSYQDSQ FTKPDYLP	GRVAL GVT GRVAL GVT 340 RQT DQAKS RQT DQAKS 400 AKI DYYAR AKI DYYAR	TM 2 TLLTMTTMG 350 QATRKAQKR QATRKAQKR FCVPLGFLA	300 AAINA AAINA 360 RAQME RAQME TM4
IVS GIUCIβ IVR GIUCIβ IVR GIUCIβ IVR GIUCIβ IVR GIUCIβ	QLTLKRQF QLTLKRQF KLPPVSYV KLPPVSYV LSERDHY(LSERDHY(310 (KVVDVV (KVVDVV (KVVDVV 370 2PPCTCH	TM T TM T THE YGPTT THE YGPTT VL GACQT VL GACQT VL GACQT 380 IL YQDYE TL YQDYE	1 MIVIVSWVSF TM3 FVFGALLEYA FVFGALLEYA 390 PTLRDRLRRY PTLRDRLRRY	WI DMHSTA WI DMHSTA FVSYQDSQ FVSYQDSQ FVSYQDSQ FTKPDYLP FTKPDYLP	GRVALGVT GRVALGVT 340 RQTDQAKS RQTDQAKS 400 AKIDYYAR AKIDYYAR	TM 2 TLLTMTTMC 1LLTMTTMC 350 QATRKAQKR QATRKAQKR FCVPLGFLA	300 AAINA AAINA 360 RAQME RAQME TM 4 FNAIY FNAIY
IVS GIUCIβ IVR GIUCIβ IVR GIUCIβ IVR GIUCIβ IVR GIUCIβ	QLTLKRQF QLTLKRQF KLPPVSYV KLPPVSYV LSERDHY(LSERDHY(310 (KVVDVV (KVVDVV (KVVDVV 370 2PPCTCH 2PPCTCH	TM - RLYGPTTI RLYGPTTI VLGACQT VLGACQT VLGACQT 380 ILYQDYE ILYQDYE	1 MIVIVSWVSF TM3 FVFGALLEYA FVFGALLEYA 990 PTLRDRLRRY PTLRDRLRRY	WI DMHSTA WI DMHSTA FVSYQDSQ FVSYQDSQ FVSYQDSQ FTKPDYLP FTKPDYLP	GRVALGVT GRVALGVT 340 RQTDQAKS RQTDQAKS 400 AKIDYYAR AKIDYYAR	TM 2 TLLTMTTMC 1LLTMTTMC 350 QATRKAQKR QATRKAQKR FCVPLGFLA	300 AAINA AAINA 360 RAQME RAQME TM 4 FNAIY FNAIY
IVS GIUCIβ IVR GIUCIβ IVR GIUCIβ IVR GIUCIβ IVR GIUCIβ	QLTLKRQF QLTLKRQF KLPPVSYV KLPPVSYV LSERDHY(LSERDHY(310 (KVVDVV (KVVDVV (KVVDVV 370 2PPCTCH 2PPCTCH	TM - RLYGPTT RLYGPTT VLGACQT VLGACQT VLGACQT ILYQDYE ILYQDYE	1 MIVIVSWVSF TM3 FVFGALLEYA FVFGALLEYA 990 PTLRDRLRRY PTLRDRLRRY	WI DMHSTA WI DMHSTA FVSYQDSQ FVSYQDSQ FVSYQDSQ FTKPDYLP FTKPDYLP	GRVALGVT GRVALGVT 340 RQTDQAKS RQTDQAKS 400 AKIDYYAR AKIDYYAR	TM 2 TLLTMTTMC 350 QATRKAQKR QATRKAQKR FCVPLGFLA	300 AAINA AAINA 360 RAQME RAQME TM4 FNAIY FNAIY

IVR GluCIB WTSCLVMVSRLI

cRNAs corresponding to the IVS and IVR GluCl α 3 and GluCl β subunits were expressed in *Xenopus* oocytes as homomers and heteromers. Initially, 5 ng of cRNA (in 26 nl) were injected into oocytes. This led to a dramatic increase in holding current, which may have been caused by spontaneous opening of channels in the absence of agonist. This was particularly so in oocytes expressing the IVS GluCl α 3-GluCl β heteromeric channels. As well, this 'leakiness' was more evident when recordings were done three or more days post-injection. As a result, the amount of RNA injected in ooctyes was reduced to ~1 ng, and recordings were done from as early as 24 hours post-injection and continued on day two post-injection.

The IVS and IVR GluCl α 3 subunits were expressed in *Xenopus* oocytes as homomers, and as heteromers with the IVS and IVR GluCl β subunits, respectively. Figure 2 illustrates the responses evoked in homomeric and heteromeric channels by increasing concentrations of glutamate, and the corresponding dose-response relationships. At a holding potential of -80mV, glutamate application elicited robust inward currents in oocytes expressing IVS and IVR GluCl α 3 subunits (Fig. 2A). The responses were dose-dependent, rapid in onset and completely reversible. The normalized dose-response curves for the IVS and IVR GluCl α 3 homomeric receptors are shown in Fig. 2B. The presence of three mutations (E114G, V235A and L246F) in the IVR GluCl α 3 subunit resulted in a shift of glutamate response curve of the receptor to the right. Glutamate EC₅₀ values showed a significant threefold increase from 29.7 ± 4 (n = 3) in IVS GluCl α 3 to 96.1 ± 4.4 µM (n = 3) in IVR GluCl α 3 (Table 3A). Hill slopes for the two receptors were 2.4 ± 0.21 and 2.4 ± 0.45, respectively, suggesting that binding of more than one glutamate was required to gate the channel. As with the GluCl α 3 subunits, the IVS GluCl β subunit also formed functional glutamate-gated homomeric channels (Figs. 2A and B). Compared to the robust responses elicited by glutamate in the GluCl α 3 homomeric channels, the IVS GluCl β channel responses were smaller (Fig. 2A). The mean maximum current for the GluCl α 3 homomeric receptor was six times that for the GluCl β homomeric receptor. The EC₅₀ value for glutamate for the GluCl β receptor was 185.6 ± 24.9 μ M (n = 3) and the Hill value 2.2 ± 0.02. The IVR GluCl β subunit channel did not respond to any of the glutamate concentrations tested (10nm to 1mM).

Glutamate also elicited responses in cells co-expressing IVS GluCl α 3 and IVS GluCl β subunits (Fig. 2C). Relative to the respective homomeric receptors, the dose-response curve of the heteromeric channel shifted to the left. The EC₅₀ value was estimated to be 13.4 ± 2.5 μ M (n = 3), and the Hill slope 1.9 ± 0.28. In contrast, co-expressing the IVR GluCl α 3 and IVR GluCl β subunits shifted the concentration-response curve to the right, relative to the IVR GluCl α 3 homomeric receptor, indicating a reduction in glutamate sensitivity. The EC₅₀ value obtained for this channel was 171.6 ± 20.7 μ M (n = 3), manifesting a significant ~ 13-fold decrease in glutamate sensitivity relative to the heteromeric IVS receptor. Uninjected and water-injected oocytes did not show any response to glutamate.

Figure 2. Activation of IVS GluCla3 and GluClβ receptors by glutamate. IVS GluCla3 and GluClß subunits form functional glutamate-activated homomeric and heteromeric channels when expressed in Xenopus oocytes (A, B). A. Representative current traces from oocytes activated with glutamate at concentrations shown above each trace. Currents were measured using the whole-cell-two-electrode voltage clamp technique, at V_{hold} = -80 mV. The GluCl β homomeric channel is less sensitive to glutamate than the GluCl α 3 homomeric and GluCl α 3/GluCl β heteromeric channels. B. Normalized doseresponse curves for glutamate-induced currents for oocytes expressing homomeric receptors (same oocytes as in A for IVS). EC₅₀ values for IVS GluClα3 (■), IVR GluCla3 (\blacktriangle) and IVS GluCl β (\circ) homometric receptors = 29.7, 96.1 and 185.6 μ M, respectively. IVR GluCl β (\Diamond) receptor showed no response to glutamate. C. Glutamate dose-response curves for ooctyes expressing homomeric GluCla3 (same as B) and heteromeric Co GluCla3/GluClB channels. Compared to the IVS heteromeric channel (\circ), the IVR heteromeric channel (\Diamond) showed a shift to the right, with a ~ 13-fold increase in EC_{50} value (13.4 μM and 171.6 μM for the IVS and IVR heteromeric channels, respectively).





.

Figure 2B. Dose responses for homomeric receptors.



Figure 2C. Dose responses for homomeric GluCla3 and heteromeric receptors.


The current-voltage relationship for the glutamate-sensitive current from the IVS GluCl α 3 homomeric channels showed a slightly outwardly rectifying voltage dependence (Fig. 3). The reversal potential was $-22 \pm 1 \text{ mV}$ (n = 3), close to the Nernst potential for chloride in *Xenopus* oocytes (-24 mV, with 103.6 mM extracellular chloride and assuming 40 mM intracellular chloride [Dascal 1987]).



Figure 3. Current-voltage (I/V) relationship for the glutamate-gated currents of the IVS GluCl α 3 receptor was tested at potentials ranging from -80 mV to +10 mV. Current amplitude were normalized by assigning 100% to the glutamate-induced current recorded at -80 mV. The reversal potential was -22 ± 1 mV (n = 3 ooctyes).

The effects of ivermectin and moxidectin were also tested on the homomeric channels. Unlike the glutamate response, the responses to ivermectin and moxidectin were slow to activate and irreversible (Fig. 4A). The current did not return to baseline even after washing for several minutes with drug-free ND96. Application of 250 µM glutamate and 5 uM moxidectin during maximal response to ivermectin did not elicit additional response. No response to ivermectin or moxidectin was observed in uninjected or waterinjected oocytes. Oocytes expressing IVS GluCla3 homomeric channels displayed an ivermettin dose-response relation with an EC₅₀ of $0.5 \pm 0.12 \,\mu\text{M}$ (n = 3) and a Hill slope of 2.1 ± 0.5 (Fig. 4B, Table 3B). As with the glutamate response, the ivermectin response of the IVR GluCla3 receptor was shifted to the right relative to the IVS GluCl α 3 receptor, with a significant increase in the ivermectin EC₅₀ of about threefold $(EC_{50} = 1.3 \ \mu M \pm 0.11)$, Hill slope 2.2 \pm 0.18, n = 3). In contrast, the IVS GluClas moxidectin dose-response curve was shifted to the left relative to the IVS GluCla3 ivermectin curve (Fig. 4B). The EC₅₀ was estimated to be $0.2 \pm 0.06 \mu M$ (n = 3), and the Hill slope 1.6 ± 0.36 . A similar trend was seen when the IVR GluCla3 receptor doseresponse curves for ivermectin and moxidectin were compared. The IVR GluCl α 3 moxidectin curve had an EC_{50} value similar to that of the IVS GluClo3 ivermectin curve $(0.5 \pm 0.05 \mu M, n = 3 \text{ and } 0.5 \pm 0.12 \mu M, n = 3$, respectively). Relative to the IVS GluCla3 moxidectin curve, IVR GluCla3 also showed a significant decrease in moxidectin EC_{50} of ~ 2.6-fold. No ivermectin- or moxidectin-activated chloride currents were observed in oocytes expressing GluCl^β homomeric channels.

Figure 4. Activation of IVS and IVR GluCl α 3 receptors by ivermectin (IVM) and moxidectin (MOXI) (A, B). Evidence that ivermectin and moxidectin are agonists of *C. oncophora* GluCl α 3 receptors. A. Currents induced by glutamate, ivermectin and moxidectin, applied at the concentrations indicated. While the glutamate response is rapid in activation of the membrane current and completely reversible, the ivermectin and moxidectin responses were slow in activation and irreversible. **B.** Normalized dose-response curves for ivermectin and moxidectin. The EC₅₀ values for ivermectin for IVS (\bullet) and IVR (\bullet) GluCl α 3 receptors = 0.5 and 1.3 μ M respectively. The EC₅₀ values for moxidectin for IVS (\bullet) and IVR (\bullet) GluCl α 3 receptors = 0.2 and 0.5 μ M respectively.





Figure 4B.



In an attempt to determine whether one, two or all three mutations in the IVR GluClo3 subunit contributed to altered response to glutamate and ivermectin, the mutants, each representing one mutation, were expressed individually in *Xenopus* oocytes. Fig. 5A compares the glutamate dose responses of the mutants with those of the IVS and IVR GluClo3 homomeric channels. Mut 1 (E114G) and Mut 2 (V235A) both showed dose response curves very similar to those of the IVS GluClo3 receptor, with EC₅₀ values of 25.7 ± 5.2 (n = 2) and 27.8 ± 2.6 (n = 3) μ M for Mut 1 and Mut 2, respectively. In contrast, the glutamate dose-response curve and EC₅₀ value (100.6 μ M ± 27 , n = 2) of the Mut 3 (L256F) receptor were very similar to those of the IVR GluClo3 receptor. A similar trend was observed with the ivermectin dose-response curves (Fig. 5B). EC₅₀ values for the Mut 1, Mut 2 and Mut 3 ivermectin curves were 0.6 ± 0.18 , 0.4 ± 0.01 and $1.2 \pm 0.11 \mu$ M, respectively.

Figure 5. Normalized dose-response curves of mutant GluCl α 3 receptors for glutamate and ivermectin (A, B). A. The EC₅₀ values for glutamate for Mut 1 (E114G, \Diamond), Mut 2 (V235A, \circ) and Mut 3 (L256F, \Box) GluCl α 3 receptors = 25.7, 27.8 and 100.6 μ M, respectively. B. The EC₅₀ values for ivermectin for Mut 1 (\Diamond), Mut 2 (\circ) and Mut 3 (\Box) receptors = 0.6, 0.4 and 1.2 μ M, respectively.



Figure 5A. Glutamate dose response curves



Figure 5B. Ivermectin dose response curves



The pharmacological properties of the homomeric channels were also examined. Ibotenate, a structural analogue of glutamate known to activate GluCls (Lea and Usherwood 1973; Arena et al. 1992; Cully et al. 1994), elicited responses in IVS GluCl α 3 and IVS GluCl β homomeric channels (Fig. 6, Table 2). For both channels, the kinetics of the glutamate and ibotenate responses were similar. While 1 mM ibotenate maximally activated the IVS GluCl β channel, the response of the IVS GluCl α 3 channel was 80 ± 6.1% of the response to 1 mM glutamate. Apart from ibotenate, no other agonist activated the IVS GluCl α 3 and IVS GluCl α 4 channel, the response of the IVS GluCl α 3 channel maximally activated the IVS GluCl α 3 and IVS GluCl β homomeric channels. Apart from ibotenate, no other agonist activated the IVS GluCl α 3 and IVS GluCl β homomeric channels (Table 2). The glutamate analogs D-glutamate, quisqualate, kainate and NMDA were inactive. As well, 1 mM concentrations of glycine and GABA failed to activate current.



Figure 6. Ibotenate activated IVS GluCl α 3 and IVS GluCl β homomeric channels. Currents activated by glutamate and ibotenate at concentrations shown above each current plot. The kinetics of the glutamate and ibotenate responses are similar in both channels.

Table 2. Pharmacology of Co GluCl α 3 and GluCl β homomeric channels. For each ligand, at least three oocytes were tested.

AGONIST (LIGAND)	IVS GluCla3 CHANNEL	IVS GluClβ CHANNEL	
	RESPONSE (nA)	RESPONSE (nA)	
250 µM Glutamate	-5936 ± 409	-311 ± 11	
1 mM Ibotenate	-4785 ± 362	-310 ± 9	
1 mM Aspartate	0	0	
1 mM D-Glutamate	0	0	
500 µM Quisqualate	0	0	
500 µM Kainate	0	0	
1 mM Glycine	0	0	
1 mM GABA	0	0	
1 mM NMDA	0	0	

Table 3. Summary of effects of (A) glutamate and (B) ivermectin (IVM) and moxidectin (MOXI) on IVS, IVR and mutant GluCl receptors. EC₅₀ and Hill slope values are expressed as means \pm S. E. of mean. To express heteromeric channels, cRNAs for the GluCla3 and β subunits were injected in a 1: 1 ratio.

Table 3A. Glutamate responses

Receptor	EC ₅₀ (μM)	Hill slope	Oocytes (n)
IVS GluCla3	29.7 ± 4	2.4 ± 0.21	3
IVR GluCla3	96.1 ± 4.4	2.4 ± 0.45	3
IVS GluClβ	185.6 ± 24.9	2.2 ± 0.02	3
IVR GluClβ	-	-	3
IVS GluCl(α 3 + β)	13.4 ± 2.5	1.9 ± 0.28	3
IVR GluCl(α 3 + β)	171.6 ± 20.7	2.5 ± 0.28	3
MUT 1 (E114G)	25.7 ± 5.2	1.8 ± 0.21	3
MUT 2 (V235A)	27.8 ± 2.6	2.2 ± 0.27	3
MUT 3 (L256F)	100.6 ± 27	1.5 ± 0.35	3

Table 3B. IVM and MOXI responses

Receptor	EC ₅₀ (µM)	Hill slope	Oocytes (n)
IVM- IVS GluClα3	0.5 ± 0.12	2.1 ± 0.5	3
IVM- IVR GluCla3	1.3 ± 0.11	2.2 ± 0.18	3
MOXI- IVS GluCla3	0.2 ± 0.06	1.6 ± 0.36	3
MOXI- IVR GluCla3	0.5 ± 0.05	1.7 ± 0.22	3
IVM- MUT 1 (E114G)	0.6 ± 0.18	2.3 ± 0.3	2
IVM- MUT 2 (V235A)	0.4 ± 0.01	2.2 ± 0.33	3
IVM- MUT 3 (L256F)	1.2 ± 0.11	1.9 ± 0.16	2

4.4 Discussion

GluCls are members of the ligand gated ion channel (LGIC) superfamily which includes the cationic channels activated by acetylcholine and serotonin, as well as anionic channels activated by γ -aminobutyric acid (GABA), glycine, histamine and serotonin (Ortells and Lunt 1995; Ranganathan et al. 2000; Zheng et al. 2002). GluCls are found only in invertebrates (Cleland 1996), and are therefore ideal drug targets. Ivermectin is a widely used anthelmintic which is known to interact with GluCls (Cully et al. 1994; Arena et al. 1995; Cully et al. 1996; Dent et al. 1997; Vassilatis et al. 1997; Dent et al. 2000). This interaction is thought to mediate at least part of its anthelmintic activity, since the ability of ivermectin to activate GluCls expressed in *Xenopus* oocytes correlates well with its biological activity (Arena et al. 1995).

In this study, we describe the function of two GluCl subunits, GluCl α 3 and GluCl β , from *C. oncophora. C. oncophora* GluCl α 3 subunit is a homologue of *C. elegans* AVR-14B/GluCl α 3B (Dent et al. 2000) and *H. contortus* gbr-2B/GluCl α 3B (Jagannathan et al. 1999). *C. elegans* and *H. contortus* GluCl α 3 genes are alternatively spliced to yield two subunits, GluCl α 3A and 3B (Jagannathan et al. 1999; Dent et al. 2000). While we did not clone a GluCl α 3A homologue from *C. oncophora*, it is likely that, given the close phylogenetic relationship among these three nematodes, the *C. oncophora* GluCl α 3 may also be alternatively spliced. Initial amplification of the full-length *C. oncophora* GluCl α 3 cDNA yielded two products (data not shown), and the second less abundant product, which most likely represents the GluCl α 3A, was not cloned. While *H. contortus* GluCl α 3B subunit binds ivermectin with high affinity, the 3A subunit does not (Cheeseman et al. 2001). The *C. elegans* AVR14A/GluCl α 3A

subunit, which is homologous to *H. contortus* GluCl α 3A, does not respond to glutamate or ivermectin when expressed in *Xenopus* oocytes (Dent et al. 2000). Like *C. elegans* GluCl α 3B, *C. oncophora* GluCl α 3 forms ivermectin-sensitive glutamate-gated channels when expressed in *Xenopus* oocytes. Other *C. elegans* GluCls that are activated by glutamate and ivermectin include glc-3 (Horoszok et al. 2001), GluCl α 2A and GluCl α 2B (Dent et al. 1997). As well, a GluCl α subunit also exists in *C. elegans* that forms a homomeric channel gated by ivermectin but not glutamate (Cully et al. 1994). The second *C. oncophora* subunit, GluCl β , is homologous to the *C. elegans* (Cully et al. 1994) and *H. contortus* (Delany et al. 1998) GluCl β subunits. Unlike *C. elegans* AVR-14B/ GluCl α 3B and *C. oncophora* GluCl α 3 receptors, homomeric channels formed by *C. oncophora* IVS GluCl β and *C. elegans* GluCl β are activated by glutamate, but not ivermectin.

C. oncophora GluCl α 3 and GluCl β subunits cloned from IVS and IVR worms show amino acid differences. Mutations in both subunits were found in the N-terminal extracellular domain, which carries the ligand binding site (Etter et al. 1996). When expressed in *Xenopus* oocytes, IVS and IVR GluCl α 3 subunits both formed ivermectinand moxidectin-sensitive glutamate gated channels. Mutations in the IVR GluCl α 3 subunit caused a modest but significant threefold decrease in sensitivity to glutamate, and significant ~ 2.5-fold reductions in sensitivity to ivermectin and moxidectin. Mutations in the IVR GluCl β subunit abolished responsiveness to glutamate. Mutations in genes encoding LGIC receptor subunits have been shown to confer drug resistance. A single A302S mutation in the *Drosophila melanogaster Rdl* gene, which encodes a GABA receptor subunit, confers resistance to cyclodiene insecticides (Ffrench-Constant et al. 1998). As well, a proline to serine mutation in the gene coding for the *D. melanogaster* GluCl α subunit causes a significant reduction in sensitivity to ivermectin and nodulisporic acid (Kane et al. 2000). In *C. elegans*, mutations of *lev-1*, a gene which encodes a non- α nicotinic acetylcholine receptor (nAchR) subunit, causes a reduction in sensitivity to levamisole, even though this subunit only forms a levamisole-sensitive channel in the presence of other subunits (Fleming et al. 1997). Also in *C. elegans*, mutations in two of three GluCl genes, *avr-14*, *glc-1* or *avr-15*, confers modest or no resistance to ivermectin, while mutations in the three genes results in a 4000-fold resistance to ivermectin (Dent et al. 2000).

To determine whether all three mutations in the IVR GluCl α 3 contribute to loss of glutamate and ivermectin sensitivity, each mutation was introduced singly to the IVS GluCl α 3 using site-directed mutagenesis. The responses of Mut 1 (E114G) and Mut 2 (V235A) receptors were similar to those of the IVS GluCl α 3 receptor, suggesting that the mutations at these positions had no influence on ligand binding. In contrast, Mut 3 glutamate and ivermectin dose-response curves were similar to the IVR GluCl α 3 receptor responses, suggesting that the L256F mutation accounted for the difference between IVS and IVR GluCl α 3 channels in response to these two ligands. Whether this amino acid contributes directly to ligand binding, or whether it causes conformational change that influences ligand binding, is not known, and requires further characterization. The L256F mutation does not map onto any of the six domains of the nAchR subunits that form the acetylcholine binding site (Corringer et al. 1995, 2000; Brejc et al. 2001). That this mutation is not found in critical regions may explain the modest loss of agonist sensitivity observed. In *D. melanogaster*, a single mutation in the TM2 region of the *rdl* gene that codes for a GABA-gated channel confers high-level resistance to the insecticide dieldrin (Ffrench-Constant et al. 1993). Also in *D. melanogaster*, a single mutation in the M2-M3 linker region of DmGluClα gene confers over 20-fold resistance to the novel insecticide and acaricide, nodulisporic acid (Kane et al. 2000).

In oocytes expressing IVR GluCl β , no glutamate-activated currents were detected. This could be due to failure of the receptor to assemble on the cell surface. However, this seems unlikely, since co-expression of this subunit with the IVR GluCl α 3 resulted in the formation of receptors that were less sensitive to glutamate than the homomeric IVR GluCl α 3 receptors, suggesting co-assembly of the two subunits. It is possible that the mutations in IVR GluCl β may have altered the binding site, preventing glutamate from binding. Heteromeric channels formed by co-expressing IVS GluCl β with IVS GluCl α 3 (Fig. 2C) or IVR GluCl α 3 (data not shown) were more sensitive to glutamate than the homomeric IVR GluCl α 3 channels. This reflects the contribution of a functional nonmutant GluCl β subunit to the glutamate sensitivity of the heteromeric channel. It is likely that, *in vivo*, the function of the IVR GluCl β 3 subunit may be replaced by other nonmutant subunits. In *C. elegans*, avr-14/GluCl α 3, which is expressed in extrapharyngeal neurons, can functionally replace avr-15 and inhibit pharyngeal pumping in *avr-15* mutants (Dent et al. 2000).

Of the two amino acid positions substituted in IVR GluCl β , the position 60 residue is highly conserved, being occupied by a Val in all GluCls, except for *D. melanogaster*. While the residue at position 100 is Arg in most GluCls, this position is more variable. An attempt to rescue the function of IVR GluCl β by mutating A60V was unsuccessful. It would appear therefore that the R100H mutation may be important in

determining the function of the GluCl β subunit. Whether the R100H mutation alone can abolish agonist sensitivity, or whether both mutations are required is not known, and requires further characterization. Interestingly, in *H. contortus* GluCl β , the residue at position 100, Arg, corresponds to the residue identified in our mutant non-functional subunit. Whether the *H. contortus* GluCl β subunit forms a functional receptor in *Xenopus* oocytes is not known.

Heteromeric channels formed after co-injection of IVS GluCla3 and GluClβ cRNA showed a higher sensitivity to glutamate than either of the homomeric channels, demonstrating the co-assembly of the two subunits. Interestingly, the IVR GluCla3/ IVR GluClß heteromeric channels were ~13-fold less sensitive to glutamate than the IVS GluCl α 3/ IVS GluCl β heteromeric channels. Whether the GluCl α 3 and GluCl β subunits assemble to form native receptors in the nematode is not known. LGIC receptors exist in their native form as heteromeric pentamers composed of two to four different yet closely related subunit types (Unwin 1993; Bechade et al. 1994; Macdonald and Olsen 1994; Chang et al. 1996). As members of the LGIC superfamily, the native GluCls are thought to exhibit the same heteropentameric structure. However, the exact composition of native GluCl receptors is not known. In C. elegans, the GluClß subunit is expressed solely in the pharynx (Laughton et al. 1997b), while AVR-14/GluCla3 is expressed in extrapharyngeal neurons (Dent et al. 2000), and it is therefore unlikely that these two subunits co-assemble to form a receptor. AVR-15/GluCl α 2 is expressed in the pharynx, and the native pharyngeal receptor may be composed of AVR-15 and GluClß subunits (Dent et al. 1997; Pemberton et al. 2001). As well, other glutamate-sensitive, ivermectininsensitive GluCl subunits may contribute to formation of the pharyngeal receptor complex (Pemberton et al. 2001). In *H. contortus*, GluCl α and β co-localize in motor neuron commisures, and most likely contribute to formation of the GluCl receptor at this site (Delany et al. 1998; Portillo et al. 2003). The splice variants of the *H. contortus gbr-*2/GluCl α 3 gene, GluCl α 3A and 3B, are also expressed in motor neurons (Jagannathan et al. 1999; Portillo et al. 2003). Whether or not these two splice variants co-localize with GluCl α and β subunits has not been confirmed (Portillo et al. 2003). *H. contortus* GluCl α 3B subunit, which is homologous to *C. oncophora* GluCl α 3, appears to be expressed in pharyngeal neurons (Portillo et al. 2003), and may therefore mediate ivermectin's inhibitory action on pharyngeal pumping (Geary et al. 1993). The complexity of native receptors is highlighted by a recent report that suggests co-assembly of a GluCl α subunit and the GABA-gated chloride channel Rdl subunit in some receptors in *Drosophila* head membranes (Ludmerer et al. 2002).

Both IVS and IVR GluCl α 3 homomeric receptors were sensitive to ivermectin and moxidectin. Application of saturating concentrations of glutamate and moxidectin during maximal response to ivermectin caused no further change in membrane potential. Similarly, glutamate and ivermectin did not elicit any additional current during maximal response to moxidectin, suggesting that glutamate, ivermectin and moxidectin activate the same channel. Heteromeric channels formed by IVS GluCl α 3 and β subunits gave a dose-response curve to ivermectin with an EC₅₀ of 0.5 μ M (data not shown), suggesting that the presence of the β subunit does not affect ivermectin sensitivity. Similar results have been demonstrated with *C. elegans* GluCl β had no apparent effect on the ivermectin dose response curve (Vassilatis et al. 1997). Our results show that the moxidectin dose response curves are left-shifted relative to the ivermectin dose-response curve, indicating a higher sensitivity to moxidectin than ivermectin. However, as for glutamate, ivermectin and moxidectin dose-response curves for the IVR GluCl α 3 homomeric channel are right-shifted relative to the respective IVS GluCl α 3 channel curves, with significant increases in EC₅₀ values. Interestingly, the moxidectin dose-response curve of the IVR GluCl α 3 channel was comparable to that of the IVS GluCl α 3 channel response to ivermectin. These findings suggest that while loss of sensitivity to ivermectin also results in a loss of moxidectin sensitivity, the efficacy of moxidectin in IVR is still comparable to the efficacy of ivermectin in IVS. In a dosetitration study, (Ranjan et al. 2002) showed that in *H. contortus*, resistance to ivermectin results in resistance to moxidectin, and *vice-versa*, though moxidectin resistance develops more slowly. Consequently, ivermectin resistant parasites may demonstrate sensitivity to the use-level of moxidectin.

In conclusion, we have shown that *C. oncophora* GluCl subunits form functional homomeric and heteromeric receptors when expressed in *Xenopus* oocytes. Mutation in GluCl α 3 subunit causes a modest loss of agonist sensitivity, while mutation in the GluCl β abolishes agonist sensitivity. The IVR worms used in this study are known to be resistant to ivermectin (Coles et al. 1998). In *C. elegans*, simultaneous mutations in three GluCl genes are required for high level resistance to manifest (Dent et al. 2000). It remains to be seen what changes occur in other *C. oncophora* GluCls, as well as other ivermectin targets such as the GABA receptor, that may enhance the effect of the L256F mutation, and help explain the mechanisms of ivermectin resistance.

Acknowledgements

We thank Dr. G. Coles for supplying the parasites, and D. Eshelby and G. Bingham for technical assistance. We are grateful to Dr. A. Orth for permitting us to use the electrophysiological equipment. This study was supported by FMC Corporation, Fort Dodge Animal Health and NSERC, Canada.

- Anziani O. S., Zimmermann G., Guglielmone A. A., Vazquez R. and Suarez V. (2001)
 Avermectin resistance in *Cooperia pectinata* in cattle in Argentina. *Vet Rec* 149, 58-59.
- Arena J. P., Liu K. K., Paress P. S., Schaeffer J. M. and Cully D. F. (1992) Expression of a glutamate-activated chloride current in *Xenopus* oocytes injected with *Caenorhabditis elegans* RNA: evidence for modulation by avermectin. *Brain Res Mol Brain Res* 15, 339-348.
- Arena J. P., Liu K. K., Paress P. S., Frazier E. G., Cully D. F., Mrozik H. and Schaeffer J.
 M. (1995) The mechanism of action of avermectins in *Caenorhabditis elegans*: correlation between activation of glutamate-sensitive chloride current, membrane binding, and biological activity. *J Parasitol* 81, 286-294.
- Bechade C., Sur C. and Triller A. (1994) The inhibitory neuronal glycine receptor. Bioessays 16, 735-744.
- Brejc K., van Dijk W. J., Klaassen R. V., Schuurmans M., van Der Oost J., Smit A. B. and Sixma T. K. (2001) Crystal structure of an ACh-binding protein reveals the ligand-binding domain of nicotinic receptors. *Nature* 411, 269-276.
- Campbell W. C. (1989) *Ivermectin and abamectin*, pp xviii, 363. Springer-Verlag, New York.
- Chang Y., Wang R., Barot S. and Weiss D. S. (1996) Stoichiometry of a recombinant GABAA receptor. *J Neurosci* 16, 5415-5424.

Cleland T. A. (1996) Inhibitory glutamate receptor channels. Mol Neurobiol 13, 97-136.

- Coles G. C., Stafford K. A. and MacKay P. H. (1998) Ivermectin-resistant *Cooperia* species from calves on a farm in Somerset. *Vet Rec* 142, 255-256.
- Coles G. C., Watson C. L. and Anziani O. S. (2001) Ivermectin-resistant *Cooperia* in cattle. *Vet Rec* 148, 283-284.
- Conder G. A., Thompson D. P. and Johnson S. S. (1993) Demonstration of co-resistance of *Haemonchus contortus* to ivermectin and moxidectin. *Vet Rec* **132**, 651-652.
- Corringer P. J., Galzi J. L., Eisele J. L., Bertrand S., Changeux J. P. and Bertrand D. (1995) Identification of a new component of the agonist binding site of the nicotinic alpha 7 homooligomeric receptor. *J Biol Chem* **270**, 11749-11752.
- Corringer P. J., Le Novere N. and Changeux J. P. (2000) Nicotinic receptors at the amino acid level. *Annu Rev Pharmacol Toxicol* **40**, 431-458.
- Craig T. M., Hatfield T. A., Pankavich J. A. and Wang G. T. (1992) Efficacy of moxidectin against an ivermectin-resistant strain of *Haemonchus contortus* in sheep. *Vet Parasitol* 41, 329-333.
- Cully D. F., Paress P. S., Liu K. K., Schaeffer J. M. and Arena J. P. (1996) Identification of a *Drosophila melanogaster* glutamate-gated chloride channel sensitive to the antiparasitic agent avermectin. *J Biol Chem* **271**, 20187-20191.
- Cully D. F., Vassilatis D. K., Liu K. K., Paress P. S., Van der Ploeg L. H., Schaeffer J. M. and Arena J. P. (1994) Cloning of an avermectin-sensitive glutamate-gated chloride channel from *Caenorhabditis elegans*. *Nature* **371**, 707-711.
- Dascal N. (1987) The use of *Xenopus* oocytes for the study of ion channels. *CRC Crit Rev Biochem* 22, 317-387.

- Delany N. S., Laughton D. L. and Wolstenholme A. J. (1998) Cloning and localisation of an avermectin receptor-related subunit from *Haemonchus contortus*. *Mol Biochem Parasitol* 97, 177-187.
- Dent J. A., Davis M. W. and Avery L. (1997) avr-15 encodes a chloride channel subunit that mediates inhibitory glutamatergic neurotransmission and ivermectin sensitivity in *Caenorhabditis elegans. EMBO J* 16, 5867-5879.
- Dent J. A., Smith M. M., Vassilatis D. K. and Avery L. (2000) The genetics of ivermectin resistance in *Caenorhabditis elegans*. *Proc Natl Acad Sci U S A* **97**, 2674-2679.
- Etter A., Cully D. F., Schaeffer J. M., Liu K. K. and Arena J. P. (1996) An amino acid substitution in the pore region of a glutamate-gated chloride channel enables the coupling of ligand binding to channel gating. *J Biol Chem* **271**, 16035-16039.
- Familton A. S., Mason P. and Coles G. C. (2001) Anthelmintic-resistant *Cooperia* species in cattle. *Vet Rec* 149, 719-720.
- Feng X. P., Hayashi J., Beech R. N. and Prichard R. K. (2002) Study of the nematode putative GABA type-A receptor subunits: evidence for modulation by ivermectin. J Neurochem 83, 870-878.
- Ffrench-Constant R. H., Pittendrigh B., Vaughan A. and Anthony N. (1998) Why are there so few resistance-associated mutations in insecticide target genes? *Philos Trans R Soc Lond B Biol Sci* 353, 1685-1693.
- Ffrench-Constant R. H., Rocheleau T. A., Steichen J. C. and Chalmers A. E. (1993) A point mutation in a Drosophila GABA receptor confers insecticide resistance. *Nature* **363**, 449-451.

- Fiel C. A., Saumell C. A., Steffan P. E. and Rodriguez E. M. (2001) Resistance of *Cooperia* to ivermectin treatments in grazing cattle of the Humid Pampa, Argentina. *Vet Parasitol* 97, 211-217.
- Fleming J. T., Squire M. D., Barnes T. M., Tornoe C., Matsuda K., Ahnn J., Fire A., Sulston J. E., Barnard E. A., Sattelle D. B. and Lewis J. A. (1997) *Caenorhabditis elegans* levamisole resistance genes lev-1, unc-29, and unc-38 encode functional nicotinic acetylcholine receptor subunits. *J Neurosci* 17, 5843-5857.
- Holden-Dye L. and Walker R. J. (1990) Avermectin and avermectin derivatives are antagonists at the 4-aminobutyric acid (GABA) receptor on the somatic muscle cells of *Ascaris*; is this the site of anthelmintic action? *Parasitology* **101 Pt 2**, 265-271.
- Horoszok L., Raymond V., Sattelle D. B. and Wolstenholme A. J. (2001) GLC-3: a novel fipronil and BIDN-sensitive, but picrotoxinin-insensitive, L-glutamate-gated chloride channel subunit from *Caenorhabditis elegans*. *Br J Pharmacol* **132**, 1247-1254.
- Jagannathan S., Laughton D. L., Critten C. L., Skinner T. M., Horoszok L. and Wolstenholme A. J. (1999) Ligand-gated chloride channel subunits encoded by the *Haemonchus contortus* and *Ascaris suum* orthologues of the *Caenorhabditis elegans* gbr-2 (avr-14) gene. *Mol Biochem Parasitol* **103**, 129-140.
- Kane N. S., Hirschberg B., Qian S., Hunt D., Thomas B., Brochu R., Ludmerer S. W., Zheng Y., Smith M., Arena J. P., Cohen C. J., Schmatz D., Warmke J. and Cully D.
 F. (2000) Drug-resistant *Drosophila* indicate glutamate-gated chloride channels are targets for the antiparasitics nodulisporic acid and ivermectin. *Proc Natl Acad Sci U S A* 97, 13949-13954.

- Laughton D. L., Lunt G. G. and Wolstenholme A. J. (1997a) Alternative splicing of a *Caenorhabditis elegans* gene produces two novel inhibitory amino acid receptor subunits with identical ligand binding domains but different ion channels. *Gene* 201, 119-125.
- Laughton D. L., Lunt G. G. and Wolstenholme A. J. (1997b) Reporter gene constructs suggest that the *Caenorhabditis elegans* avermectin receptor beta-subunit is expressed solely in the pharynx. *J Exp Biol* **200** (**Pt 10**), 1509-1514.
- Lea T. J. and Usherwood P. N. R. (1973) Effect of ibotenic acid on chloride permeability of insect muscle-fibres. *Comp Gen Pharmacol* 4, 351-363.
- Ludmerer S. W., Warren V. A., Williams B. S., Zheng Y., Hunt D. C., Ayer M. B., Wallace M. A., Chaudhary A. G., Egan M. A., Meinke P. T., Dean D. C., Garcia M. L., Cully D. F. and Smith M. M. (2002) Ivermectin and nodulisporic acid receptors in *Drosophila melanogaster* contain both gamma-aminobutyric acid-gated Rdl and glutamate-gated GluCl alpha chloride channel subunits. *Biochemistry* 41, 6548-6560.
- Macdonald R. L. and Olsen R. W. (1994) GABAA receptor channels. Annu Rev Neurosci 17, 569-602.
- McKellar Q. A. and Benchaoui H. A. (1996) Avermectins and milbemycins. J Vet Pharmacol Therap 19, 331-351.
- Ortells M. O. and Lunt G. G. (1995) Evolutionary history of the ligand-gated ion-channel superfamily of receptors. *Trends Neurosci* 18, 121-127.
- Pankavich J. A., Berger H. and Simkins K. L. (1992) Efficacy of moxidectin, nemadectin and ivermectin against an ivermectin-resistant strain of *Haemonchus contortus* in sheep. Vet Rec 130, 241-242, 243.

- Parmentier H. K., Ploeger H. W., Nieuwland M. G., Souren P. J., Van Pinxteren L. A., Rietveld F. W., De Vries Reilingh G. and Kloosterman A. (1995) Low molecular weight *Cooperia oncophora* antigens: characterization and humoral immune responses in calves mono-infected with 100,000 infective larvae. *Vet Parasitol* 59, 219-230.
- Pemberton D. J., Franks C. J., Walker R. J. and Holden-Dye L. (2001) Characterization of glutamate-gated chloride channels in the pharynx of wild-type and mutant *Caenorhabditis elegans* delineates the role of the subunit GluClalpha2 in the function of the native receptor. *Mol Pharmacol* **59**, 1037-1043.
- Portillo V., Jagannathan S. and Wolstenholme A. J. (2003) Distribution of glutamategated chloride channel subunits in the parasitic nematode *Haemonchus contortus*. J Comp Neurol 462, 213-222.
- Ranganathan R., Cannon S. C. and Horvitz H. R. (2000) MOD-1 is a serotonin-gated chloride channel that modulates locomotory behaviour in *C. elegans. Nature* **408**, 470-475.
- Ranjan S., Wang G. T., Hirschlein C. and Simkins K. L. (2002) Selection for resistance to macrocyclic lactones by *Haemonchus contortus* in sheep. *Vet Parasitol* 103, 109-117.
- Shoop W. L., Mrozik H. and Fisher M. H. (1995) Structure and activity of avermeetins and milbemycins in animal health. *Vet Parasitol* **59**, 139-156.
- Shoop W. L., Haines H. W., Michael B. F. and Eary C. H. (1993) Mutual resistance to avermectins and milbemycins: oral activity of ivermectin and moxidectin against ivermectin-resistant and susceptible nematodes. *Vet Rec* 133, 445-447.

- Unwin N. (1993) Nicotinic acetylcholine receptor at 9 Å resolution. J Mol Biol 229, 1101-1124.
- Vassilatis D. K., Arena J. P., Plasterk R. H., Wilkinson H. A., Schaeffer J. M., Cully D. F. and Van der Ploeg L. H. (1997) Genetic and biochemical evidence for a novel avermectin-sensitive chloride channel in *Caenorhabditis elegans*. Isolation and characterization. *J Biol Chem* 272, 33167-33174.
- Vermunt J. J., West D. M. and Pomroy W. E. (1996). Inefficacy of moxidectin and doramectin against ivermectin-resistant *Cooperia* spp. of cattle in New Zealand. N Z Vet Journal 44, 188-193.
- Zheng Y., Hirschberg B., Yuan J., Wang A. P., Hunt D. C., Ludmerer S. W., Schmatz D. M. and Cully D. F. (2002) Identification of two novel *Drosophila melanogaster* histamine-gated chloride channel subunits expressed in the eye. J Biol Chem 277, 2000-2005.

150

CONNECTING STATEMENT III

Apart from the GluCls, other genes have been implicated in ivermectin's mode of action and mechanism of resistance. Based on genetic variability analysis, beta-tubulin is thought to be involved in ivermectin resistance in *H. contortus*. To determine whether this gene is involved in ivermectin resistance in *C. oncophora*, we cloned two isotype cDNAs, and analyzed the genetic variability in IVS and IVR worms. However, we found no association between the two isotype genes and ivermectin resistance. Mutations in beta-tubulin are known to confer resistance to the benzimidazole class of anthelmintics. We therefore decided to analyze the two beta-tubulin isotypes for benzimidazole resistance-associated changes. The results of this analysis are presented in the following chapter.

CHAPTER 5

Cloning two full-length beta-tubulin isotype cDNAs from *Cooperia oncophora*, and screening for benzimidazole resistance-associated mutations in two isolates

Annete I. Njue and Roger K. Prichard

Parasitology (2003) 127: 579-588

Abstract

Two full-length beta-tubulin cDNAs, representing isotypes 1 and 2, were cloned from the cattle nematode Cooperia oncophora. The predicted protein sequences span 448 amino acids, and show a high degree of identity to beta-tubulins from other nematodes. While C. oncophora isotype 1 sequence had highest identity to Haemonchus contortus isotype 1 and Teladorsagia circumcincta sequences (95% identity), the C. oncophora isotype 2 sequence was most similar to H. contortus isotype 2 and Trichostrongylus colubriformis Alignment of the two C. oncophora sequences with other (92% identity). trichostrongylid beta-tubulins deposited in GenBank showed a clear distinction between isotype 1 and 2 beta-tubulin classes. The two classes differed at 19 amino acid positions, most notably at the carboxy terminus. These isotype-defining residues were conserved among different trichostrongylid species within a class. Analysis of fragments of both genes revealed a high degree of genetic variability in coding and non-coding regions. However, all nucleotide differences detected in the coding region were silent, as they did not result in any amino acid substitution. Analysis of two groups of worms for the codon 200 polymorphism associated with benzimidazole resistance revealed a proportion of worms in one of the groups bearing a tyrosine at this position.

5.1 Introduction

Cooperia oncophora (Strongylida: Trichostrongyloidea) is an important parasite of cattle and is prevalent in the temperate regions of the world (Parmentier et al. 1995). The ability of infected animals to mount an effective immune response to this parasite increases with age, and infections are usually restricted to young calves (Kloosterman, Although considered a parasite of relatively low Ploeger & Frankena, 1991). pathogenicity, infections can sometimes cause clinical disease (Vermunt, West & Pomroy, 1996; Armour et al. 1987). While broad-spectrum anthelmintic drugs have been effective in controlling Cooperia infections in cattle, their usefulness is threatened by the emergence of anthelmintic resistance. Resistance is widespread in nematode parasites of sheep, including Haemonchus contortus and Teladorsagia circumcincta (Prichard, 1994; Waller et al. 1996; Waller, 1997; Gopal, Pomroy & West, 1999), and while it has been slower to emerge in nematodes of cattle, there are several reports of its occurrence. Benzimidazole (BZ) resistance in species of Cooperia has been reported in New Zealand (Jackson et al. 1995, McKenna, 1996), while ivermectin resistance in this species has been reported in New Zealand, UK and Argentina (e. g., Vermunt et al. 1996; Coles, Stafford & MacKay, 1998; Coles, Watson & Anziani, 2001; Familton, Mason & Coles, 2001; Anziani et al. 2001; Fiel et al. 2001). Also, multiple resistance of Cooperia species against the avermectin and BZ classes has been reported in New Zealand (Vermunt, West & Pomroy, 1995). Since resistance is not yet widespread, the efficacy of anthelmintics against nematodes of cattle can be maintained if control strategies which limit the frequency of treatment are adopted (Coles, 2002). In addition, the development of sensitive tests that detect resistance at the earliest stage would enable action to be taken before significant selection had occurred.

The faecal egg count reduction test (FECRT) is the most widely used method for detecting anthelmintic resistance (Taylor, Hunt & Goodyear, 2002). This test is suitable for all types of anthelmintics, and has been used to detect all cases of anthelmintic resistance in Cooperia species reported to date. It has several limitations, including the necessity for repeat visits to affected farms, and a lack of sensitivity (Waller, 1997). Sensitive molecular tests have been described for detecting BZ resistance in H. contortus, T. circumcincta and Trichostrongylus colubriformis (Kwa, Veenstra & Roos, 1994; Elard & Humbert, 1999; Silvestre & Humbert, 2000). These tests detect a mutation (phenylalanine to tyrosine) at codon 200 of the beta-tubulin isotype 1 gene, a change which is linked to BZ resistance (Kwa et al. 1994, 1995; Elard & Humbert, 1999). Designing a molecular test for monitoring BZ resistance in C. oncophora requires knowledge of the beta-tubulin sequence. Here, we describe the cloning of isotypes 1 and 2 beta-tubulin cDNAs from this parasite, which show a high degree of sequence identity to similar sequences from other trichostrongylids. The genetic variability of both isotype genes was also analyzed using the single-strand conformation polymorphism (SSCP) method. Also, the presence of the Phe-Tyr mutation was investigated in two C. oncophora populations, and shown to occur at a low frequency in one of the populations.

5.2 Materials and Methods

5.2.1 Parasites

Two *C. oncophora* isolates (IVS and IVR), kindly provided by Dr. Coles (University of Bristol, UK) were used in this study. The IVS isolate was maintained without anthelmintic pressure at Weybridge Experimental Station, UK, while IVR represents a field isolate originally obtained from a farm in Somerset, UK, where ivermectin resistance was reported (Coles *et al.* 1998). The two isolates were maintained by regular passage through donor (male Holstein) calves at the Macdonald Campus, McGill University Farm. The animals were housed indoors. Third-stage larvae (L3) were obtained by copro-cultures (Borgsteede & Hendriks, 1979) and kept in water at 4°C until used to infect other calves. To ensure viability, L3's were less than 3 months old when used for infection. Adult *C. oncophora* were collected live at necropsy from the small intestine, washed in RPMI medium (Sigma) at 37°C, and stored in liquid nitrogen until used for RNA and DNA extraction.

5.2.2 Cloning of the full-length beta-tubulin Isotypes 1 and 2 cDNAs of C. oncophora

Total RNA was isolated from bulk adult worms using the TRIZOL method (GibcoBRL, Burlington, ON, Canada). First strand cDNA synthesis was performed using 2 μ g of the total RNA using 200 U murine Moloney leukemia virus reverse transcriptase (M-MLV, GibcoBRL) and 0.2 μ g oligo dT₁₂₋₁₈ primer (GibcoBRL). For the initial isolation of the *C. oncophora* beta-tubulin isotype 1 (Co b1) sequence, cDNA was amplified using Advantage 2 cDNA kit (ClonTech) with degenerate primers. These primers were designed based on conserved regions of beta-tubulin isotype 1 sequences from *H*.



contortus and T. circumcincta. Two sets of primers were designed for a nested PCR approach. A fragment of the Co b1 cDNA was first amplified with the outer sense primer, Co b1 Deg F1 (5' GGNCCNTAYGGNCARCTNTTYCGNC 3'), and the outer antisense primer, Co b1 Deg R1 (5' CYTCNGCRTCNAGRTCNCCCATRTC 3'). This first-round reaction was then used as template for subsequent amplification using the nested primers Co b1 Deg F2 (5' GARGGNGCNCARCTNGTNGAYAAYG 3'), and Co b1 Deg R2 (5' GNGTNAGYTCNGCNACNGTNGANGC 3'). The PCR reaction conditions were: an initial denaturation at 94°C for 30 s, followed by 30 cycles of 94°C for 20 s, 50°C for 30 s, and 72°C for 1 min, and a final extension at 72°C for 5 min. The PCR product from the second (nested) reaction was examined on a 1% agarose gel (TBE) stained with 0.5 µg/ml ethidium bromide, and the fragment of the expected length (~ 550 bp) was purified using the Nucleospin Gel Extraction Kit (ClonTech). The purified PCR products were then sub-cloned into a TA cloning vector (Invitrogen) as described by the manufacturer, and then sequenced using standard M13 forward and/or reverse primers. Three independent clones were sequenced to obtain a consensus. Based on the sequenced fragment, gene-specific primers were designed for the 5' and 3' RACE (Rapid Amplification of cDNA Ends) reactions.

To identify the 5' end of the Co b1 cDNA, a trans-spliced 22-nucleotide conserved leader sequence (SL1) (Blaxter & Liu, 1996) was used. This sequence has been identified at the 5' end of a number of nematode mRNAs. The SL1 primer (5' GGTTTAATTACCCAAGTTTGAG 3') was used along with two gene-specific antisense primers Co b1 5' RACE 1 (5' CCACAACGGTGTCGGAGACCTTTGGG 3') and Co b1 5' RACE 2 (5' CCCATACCGGATCCGGTACCTCCTC 3') in a semi-nested PCR reaction

using the Advantage 2 cDNA kit. Amplification conditions were as outlined above, with an annealing temperature of 54°C. The resulting PCR fragment was purified, ligated into a TA cloning vector (Invitrogen) and subsequently sequenced in both directions using vector primers.

To amplify the 3' end of Co b1 cDNA, a nested PCR approach was employed using the Marathon cDNA Amplification Kit (ClonTech). Two gene-specific sense primers were designed from the sequenced fragment (Co b1 3' RACE 1, 5' GGCTTCGTTCTCTGTTGTTGTTCCTTCA 3' and Co b1 3' RACE 2, 5' CCAAAGGTCTCCGACACCGTTGTGG 3'), and used with the two antisense adaptor primers AP1 and AP2, respectively, as outlined by the manufacturer.

C. oncophora isotype 2 (Co b2) was isolated using the same procedures as employed for isotype 1, except for the primers. The degenerate primers used for the initial isolation of a fragment of Co b2 cDNA were designed according to an alignment of the amino acid sequences of beta-tubulins from H. contortus (tub 12-16), H. contortus (tub 12-164), H. contortus (tub 8-9), T. circumcincta, T. colubriformis and Caenorhabditis elegans BEN-1. A ~ 600 bp fragment of the Co b2 cDNA was first amplified with primers Co b2 Deg F1 (5' GGNGCNGGNAAYAAYTTGGC 3'), and Co b2 Deg R1 (5' TCATRTTYTTNGCRTCRAAC 3'). This first-round reaction was then used as template for further amplification using the nested primers Co b2 Deg F2 (5' (5' R2 3'), Co b2 Deg GGNCAYTAYACNGARGGNGC and AANGGNACCATRTTNACNGC 3'). Following cloning and sequencing of the resultant 470 bp fragment, gene-specific primers were designed for the 5' and 3' RACE reactions. 5' RACE 1 (5' Co b2 5' RACE primers, The two antisense

CCACAAGTTGGTGCACAGAAAGCGTGGC 3') and Co b2 5' RACE 2 (5' GGTTCAACTACGGTATCGGAAACCTTGG 3') were used in a semi-nested PCR with SL1 to amplify the 5' end of the Co b2 cDNA. The two 3' RACE primers, Co b2 3' RACE 1 (5' TTGACGTTGTTCGCAAGGAGGCAGAAGG 3') and Co b2 3' RACE 2 (5' CCTTCAGGGTTTCCCACTCACGCACTCG 3'), were used with the antisense adapter primers AP1 and AP2, respectively, in a nested PCR reaction.

5.2.3 Determining the presence/absence of BZ resistance mutations in IVS and IVR C. oncophora

BZ resistance in *H. contortus* and *T. circumcincta* is proposed to be mediated by a phenylalanine-to-tyrosine mutation of beta-tubulin isotype 1 at position 200 (Kwa *et al.* 1995; Elard, Comes & Humbert, 1996; Silvestre & Humbert, 2000). The same mutation at position 167 has been shown to confer BZ resistance in *H. contortus* (Prichard, 2001). To determine whether these mutations were present in the *C. oncophora* isotype 1 (Co b1) gene, genomic DNA was isolated from individual male worms (35 IVS and 33 IVR) as described by Beech, Prichard & Scott (1994). PCR products spanning positions 167 and 200 were amplified from these samples, purified using CloneTech's Nucleospin Extraction Kit, and sequenced.

5.2.4 Analyzing genetic variability using SSCP, and sequencing of alleles

To investigate the genetic variability of the gene encoding Co b1, 60 genomic DNA samples were examined by PCR-SSCP (30 IVS and 30 IVR) using the gene-specific primers Co b1 F (5' GTAACAACTGGGCAAAGGG 3') and Co b1 R (5'

TGTCAGGGTACTCCTCACG 3'). 113 samples were tested for the Co b2 gene (57 IVS and 56 IVR) using the primers Co b2 F (5' GAAATAACTGGGCGAAGGG 3') and Co b2 R (5' ATCAGGGTACTCTTCACGG 3'). The genomic DNA samples were amplified in standard PCR reactions using Taq polymerase (GibcoBRL), and the PCR products were analyzed on a 1% agarose gel to check both the size and specificity of the products.

For SSCP screening of the Co b1 samples, 2μ l of each PCR product were mixed with 2 μ l of loading buffer containing 95% formamide, 10 mM NaOH, 0.25% bromophenol blue, and 0.25% xylene cyanol. The samples were then denatured at 95°C for 5 min, and immediately placed on ice. 4 μ l of each sample were loaded onto a 15% nondenaturing polyacrylamide gel and subjected to electrophoresis in 1XTBE for 17 hours at room temperature and 100V. For Co b2, the pcr samples were mixed with the loading dye in a product: dye ratio of 1:15. 10 μ l of each sample were loaded onto a 15% polyacrylamide gel and subjected to electrophoresis for 18 hrs at 110V and room temperature. Following electrophoresis, the gels were stained with 0.5 μ g/ml ethidium bromide and visualized using the Biorad Molecular Imager FX with its corresponding Quantity One (Version 4.2.1) software.

To determine the nucleotide sequence of the Co b1 and Co b2 alleles identified, at least three individual PCR samples representing each allele were selected, where possible, and sequenced using PCR primers.

5.2.5 Sequence analysis

Allele sequences were aligned using CLUSTAL W on the SDSC Biology Workbench (Version 3.2). Analysis of the beta-tubulin protein sequences was performed by first
generating a multiple sequence alignment using CLUSTAL W. Phylogenetic relationships were then determined using MEGA Version 2.1 (Kumar et al. 2001), and the statistical significance of the trees was tested by bootstrap analysis using 1000 replicates.

5.3 Results

The amino acid sequences deduced from Co b1 and Co b2 cDNAs are shown in Fig. 1, aligned with sequences representing isotype 1 and isotype 2 beta-tubulins from other trichostrongylids. They were all 448 amino acids long. A characteristic feature of β tubulins is the autoregulation recognition element present at the amino terminus, which is represented by the first four amino acids Met-Arg-Glu-Ile (MREI). This conserved sequence was present in the two *C. oncophora* isotypes isolated (shown in Fig. 1). Also, residues 140-146 serve as the signature pattern for α , β and γ tubulin subunits. Co b1 and Co b2 amino acid sequences differed from each other at several positions, notably at the carboxy terminus (Fig. 1). In this region, there were 12 amino acid differences between

the two isotypes. Also, 13 other amino acid differences were identified at other positions, including 18, 35, 81, 83, and 90.

Figure 1. Alignment of the Co b1 and Co b2 sequences with other trichostrongylid betatubulins. The 24 amino acids unique to isotypes 1 and 2 are indicated by a filled circle. This alignment shows that the sequences in an isotype class show higher identity to each other than they do to the other isotype sequence from the same species. The autoregulation signal, MREI (Cleveland 1988), is highlighted by a black line, as is the tubulin signature sequence at position 140 - 146. Binding of GTP to this glycine-rich region stimulates microtubule assembly (Hesse, Thiefauf & Ponstingl, 1987).

Figure 1	C. oncophora isotype 1 H. contortus isotype 1 T. circumcincta isotype 1 H. contortus isotype 2 T. colubriformis isotype 2 C. oncophora isotype 2	MREIVHVQ/ MREIVHVQ/ MREIVHVQ/ MREIVHVQ/ MREIVHVQ/ MREIVHVQ/ MREIVHVQ/	10 AGOCGNOI AGOCGNOI AGOCGNOI AGOCGNOI AGOCGNOI	• 20 GSKFWEVI GSKFWEVI GAKFWEVI GAKFWEVI GAKFWEVI	30 SDEHGI OPC SDEHGI OPC SDEHGI OPC SDEHGI OPC SDEHGI OPC	● 40 GTYKGESDLOLE GTYKGESDLOLE GTYKGESDLOLE GSYKGESDLOLE GSYKGESDLOLE GSYKGESDLOLE	50 RINVYYNEA RINVYYNEA RINVYYNEA RINVYYNEA RINVYYNEA RINVYYNEA RINVYYNEA	60 GGKYVPRAVI GGKYVPRAVI GGKYVPRAVI GGKYVPRAVI GGKYVPRAVI	70 LVDLEP LVDLEP LVDLEP LVDLEP LVDLEP
	C. oncophora isotype 1 H. contortus isotype 1 T. circumcincta isotype 1 H. contortus isotype 2 T. colubriformis isotype 2 C. oncophora isotype 2	GT MDSVRS GT MDSERS GT MDSVRS GT MDSVRS GT MDSVRS GT MDSVRS	BD O O O O O O O O O O O O O O O O O O O	PDNYVFGC PDNYVFGC PDNYVFGC PDNFVFGC PDNFVFGC PDNFVFGC	100 2 S G A G N N WA 2 S G A G N N WA	110 GHYTEGAELVD GHYTEGAELVD GHYTEGAELVD GHYTEGAELVD GHYTEGAELVD GHYTEGAELVD	120 VLDVVAKEAE VLDVVAKEAE VLDVVAKEAE VLDVVAKEAE VLDVVAKEAE	130 GCDCLOGFO GCDCLOGFO GCDCLOGFO GCDCLOGFO GCDCLOGFO	140 LTHSLG LTHSLG LTHSLG LTHSLG LTHSLG
	C. oncophora isotype 1 H. contortus isotype 1 T. circumcincta isotype 1 H. contortus isotype 2 T. colubriformis isotype 2 C. oncophora isotype 2	1 GGT GS GMG GGT GS GMG GGT GS GMG GGT GS GMG GGT GS GMG	50 ● TLLISKIR TLLISKIR TLLISKIR TLLIAKIR TLLIAKIR TLLIAKIR	160 EEYPDAIN EEYPDAIN EEYPDAIN EEYPDAIN EEYPDAIN EEYPDAIN	170 ASFSVVPSF ASFSVVPSF SSFSVVPSF SSFSVVPSF SSFSVVPSF	180 * K V S D T V V E P Y N A * K V S D T V V E P Y N A * K V S D T V V E P Y N A * K V S D T V V E P Y N A * K V S D T V V E P Y N A	190 ATLSVHQLVEN ATLSVHQLVEN ATLSVHQLVEN ATLSVHQLVEN ATLSVHQLVEN	200 VTDETFCIDN VTDETFCIDN VTDETFCIDN VTDETFCIDN VTDETFCIDN VTDETFCIDN	210 EALYDI EALYDI EALYDI EALYDI EALYDI EALYDI
	C. oncophora isotype 1 H. contortus isotype 1 T. circumcincta isotype 1 H. contortus isotype 2 T. colubriformis isotype 2 C. oncophora isotype 2	2 CFRTLKLT CFRTLKLT CFRTLKLT CFRTLKLT CFRTLKLT CFRTLKLT	20 NPTYGDLN NPTYGDLN NPTYGDLN NPTYGDLN NPTYGDLN NPTYGDLN	230 HLVSVTMS HLVSVTMS HLVSVTMS HLVSVTMS HLVSVTMS	240 GVTTCLRFF GVTTCLRFF GVTTCLRFF GVTTCLRFF GVTTCLRFF	250 GOL NADL RKLAN GOL NADL RKLAN GOL NADL RKLAN GOL NADL RKLAN GOL NADL RKLAN	260 /NMVPFPRLH/ /NMVPFPRLH/ /NMVPFPRLH/ /NMVPFPRLH/ /NMVPFPRLH/	270 FMPGFAPLS FFMPGFAPLS FFMPGFAPLS FFMPGFAPLS FFMPGFAPLS FEMPGFAPLS	280 A K G A O A A K G A Q A
	C. oncophora isotype 1 H. contortus isotype 1 T. circumcincta isotype 1 H. contortus isotype 2 T. colubriformis isotype 2 C. oncophora isotype 2	YRASTVAE YRASTVAE YRASTVAE YRALTVSE YRALTVSE YRALTVSE	90 L T Q Q MF D A L T Q Q MF D A	300 KNMMAACE KNMMAACE NNMMAACE NNMMAACE KNMMAACE KNMMAACE	310 DPRHGRYLTI DPRHGRYLTI DPRHGRYLTI DPRHGRYLTI DPRHGRYLTI	320 V A A MF R G R MS MR V A A MF R G R MS MR	330 EVDDOMMSVO VDDOMMSVO VDDOMMSVO VDDOMMSVO VDDOMMSVO VDDOMMSVO	340 NKNSSYFVEW NKNSSYFVEW NKNSSYFVEW NKNSSYFVEW NKNSSYFVEW NKNSSYFVEW	350 H PNNVK H PNNVK H PNNVK H PNNVK H PNNVK H PNNVK
	C. oncophora isotype 1 H. contortus isotype 1 T. circumcincta isotype 1 H. contortus isotype 2 T. colubriformis isotype 2 C. oncophora isotype 2	TAVCDI PP TAVCDI PP TAVCDI PP TAVCDI PP TAVCDI PP TAVCDI PP TAVCDI PP	860 RGL KMAAT RGL KMAAT RGL KMAAT RGL KMAAT RGL KMAAT	370 FVGNSTAI FVGNSTAI FVGNSTAI FVGNSTAI FVGNSTAI	380 QELFKRISI QELFKRISI QELFKRISI QELFKRISI QELFKRISI	390 E OF T AMF R R K AF E OF T AMF R R K AF	400 HWYTGEGMD HWYTGEGMO HWYTGEGMO HWYTGEGMD HWYTGEGMD	410 EMEFTEAESN EMEFTEAESN EMEFTEAESN EMEFTEAESN EMEFTEAESN EMEFTEAESN	MNDLIS MNDLIS MNDLIS MNDLVS MNDLVS MNDLVS
	C. oncophora isotype 1 H. contortus isotype 1 T. circumcincta isotype 1 H. contortus isotype 2 T. colutiformis isotype 2	E YOQYOEA E YOQYOEA E YOQYOEA E YOQYOEA E YOQYOEA	I30 ● ●● TADDMGDL TADDMGDL TADDMGDL TADDEGEM TADEEGEM	DAEGAEEF DAEGGEE/ DAEGAEEF DAEGAEEF IEGAVEND	● ● 450 ● ♥ ₽ E E ● ♥ ₽ E E ● ♥ ₽ E E T ♥ A E E T ♥ A E E	460	470	480	490

C. oncophora isotype 2

EVOQVOEATADEEGEMEGAVENDTYAEE

The two predicted protein sequences were compared with other full length beta-tubulins reported in current databases (Table 1). The predicted protein sequence of Co b1 had the highest identity (95%) with the isotype 1 sequences representing T. circumcincta and H. contortus, whereas the Co b2 predicted protein was 92% identical to isotype 2 sequences of T. colubriformis and H. contortus. Other sequences which showed high identity to the Co b1 predicted protein sequence included the horse cyathostome Cyathostomum coronatum (94%) and C. elegans BEN 1 (90%); Co b2 showed 87% and 88% identity to these two sequences respectively. Co b1 showed 87% identity to beta-tubulin sequences from the filarial nematodes Onchocerca volvulus, Dirofilaria immitis and Brugia pahangi, while Co b2 showed 86% identity to these three sequences. The two C. oncophora sequences showed 86% identity to Drosophila melanogaster beta-tubulin. Comparison of the H. contortus and C. oncophora isotype 1 and isotype 2 sequences showed that the homologous isotypes from the two species had higher identity to one another than either isotype had to the other beta-tubulin isotype from the same species; H. contortus isotype 1 and Co b1 were 95% identical at the amino acid level, while H. contortus isotype 2 (β 12-16 and β 12-164) and Co b2 were 92% identical at the amino acid level. In contrast, H. contortus isotypes 1 and 2 shared 90% identity at the amino acid level, as did the Co b1 and Co b2 sequences. Phylogenetic analysis by distancebased and parsimony methods produced trees with similar topologies, and the Neighbour-Joining tree is illustrated in Fig. 2. Isotype 1 and isotype 2 predicted protein sequences of strongylids were found in distinct clusters highly supported by bootstrap analysis (Fig. 2). Co b1 grouped with isotype 1 sequences from H. contortus, T. circumcincta, Cylicocyclus nassatus and C. coronatum with a bootstrap value of 100%. Co b2 clustered with isotype

2 sequences from *H. contortus* and *T. colubriformis* (bootstrap value 100%). Sequences from the three filarial nematodes *B. pahangi*, *D. immitis* and *O. volvulus* also formed a distinct group.

Table 1. Comparison of *C. oncophora* beta-tubulin (Co b1 and Co b2) predicted proteins with other beta-tubulins for identity. The values represent pairwise percent identity in amino acid sequence. Highest identity to a different species is shown in bold type. Co b1 had highest identity to *H. contortus* and *T. circumcincta* isotype 1 sequences (95% identity). Co b2 had highest identity to *H. contortus* β 12-16 and β 12-164, and *T. colubriformis*, which are isotype 2 sequences.

	Co b1	Co b2
Co b2	90	-
H. contortus isotype 1	95	88
T. circumcincta isotype 1	95	87
C. nassatus	94	87
C. coronatum	94	87
C. elegans BEN 1	90	88
H. contortus β 12-16	90	92
H. contortus β 12-164	90	92
T. colubriformis	90	92
C. elegans TBB 1	87	86
O. volvulus	87	86
D. immitis	87	86
B. pahangi	87	86
Mus musculus	86	85
Homo sapiens	87	86
D. melanogaster	86	86

Figure 2. Neighbour-joining tree showing the relationships among beta-tubulin predicted protein sequences. GenBank accession numbers are indicated with each sequence. The two beta-tubulin sequences cloned from *C. oncophora* are in bold type. The first sequence, *C. oncophora* isotype 1, groups with isotype 1 sequences from *H. contortus* and *T. circumcincta*. The second sequence, *C. oncophora* isotype 2, is found on a distinct branch with similar sequences from *H. contortus* and *T. colubriformis*. Bootstrap values range from 50% to 100%. Sequences were aligned using CLUSTAL W on the SDSC Biology Workbench (Version 3.2), and phylogenetic analyses performed using MEGA Version 2.1 (Kumar *et al.* 2001).



0.01

PCR fragments spanning amino acids 167 and 200 of Co b1 were amplified from individual male worms and then sequenced to determine whether the tyrosine resistance-associated mutation was present. The fragments were 310 bp in length, and 35 IVS and 33 IVR individual male worms were examined. All IVS and IVR worms were homozygous for phenylalanine (Phe/Phe) at position 167 (Table 2). At position 200, all (n=35) IVS worms were homozygous Phe/Phe. From the IVR group, eight worms were found to be heterozygous Phe/Tyr (24.2%), while one worm was homozygous Tyr/Tyr (3%). All other IVR worms (n=24) were homozygous Phe/Phe at this position (72.7%).

Table 2. Determining the genotypes of individual worms based on amino acids at positions 167 and 200 of Co b1 gene. At codon position 167, all IVS and IVR worms were homozygous for Phe. At position 200, all IVS worms were homozygous for Phe. Eight of the IVR worms were heterozygous Phe/Tyr, while one was homozygous for the BZ resistance mutation.

<u> </u>	GENOTYPES	Phe/Phe (%)	Phe/Tyr (%)	Tyr/Tyr (%)	
	PHENOTYPES	BZ-susceptible	BZ-susceptible	BZ-resistant	
CODON 167	IVS, n=35	35 (100)	0	0	
	IVR, n=33	33 (100)	0	0	
CODON 200	IVS, n=35	35 (100)	0	0	
	IVR, n=33	24 (72.7)	8 (24.2)	1 (3)	

Using SSCP analysis, 60 Co b1 and 113 Co b2 genomic DNA samples were screened for genetic variability. Figure 3 shows sequence variability of the Co b2 gene as demonstrated using SSCP. A total of nine alleles were identified by PCR-SSCP for both Co b1 and Co b2 genes (Table 3). For the Co b1 gene, all nine alleles were found in both the IVS and IVR groups. Allele A was the most abundant in both groups, and alleles F, G and H were found at lower frequencies in both groups when compared to other alleles. For the Co b2 gene, allele A was the most common in both IVS and IVR with frequencies of 0.48 and 0.46 respectively. Allele F was found only in the IVS group, and allele L was found only in IVR. These two alleles were present at low frequencies (0.03 and 0.01 respectively). Alignment of the Co b1 and Co b2 allele sequences showed nucleotide polymorphisms in the intron and exon regions (data not shown). However, all nucleotide differences in the exons were silent, as they did not result in a change in the amino acid.



Figure 3. SSCP of the Co b2 gene. The polyacrylamide gel was stained with ethidium bromide, and the banding pattern visualized using the Biorad Molecular Imager FX. Samples in this figure were all from the IVS group. Eight of the nine alleles identified for the Co b2 gene are shown here, labeled using a letter code. Homozygotes are represented by two bands (e.g., AA) and most of the heterozygotes are represented by four bands (e.g., EH). However, where two of the bands in the heterozygote have comigrated, three bands are seen (e.g., DJ).

GENE (FRAGMENT SIZE)	ALLELE	IVS	IVR
Co b1 (234 bp)		(n = 30)	(n = 30)
	A	0.37	0.4
	B	0.23	0.15
	С	0.13	0.12
	D	0.12	0.13
	E	0.05	0.1
	F	0.02	0.03
	G	0.02	0.02
	Н	0.02	0.03
	J	0.05	0.02
Co b2 (252 bp)		(n = 57)	(n = 56)
	A	0.48	0.46
	B	0.01	0.04
	С	0.18	0.21
	В	0.12	0.16
	E	0.17	0.1
	F	0.03	0
	H	0.01	0.01
	J	0.01	0.01
	L	0	0.01

Table 3. Allele frequencies for Co b1 and Co b2 genes in IVS and IVR worms. In both genes, nine alleles were identified. All nine Co b1 alleles were found in both IVS and IVR groups, while eight of the nine Co b2 alleles were found in each group.

5.4 Discussion

Beta-tubulin sequences from three sheep trichostrongylid nematodes have been reported to date, namely *H. contortus*, *T. colubriformis* and *T. circumcincta*. However, no beta-tubulin from a cattle trichostrongylid nematode had been sequenced. Here, we report the cloning of two full-length beta-tubulin cDNAs from the cattle nematode *C. oncophora*, representing isotypes 1 and 2.

Studies by Geary et al. (1992) showed that there are at least two beta-tubulin isotype classes in H. contortus. The present results show that the same applies to C. oncophora, and that a high degree of sequence identity exists between similar isotypes from both nematodes. At the amino acid level, Co b1 sequence was 95% identical to the H. contortus isotype 1, while Co b2 was 92% identical to the two H. contortus isotype 2 sequences β 12-16, and β 12-164. This high sequence conservation extended to betatubulin sequences from other organisms, including the O. volvulus, D. melanogaster, and vertebrates, where the percentage identity to both Co b1 and Co b2 ranged from 85 to 87%. This high identity across species suggests that evolutionary divergence of betatubulins is limited by functional constraints. Phylogenetic analysis placed the Co b1 sequence with isotype 1 sequences of H. contortus (Geary et al. 1992), T. circumcincta (Elard et al. 1996), C. nassatus (Pape, Samson-Himmelstjerna & Schneider, 1999) and C. coronatum (Samson-Himmelstjerna et al. 2001). Co b2 was found on a distinct branch with isotype 2 sequences of *H. contortus* (β 12-16 and β 12-164) and *T. colubriformis*. Bootstrap analysis showed strong support for these groupings. The trichostrongylid isotypes 1 and 2 can be distinguished based on differences at their carboxy termini, as well as 12 other amino acid positions. These residues are identical among species within

a single isotype class, as seen in Fig. 1, and suggest that in trichostrongyles, at least two distinct beta-tubulin isotype classes exist. Whether these two isotypes have unique functions is not yet known. In vertebrates, seven beta-tubulin isotypes have been identified, and are distinguished based on the sequence at the isotype-defining carboxy terminus (Luduena, 1993). They differ in their cellular distribution and relative stabilities and hence, *in vivo* function (Roach *et al.* 1998; Schwarz, Liggins & Luduena, 1998). The isotype-defining carboxy terminus is thought to interact with microtubule associated proteins (MAPs) and motor proteins (Nogales, Wolf & Downing, 1998; Downing & Nogales, 1998). Isotype differences found in regions other than the carboxy terminus are proposed to determine microtubule stability (Downing & Nogales, 1998).

A Phe-Tyr mutation at position 200 of beta-tubulin isotype 1 gene is the major determinant of BZ resistance in trichostrongylid nematodes (Kwa *et al.* 1995; Elard *et al.* 1996). The same mutation at position 167 has also been reported in the absence of the position 200 mutation in resistant *H. contortus* (Prichard, 2001). To determine the prevalence of the codon 167 and 200 isotype 1 mutations in IVS and IVR worms, a fragment of the Co b1 gene spanning the two amino acid positions was amplified from randomly selected individual male worms from both groups, and sequenced. The two groups were considered independently because they originated from different locations. At position 167, all individual worms (IVS and IVR) were identified as homozygous non-mutant (Phe/Phe). The mutation at position 167 appears to be rare under field conditions in trichostrongylids of sheep, and may reflect the fitness cost associated with this mutation (Silvestre & Cabaret, 2002).

At position 200 of the Co b1 gene, all 35 IVS worms were found to be homozygous Phe/Phe at position 200 (Table 2). From the IVR group, eight of the 33 worms were identified as heterozygous Phe/Tyr at this position (24.2%). One worm was homozygous Tyr/Tyr (3%), and all other worms (n=24) were homozygous Phe/Phe (72.7%). These results are similar to those obtained by Elard & Humbert (1999) with the 'SuPRO' susceptible population of T. circumcincta, where 18% of the worms were heterozygous (Phe/Tyr) and 3% were homozygous resistant (Tyr/Tyr). This population was classified as being susceptible using the FECRT to estimate BZ resistance, since no eggs were found in the faeces after BZ treatment. The presence of the BZ resistance mutation in a proportion of the IVR worms suggests that this group has the potential to develop BZ resistance if selective pressure is applied, assuming BZ resistance mechanisms to be similar in C. oncophora and other trichostrongyles. The Phe-Tyr mutation at position 200 of isotype 1 is recessive (Elard, Suave & Humbert, 1998), and heterozygous Phe/Tyr worms are therefore eliminated along with homozygous Phe/Phe worms by using the recommended drug dose. However, underdosing seems to favour survival of heterozygous Phe/Tyr susceptible worms over homozygous Phe/Phe susceptible worms (Silvestre, Cabaret & Humbert, 2001), and benzimidazole resistance is more likely to spread rapidly when lower-than-recommended drug doses are used.

A reduction in genetic variability of isotype 1 and 2 genes has been linked to resistance in *H. contortus*, *T. colubriformis* and *T. circumcincta* (see Kwa *et al.* 1993, Beech, Prichard & Scott, 1994, Grant & Mascord, 1996, Elard & Humbert, 1999). In the most resistant worms, a loss of isotype 2 is evident (Kwa *et al.* 1993; Lubega *et al.* 1994; Roos, Kwa & Grant, 1995). SSCP analysis of short Co b1 and Co b2 gene fragments

from individual *C. oncophora* male worms showed that both genes were polymorphicnine alleles were identified for both isotypes. Trichostrongylid nematodes show high genetic diversity (Blouin *et al.* 1992), and are therefore able to respond to selection pressure (Grant 1994). The variability of both isotypes in the IVS and IVR groups was comparable, even though the BZ resistance- associated mutation was found at very low frequency in the latter group. At very low levels of BZ resistance, there is no detectable loss of variability of isotype 1 and 2 genes (Kwa *et al.* 1993). Sequencing of the alleles revealed nucleotide variations in coding and non-coding regions, although all differences in the coding region were silent, suggesting that allelic variation is limited to positions that will not result in amino acid changes, as such changes may affect function.

To prevent the spread of resistance, early detection is essential. The FECRT, which is most widely used to monitor anthelmintic resistance in domestic animals, can only detect resistance when the proportion of resistant worms in the population is at least 25% (Martin, Anderson & Jarrett, 1989). This is much greater than the 1-2% suggested by mathematical modelling if resistance is to be managed by using a second drug before resistance develops to the first (Sangster *et al.* 2002). Sensitive molecular tests that detect very low resistance gene frequencies have been described for *H. contortus*, *T. circumcincta*, *T. colubriformis*, and small strongyles (Kwa *et al.* 1994; Elard & Humbert, 1999; Silvestre & Humbert, 2000; Samson-Himmelstjerna *et al.* 2002). Such a test would be useful for studying the prevalence of BZ resistance in *C. oncophora*.

The finding of BZ-resistance associated mutations in the IVR group, albeit at a low frequency, is important, since this group is resistant to ivermectin (Coles *et al.* 1998). Recent findings suggest that ivermectin-resistant *C. oncophora* are more pathogenic than

the susceptible parasites (Coles *et al.* 2001). We found that infecting calves with 10 000 IVR L3 larvae caused diarrhoea and poor body condition, while age-matched calves infected with the same number of IVS worms did not show any clinical signs (unpublished results). The potential for multiple anthelmintic resistance developing in an isolate which appears to be more pathogenic is of concern, and indicates the need for monitoring and early detection of resistance so as to maintain the efficacy of currently available anthelmintics. Unlike nematode parasites of sheep, BZ resistance is still considered to be rare among nematodes of cattle (Prichard 1994; Coles 2002). Importantly, knowledge of the *C. oncophora* beta-tubulin isotype 1 sequence will allow for the development of a sensitive molecular test that can be used to monitor the emergence of BZ resistance, so that measures can be taken to counter it before it becomes widespread.

Acknowledgements

This study was supported by Fort Dodge Animal Health and NSERC, Canada. We would like to thank Dr. G. Coles for supplying two isolates, and D. Eshelby and G. Bingham for technical assistance.

References

- ANZIANI, O. S., ZIMMERMANN, G., GUGLIELMONE, A. A., VAZQUEZ, R. & SUAREZ, V. (2001). Avermectin resistance in *Cooperia pectinata* in cattle in Argentina. *Veterinary Record* 149, 58-59.
- ARMOUR, J., BAIRDEN, K., HOLMES, P. H., PARKINS, J. J., PLOEGER, H., SALMAN, S. K. & McWILLIAM, P. N. (1987). Pathophysiological and parasitological studies on *Cooperia oncophora* in cattle. *Research in Veterinary Science* 42, 373-381.
- BEECH, R. N., PRICHARD, R. K. & SCOTT, M. E. (1994). Genetic variability of the β-tubulin genes in BZ-susceptible and –resistant strains of *Haemonchus contortus*. *Genetics* **138**, 103-110.
- BLAXTER, M. & LIU, L. (1996). Nematode spliced leaders Ubiquity, evolution and utility. *International Journal for Parasitology* **26**, 1025-1033.
- BLOUIN, M. S., DAME J. B., TARRANT, C. A. & COURTNEY, C. H. (1992). Unusual population genetics of a parasitic nematode: mtDNA variation within and among populations. *Evolution* **46**, 470-476.
- BORGSTEEDE, F. H. & HENDRIKS, J. (1979). Experimental infections with *Cooperia* oncophora (Railliet, 1918) in calves. Results of single infections with two graded dose levels of larvae. *Parasitology* **78**, 331-342.
- CLEVELAND, D. W. (1988). Autoregulated instability of tubulin mRNAs: a novel eukaryotic regulatory mechanism. *Trends in Biochemical Sciences* **13**, 339-343.

- COLES, G. C., STAFFORD, K. A. & MACKAY, P. H. S. (1998). Ivermectin-resistant *Cooperia* species from calves on a farm in Somerset. *Veterinary Record* 142, 255-256.
- COLES, G. C., WATSON, C. L. & ANZIANI, O. S. (2001). Ivermectin-resistant *Cooperia* in cattle. *Veterinary Record* 148, 283-284.
- COLES, G. C. (2002). Sustainable use of anthelmintics in grazing animals. *Veterinary Record* **151**, 165-169.
- COLES, G. C. (2002). Cattle nematodes resistant to anthelmintics: why so few cases? *Veterinary Research* **33**, 481-489.
- DOWNING, K. H. & NOGALES, E. (1998). Tubulin structure, insights into microtubule properties and functions. *Current Opinion in Structural Biology* **8**, 785-791.
- ELARD, L. & HUMBERT, J. F. (1999). Importance of the mutation of amino acid 200 of the isotype 1 β-tubulin gene in the benzimidazole resistance of the small ruminant parasite *Teladorsagia circumcincta*. *Parasitology Research* **85**, 452-456.
- ELARD, L., COMES, A. M. & HUMBERT, J. F. (1996). Sequences of β-tubulin cDNA from BZ-susceptible and –resistant strains of *Teladorsagia circumcincta*, a nematode parasite of small ruminants. *Molecular and Biochemical Parasitology* **79**, 249-253.
- ELARD, L., SUAVE, C. & HUMBERT, J. F. (1998). Fitness of BZ-resistant and susceptible worms of *Teladorsagia circumcincta*, a nematode parasite of small ruminants. *Parasitology* **117**, 571-578.
- FAMILTON, A. S., MASON, P. & COLES, G. C. (2001). Anthelmintic-resistant Cooperia species in cattle. Veterinary Record 149, 719-720.

- FIEL, C. A., SAUMELL, C. A., STEFFAN, P. E. & RODRIGUEZ, E. M. (2001). Resistance of *Cooperia* to ivermectin treatments in grazing cattle of the Humid Pampa, Argentina. *Veterinary Parasitology* 97, 211-217.
- GEARY, T. G., NULF, S. C., FAVREAU, M. A., TANG, L., PRICHARD, R. K., HATZENBUHLER, N. T., SHEA, M. H., ALEXANDER, S. J. & KLEIN, R. D. (1992). Three β-tubulin cDNAs from the parasitic nematode Haemonchus contortus. Molecular and Biochemical Parasitology 50, 295-306
- GOPAL, R. M., POMROY, W. E. & WEST D. M. (1999). Resistance of field isolates of Trichostrongylus colubriformis and Ostertagia circumcincta to ivermectin. International Journal for Parasitology 29, 781-786.
- GRANT, W. N. (1994). Genetic variation in parasitic nematodes and its implications. International Journal for Parasitology 24, 821-830.
- GRANT, W. N. & MASCORD, L. J. (1996). Beta-tubulin polymorphism and benzimidazole resistance in *Trichostrongylus colubriformis*. International Journal for Parasitology 26, 71-77.
- HESSE, J., THIEFAUF, M. & PONSTINGL, H. (1987). Tubulin sequence region beta 155-174 is involved in binding exchangeable guanosine triphosphate. *Journal of Biological Chemistry* 262, 15472-15475.
- JACKSON, R. A., TOWNSEND, K. G., PYKE, C., & LANCE, D. M. (1995). Isolation of oxfendazole resistant *Cooperia oncophora* in cattle. *New Zealand Veterinary Journal* 35, 187-189.

- KLOOSTERMAN, A., PLOEGER, H. W. & FRANKENA, K. (1991). Age resistance in calves to Ostertagia ostertagi and Cooperia oncophora. Veterinary Parasitology 39, 101-113.
- KUMAR, S., TAMURA, K., JAKOBSEN, I. B. & NEI, M. (2001). MEGA2: Molecular evolutionary genetics analysis software. *Bioinformatics* **17**, 1244-1245.
- KWA, M. S., KOOYMAN, F. N. J., BOERSEMA, J. H. & ROOS, M. H. (1993). Effect of selection for benzimidazole resistance in *Haemonchus contortus* in β-tubulin isotype 1 and isotype 2 genes. *Biochemical and Biophysical Research Communications* 191, 413-419.
- KWA, M. S. G., VEENSTRA, J. G. & ROOS, M. H. (1994). Benzimidazole resistance in *Haemonchus contortus* is correlated with a conserved mutation at amino acid 200 in β-tubulin isotype 1. *Molecular and Biochemical Parasitology* 63, 299-303.
- KWA, M. S. G., VEENSTRA, J. G., VAN DIJK, M. & ROOS, M. H. (1995). β-tubulin genes from the parasitic nematode *Haemonchus contortus* modulate drug resistance in *Caenorhabditis elegans*. Journal of Molecular Biology **246**, 500-510.
- LUBEGA, G. W., KLEIN, R. D., GEARY, T. G. & PRICHARD, R. K. (1994). *Haemonchus contortus*, The role of two β-tubulin gene subfamilies in the resistance to benzimidazole anthelmintics. *Biochemical Pharmacology* **47**, 1705-1715.
- LUDUENA, R. F. (1993). Are tubulin isotypes functionally significant. Molecular Biology of the Cell 4, 445-457.
- MARTIN, P. J., ANDERSON, N. & JARRETT, R. G. (1989). Detecting benzimidazole resistance with faecal egg count reduction tests and in vitro assays. *Australian Veterinary Journal* **66**, 236-240.

- McKENNA, P. B. (1996). Anthelmintic resistance in cattle nematodes in New Zealand, is it increasing? *New Zealand Veterinary Journal* 44, 76.
- NOGALES, E., WOLF, S. G. & DOWNING, K.H. (1998). Structure of the αβ tubulin dimer by electron crystallography. *Nature* **391**, 199-203.
- PAPE, M., SAMSON-HIMMELSTJERNA, G. VON & SCHNEIDER, T. (1999). Characterization of the beta-tubulin gene of *Cylicocyclus nassatus*. International Journal for Parasitology **29**, 191-1947.
- PARMENTIER, H. K., PLOEGER, H. W., NIEUWLAND, M. G. B., SOUREN, P. J. E., VAN PINXTEREN, L. A. H., RIETVELD, F. W., DEVRIES REILENGH, G. & KLOOSTERMAN, A. (1995). Low molecular weight *Cooperia oncophora* antigens: characterization and humoral immune responses in calves mono-infected with 100 000 infective larvae. *Veterinary Parasitology* 59, 219-230.
- PRICHARD, R. (1994). Anthelmintic resistance. Veterinary Parasitology 54, 259-268.
- PRICHARD, R. K. (2001). Genetic variability following selection of *Haemonchus* contortus with anthelmintics. *Trends in Parasitology* **17**, 445-453.
- ROACH, M. C., BOUCHER, V. L., WALSS, C., RAVDIN, P. M. & LUDUENA, R. F. (1998). Preparation of a monoclonal antibody specific for the class 1 isotype of βtubulin. *Cell Motility and the Cytoskeleton* **39**, 273-285.
- ROOS, M. H., KWA, M. S. G. & GRANT, W. N. (1995). New genetic and practical implications of selection for anthelmintic resistance in parasitic nematodes. *Parasitology Today* 11, 148-150.

- SAMSON-HIMMELSTJERNA, G. VON, HARDER, A., PAPE, M. & SCHNEIDER, T. (2001). Novel small strongyle (*Cyathostominae*) beta-tubulin sequences. *Parasitology Research* 87, 122-125.
- SAMSON-HIMMELSTJERNA, G. VON, PAPE, M., VON WITZENDORFF, C. & SCHNEIDER, T. (2002). Allele-specific PCR for the beta-tubulin codon 200 TTC/TAC polymorphism using single adult and larval small strongyle (Cyathostominae) stages. *Journal of Parasitology* **88**, 254-257.
- SANGSTER, N., BATTERHAM, P., CHAPMAN, D. H., DURAISINGH, M., LE JAMBRE, L., SHIRLEY, M., UPCROFT, J. & UPCROFT, P. (2002). Resistance to antiparasitic drugs, the role of molecular diagnosis. *International Journal for Parasitology* **32**, 637-653.
- SCHWARZ, P. M., LIGGINS, J. R. & LUDUENA, R. F. (1998). β-tubulin isotypes purified from bovine brain have different relative stabilities. *Biochemistry* **37**, 4687-4692.
- SILVESTRE, A., CABARET, J. & HUMBERT, J. -F. (2001). Effect of benzimidazole under-dosing on the resistant allele frequency in *Teladorsagia circumcincta* (Nematoda). *Parasitology* 123, 103-111.
- SILVESTRE, A. & CABARET, J. (2002). Mutation in position 167 of isotype 1 βtubulin gene of Trichostrongylid nematodes, role in benzimidazole resistance? *Molecular and Biochemical Parasitology* **120**, 297-300.
- SILVESTRE, A. & HUMBERT, J. F. (2000). A molecular tool for species identification and benzimidazole resistance diagnosis in larval communities of small ruminant parasites. *Experimental Parasitology* **95**, 271-276.

- TAYLOR, M. A., HUNT, K. R. & GOODYEAR, K. L. (2002). Anthelmintic resistance detection methods. *Veterinary Parasitology* **103**, 183-194.
- VERMUNT, J. J., WEST, D. M. & POMROY, W. E. (1995). Multiple resistance to ivermectin and oxfendazole in *Cooperia* species in cattle in New Zealand. *Veterinary Record* 137, 43-45.
- VERMUNT, J. J., WEST, D. M. & POMROY, W. E. (1996). Inefficacy of moxidectin and doramectin against ivermectin-resistant *Cooperia* spp. of cattle in New Zealand. *New Zealand Veterinary Journal* 44, 188-193.

WALLER, P. J. (1997). Anthelmintic resistance. Veterinary Parasitology 72, 391-412.

WALLER, P. J., ECHEVARRIA, F., EDDI, C., MACIEL, S., NARI, A. & HANSEN, J.
W. (1996). The prevalence of anthelmintic resistance in nematode parasites of sheep in Southern Latin America: general overview. *Veterinary Parasitology* 62, 181-187.

CHAPTER 6

General Discussion

Gastrointestinal helminth infections play a significant role in limiting livestock productivity in many parts of the world. Control relies heavily on the use of broad-Three classes are currently available for use- the spectrum anthelmintics. benzimidazoles, which were introduced in the early 1960s, the tetrahydopyrimidines/ imidazothiazoles, introduced in the 70s, and the macrocyclic lactones (MLs), introduced in the 80s. They are classified based on their modes of action, with drugs in the same class having a common mode of action and showing cross-resistance. The effectiveness of these drugs is threatened by the development of resistance. Drug exposure selects for resistance alleles that are initially present at low frequencies in the population. Over time, and with continued selection pressure, the frequency of these resistance alleles increases, ultimately leading to treatment failure. There are currently no new drugs on the market that can be used as alternatives if existing broad-spectrum anthelmintics fail. As well, with the high cost of drug development and the demands for high returns on research investments, many animal health companies have turned to developing products for the most profitable market segments (Geary and Thompson 2003). Parasites of companion animals are considered a priority, and fewer companies are currently involved in the discovery of new antiparasitic drugs. In companion animals, anthelmintics tend to

be used against a narrower spectrum of parasites as compared to domestic ruminants. While anthelmintic resistance is a major concern in domestic ruminants, it has not been reported in companion animals. With this state of affairs, efforts should be made to preserve the shelf-life of existing anthelmintics. A key aspect of this is the ability to detect resistance at a very early stage, when the frequency of resistant individuals is low. This would allow the drug in question to be withdrawn before it looses effectiveness (Sangster 2001). This requires sensitive molecular tests that can detect low levels of resistance in populations (1-2% resistant individuals), and the development of such tests requires a good understanding of the genetic basis of anthelmintic resistance.

Studies on the molecular mechanisms of benzimidazole resistance in trichostrongylid nematodes determined that this phenotype is conferred by a phenylalanine-to-tyrosine mutation at position 200 of beta-tubulin isotype 1 (Kwa et al. 1994; Kwa et al. 1995; Elard et al. 1999). This led to the development of sensitive PCR-based tests that can detect low levels of the benzimidazole resistance-associated allele in three nematode species of sheep, including *Teladorsagia circumcincta*, *Haemonchus contortus* and *Trichostrongylus colubriformis* (Elard et al. 1999; Silvestre and Humbert 2000). No such test exists for the MLs. Such a test would be particularly useful for monitoring resistance in cattle parasites, since resistance is still not widespread. Ivermectin resistance in cattle is emerging, and all cases that have so far been reported involve *Cooperia* species (Vermunt et al. 1995; Coles et al. 1998; Anziani et al. 2001; Coles et al. 2001; Familton et al. 2001; Fiel et al. 2001). The glutamate-gated chloride channels (GluCls) are important targets for the MLs and are also thought to be involved in resistance to this class of drugs (Blackhall et al. 1998; Dent et al. 2000). This thesis

describes the genetic and functional analyses of two GluCl subunits from ivermectinsusceptible and -resistant isolates of *Cooperia oncophora*, which is the important *Cooperia* species in temperate regions. As well, the cloning of two *C. oncophora* betatubulin isotype cDNAs and the screening for benzimidazole resistance-associated changes is presented.

In studying the mechanisms of anthelmintic resistance, differences between susceptible and resistant parasites can yield useful information on changes which may be linked to drug resistance (Sangster and Gill 1999). Two isolates, IVS (ivermectinsusceptible) and IVR (ivermectin-resistant) were used in the experiments described in this thesis. In a controlled test to determine the ivermectin susceptibility/ resistance status of the two isolates, the IVS isolate was found to be 100% susceptible to ivermectin (Chapter 2). In contrast, ivermectin was only 77.8% effective at reducing the fecal egg count in one of the calves infected with the IVR isolate. Of interest is the observation that the IVR isolate was more pathogenic than the IVS isolate. Calves infected with the IVR isolate showed clinical signs of parasitic gastroenteritis, including diarrhea, rough hair coats, and poor body condition. As well, despite having a higher worm burden, the untreated IVS-infected animal had a lower fecal egg count than the untreated IVRinfected animal. Similar results were reported by (Kelly et al. 1978), who found a benzimidazole-resistant strain of Haemonchus contortus to be more pathogenic than a benzimidazole-susceptible strain. However, (Barrett et al. 1998) found no difference in pathogenicity between benzimidazole-susceptible and -resistant isolates of Teladorsagia circumcincta. Whether there is any association between anthelmintic resistance and pathogenicity is not known and requires further study. The genes that contribute to pathogenicity may be unrelated to anthelmintic resistance, and the two traits may occur independently of each other. Indeed, (Coles et al. 2001) have suggested that *C. oncophora* may have changed in recent years to become more pathogenic, and while this change may not be associated with anthelmintic resistance, it may represent a 'new' cattle disease, since *Cooperia* are generally considered to be parasites of low pathogenicity (Vermunt et al. 1995). Lower pathogenicity is considered an adaptation of host and parasite co-evolution. Parasites that are more pathogenic are less well adapted and are more likely to kill their hosts, and themselves in the process (Le Jambre et al. 1982).

In order to examine the role of the C. oncophora GluCls in ivermectin resistance, two cDNAs, encoding GluCl α 3 and GluCl β subunits, were cloned (Chapter 3). These subunits showed features that identified them as members of the cys-loop superfamily of receptors, including a long N-terminal extracellular domain, four transmembrane domains and the signature 15 amino-acid dicysteine loop that gives this superfamily its name. C. oncophora GluCl α 3 showed highest sequence identity to GluCl α 3B subunits from H. contortus and C. elegans, while C. oncophora GluClß showed highest identity to similar subunits from *H. contortus* and *C. elegans*. In *C. elegans* and *H. contortus*, two GluCl α 3 subunits (3A and 3B) are produced by alternative gene splicing (Jagannathan et al. 1999; Dent et al. 2000). The 3A splice variant was not cloned from C. oncophora. It is highly likely that the C. oncophora GluCla3 gene is also alternatively spliced. During amplification of the full-length GluCl α 3, two products were amplified, and the second less-abundant product that was not cloned most likely represented the 3A splice variant, since it was similar in size (~2 kb) to the H. contortus GluCla3A cDNA amplification product (Jagannathan et al. 1999). However, the fact that this splice variant was not cloned does not change the results presented in this thesis, since the genetic variability analysis examined the gene, and the same gene encodes the two products. As well, for the functional analysis, previous studies have shown that heterologously expressed 3A subunits from *H. contortus* and *C. elegans* do not respond to ivermectin (Dent et al. 2000; Cheeseman et al. 2001).

Analysis of the genetic variability of the C. oncophora GluCl α 3 and GluCl β genes revealed that there was selection at the GluCl α 3 gene, but not the GluCl β gene, with ivermectin resistance. These results are consistent with the observation that ivermectin binds to α -type GluCl subunits, but not the β subunit (Cully et al. 1994; Cully et al. 1996; Dent et al. 1997; Vassilatis et al. 1997; Dent et al. 2000; Horoszok et al. 2001), and suggest that the C. oncophora GluCla3 gene is a target for ivermectin, and may be involved in resistance to this drug. Selection for ivermectin resistance caused a significant change in allele frequency of the GluCl α 3 gene without causing a change in the allele frequencies of the GluCl β and actin genes, suggesting that the changes seen at the GluCla3 gene may be due to ivermectin selection and not population bottlenecking. Selection did not result in a loss of genetic variability, and whether it is maintained or lost with higher levels of selection is not known. Selection for benzimidazole resistance causes a reduction in variability of beta-tubulin isotype 1 and 2 genes (Roos et al. 1990; Kwa et al. 1993b; Kwa et al. 1993a; Beech et al. 1994; Lubega et al. 1994). That genetic variability of the C. oncophora GluCla3 gene was not lost may indicate the potential for further increase in the level of resistance.

The ligand binding domain lies in the N-terminal extracellular domain of cys-loop receptors, and mutations of residues that are involved in ligand binding alter channel

properties (Corringer et al. 2000). The genetic variability analysis of the *C. oncophora* GluCl α 3 gene was based on a short fragment in the N-terminal domain. Sequencing of all GluCl α 3 alleles (nine in total) showed polymorphism at the nucleotide level, and identity at the amino acid level, indicating that changes in this region do not contribute to ivermectin resistance. Cloning of full-length alleles from IVS and IVR worms revealed the presence of three non-synonymous mutations in the N-terminal extracellular domain. Two mutations were also identified in the N-terminal domain of the IVR GluCl β subunit. To determine the functional difference between the IVS and IVR subunits, they were expressed in *Xenopus laevis* oocytes, and the results are described in Chapter 4.

Except for IVR GluCl β , all receptors formed glutamate-gated homomeric channels when expressed in *Xenopus* oocytes. The IVS and IVR GluCl α 3 homomeric channels also responded to ivermectin and moxidectin. Responses to glutamate were rapid in activation and reversible, similar to what has previously been demonstrated in *C. elegans* and *H. contortus* (Cully et al. 1994; Dent et al. 1997; Vassilatis et al. 1997; Dent et al. 2000; Horoszok et al. 2001). In contrast, responses to ivermectin and moxidectin were slower to activate and essentially irreversible, consistent with the slow rate of dissociation observed in binding studies, and it is likely that ivermectin's neuromuscular paralysis in the worm is mediated at least in part through GluCls. Ivermectin inhibits pharyngeal pumping, and induces paralysis of somatic musculature, with pharyngeal muscle being the more sensitive site (Geary et al. 1993). In *H. contortus*, the GluCl α 3B subunit, which is orthologous to the *C. oncophora* GluCl α 3, is expressed in the pharynx, and is likely to contribute to formation of the ivermectin-sensitive receptor in this site (Portillo et al. 2003). This subunit is also expressed in motor neuron commissures, and

most likely contributes to the formation of receptors which mediate ivermectin's effects on locomotion (Jagannathan et al. 1999; Portillo et al. 2003). Given the close phylogenetic relationship between *C. oncophora* and *H. contortus*, it is likely that the *C. oncophora* GluCl α 3 subunit is expressed in the same sites as the *H. contortus* GluCl α 3B subunit.

Mutations in the *C. oncophora* GluC α 3 caused modest but significant reductions in glutamate, ivermectin and moxidectin sensitivity. One of these mutations, L256F, accounted for the difference in response between the IVS and IVR GluCl α 3 subunits. A similar mutation (L-F) in voltage-gated sodium channels has been found to confer pyrethroid resistance in insects (Liu et al. 2002). This mutation is associated with lowlevel resistance, and additional sodium channel mutations, which by themselves do not confer resistance, enhance the effect of the L-F mutation, resulting in higher levels of pyrethroid resistance. Whether the L256 mutation in *C. oncophora* GluCl α 3 is modified by other mutations, or whether it enhances the effects of another, more important, mutation, is not known. As well, the prevalence of the L256F mutation in ivermectinresistant *C. oncophora* populations remains to be seen. It is also possible that mutations in other GluCls, as well as other genes, contribute independently to ivermectin resistance, with high-level resistance involving changes in several of these.

The IVS and IVR GluCla3 homomeric channels were more sensitive to moxidectin than ivermectin, though selection for ivermectin resistance also caused a reduction in sensitivity to moxidectin. These drugs both act on GluCls, and appear to have a similar mechanism of resistance. Based on our results, moxidectin may be useful

against ivermectin-resistant worms, though how useful will depend on how far the ivermectin dose-response curve has shifted to the right before moxidectin is introduced.

Studies in H. contortus have suggested a role for beta-tubulin in ivermectin resistance (Blackhall, PhD Thesis, 1999). Ivermectin is not known to bind to betatubulin, and Blackhall (PhD Thesis, 1999) proposed that changes seen at the beta-tubulin gene may have been due to a functional linkage between beta-tubulin and ivermectin targets. Vertebrate cys-loop receptors have been extensively studied, and microtubules are known to anchor receptors at synaptic junctions by interacting with clustering proteins, thereby facilitating efficient neurotransmission (Kneussel and Betz 2000). Two beta-tubulin isotype cDNAs were therefore cloned from C. oncophora for genetic variability analysis. The sequences showed a high degree of identity to beta-tubulins of other trichostrongylid species. Genetic variability analysis of the two C. oncophora betatubulin isotype genes revealed no association between beta-tubulin and ivermectin resistance, suggesting that beta-tubulin does not play a role in ivermectin resistance in this nematode. The association between beta-tubulin and benzimidazole resistance is well established, and the C. oncophora beta-tubulin sequences were therefore examined for benzimidazole resistance-associated changes (Chapter 5). There was no reduction in genetic variability of both beta-tubulin isotype 1 and 2 genes, suggesting little or no selection for benzimidazole resistance. A proportion of the IVR worms were found to carry the benzimidazole resistance allele. Benzimidazole treatment will most likely impose selective pressure, leading to an increase in frequency of these alleles. That this group (IVR) is also resistant to ivermectin suggests that the use of benzimidazoles will most likely lead to multiple (benzimidazole and ivermectin) resistance. Knowledge of

the *C. oncophora* beta-tubulin isotype 1 sequence will allow for the development of a sensitive molecular test that can be used to monitor benzimidazole resistance allele frequency in the field.

In summary, this thesis demonstrates that GluCl subunits of *C. oncophora* are targets for the MLs. Genetic variability analysis provides indirect evidence that the *C. oncophora* GluCl α 3 gene is involved in ivermectin resistance. Functional analysis demonstrates that a L-F mutation of the GluCl α 3 subunit results in loss of sensitivity to the MLs. Analysis of beta-tubulin demonstrates the presence of a benzimidazole resistance associated similar to that observed in trichostrongylids of sheep. Understanding the mechanisms of drug resistance will allow for the development of molecular tools for studying the development of resistance in worm populations. Such tests would also be useful for examining the factors which contribute to the slower rate of resistance development in cattle nematodes.

- Anziani O. S., Zimmermann G., Guglielmone A. A., Vazquez R. and Suarez V. (2001)
 Avermectin resistance in *Cooperia pectinata* in cattle in Argentina. *Vet Rec* 149, 58-59.
- Barrett M., Jackson F. and Huntley J. F. (1998) Pathogenicity and immunogenicity of different isolates of *Teladorsagia circumcincta*. *Veterinary Parasitology* **76**, 95-104.
- Beech R. N., Prichard R. K. and Scott M. E. (1994) Genetic variability of the beta-tubulin genes in benzimidazole-susceptible and -resistant strains of *Haemonchus contortus*. *Genetics* 138, 103-110.
- Blackhall W. J., Pouliot J. F., Prichard R. K. and Beech R. N. (1998) *Haemonchus contortus*: selection at a glutamate-gated chloride channel gene in ivermectin- and moxidectin-selected strains. *Exp Parasitol* **90**, 42-48.
- Blackhall W.J. (1999) Genetic variation and multiple mechanisms of anthelmintic resistance in *Haemonchus contortus*. PhD Thesis, McGill University.
- Cheeseman C. L., Delany N. S., Woods D. J. and Wolstenholme A. J. (2001) Highaffinity ivermectin binding to recombinant subunits of the *Haemonchus contortus* glutamate-gated chloride channel. *Mol Biochem Parasitol* **114**, 161-168.
- Coles G. C., Stafford K. A. and MacKay P. H. (1998) Ivermectin-resistant *Cooperia* species from calves on a farm in Somerset. *Vet Rec* 142, 255-256.
- Coles G. C., Watson C. L. and Anziani O. S. (2001) Ivermectin-resistant *Cooperia* in cattle. *Vet Rec* 148, 283-284.

- Corringer P. J., Le Novere N. and Changeux J. P. (2000) Nicotinic receptors at the amino acid level. *Annu Rev Pharmacol Toxicol* **40**, 431-458.
- Cully D. F., Wilkinson H., Vassilatis D. K., Etter A. and Arena J. P. (1996) Molecular biology and electrophysiology of glutamate-gated chloride channels of invertebrates. *Parasitology* **113 Suppl**, S191-200.
- Cully D. F., Vassilatis D. K., Liu K. K., Paress P. S., Van der Ploeg L. H., Schaeffer J. M. and Arena J. P. (1994) Cloning of an avermectin-sensitive glutamate-gated chloride channel from *Caenorhabditis elegans*. *Nature* **371**, 707-711.
- Dent J. A., Davis M. W. and Avery L. (1997) avr-15 encodes a chloride channel subunit that mediates inhibitory glutamatergic neurotransmission and ivermectin sensitivity in *Caenorhabditis elegans. Embo J* 16, 5867-5879.
- Dent J. A., Smith M. M., Vassilatis D. K. and Avery L. (2000) The genetics of ivermectin resistance in *Caenorhabditis elegans*. *Proc Natl Acad Sci U S A* **97**, 2674-2679.
- Elard L., Cabaret J. and Humbert J. F. (1999) PCR diagnosis of benzimidazolesusceptibility or -resistance in natural populations of the small ruminant parasite, *Teladorsagia circumcincta. Vet Parasitol* **80**, 231-237.
- Familton A. S., Mason P. and Coles G. C. (2001) Anthelmintic-resistant *Cooperia* species in cattle. *Vet Rec* 149, 719-720.
- Fiel C. A., Saumell C. A., Steffan P. E. and Rodriguez E. M. (2001) Resistance of *Cooperia* to ivermectin treatments in grazing cattle of the Humid Pampa, Argentina. *Vet Parasitol* 97, 211-217.
- Geary T. G. and Thompson D. P. (2003) Development of antiparasitic drugs in the 21st century. *Vet Parasitol* **115**, 167-184.
- Geary T. G., Sims S. M., Thomas E. M., Vanover L., Davis J. P., Winterrowd C. A., Klein R. D., Ho N. F. and Thompson D. P. (1993) *Haemonchus contortus*: ivermectin-induced paralysis of the pharynx. *Exp Parasitol* 77, 88-96.
- Horoszok L., Raymond V., Sattelle D. B. and Wolstenholme A. J. (2001) GLC-3: a novel fipronil and BIDN-sensitive, but picrotoxinin-insensitive, L-glutamate-gated chloride channel subunit from *Caenorhabditis elegans*. *Br J Pharmacol* **132**, 1247-1254.
- Jagannathan S., Laughton D. L., Critten C. L., Skinner T. M., Horoszok L. and Wolstenholme A. J. (1999) Ligand-gated chloride channel subunits encoded by the *Haemonchus contortus* and *Ascaris suum* orthologues of the *Caenorhabditis elegans* gbr-2 (avr-14) gene. *Mol Biochem Parasitol* 103, 129-140.
- Kelly J. D., Whitlock H. V., Thompson H. G., Hall C. A., Martin I. C. and Le Jambre L.
 F. (1978) Physiological characteristics of free-living and parasitic stages of strains of *Haemonchus contortus*, susceptible or resistant to benzimidazole anthelmintics. *Res Vet Sci* 25, 376-385.
- Kneussel M. and Betz H. (2000) Receptors, gephyrin and gephyrin-associated proteins: novel insights into the assembly of inhibitory postsynaptic membrane specializations. *J Physiol* **525**, 1-9.
- Kwa M. S., Veenstra J. G. and Roos M. H. (1993a) Molecular characterisation of betatubulin genes present in benzimidazole-resistant populations of *Haemonchus contortus. Mol Biochem Parasitol* **60**, 133-143.
- Kwa M. S., Veenstra J. G. and Roos M. H. (1994) Benzimidazole resistance in *Haemonchus contortus* is correlated with a conserved mutation at amino acid 200 in beta-tubulin isotype 1. *Mol Biochem Parasitol* 63, 299-303.

- Kwa M. S., Kooyman F. N., Boersema J. H. and Roos M. H. (1993b) Effect of selection for benzimidazole resistance in *Haemonchus contortus* on beta-tubulin isotype 1 and isotype 2 genes. *Biochem Biophys Res Commun* **191**, 413-419.
- Kwa M. S., Veenstra J. G., Van Dijk M. and Roos M. H. (1995) Beta-tubulin genes from the parasitic nematode *Haemonchus contortus* modulate drug resistance in Caenorhabditis elegans. J Mol Biol 246, 500-510.
- Le Jambre L. F., Martin P. J. and Jarrett R. G. (1982) Comparison of changes in resistance of *Haemonchus contortus* eggs following withdrawal of thiabendazole selection. *Res Vet Sci* **32**, 39-43.
- Liu Z. Q., Tan J. G., Valles S. M. and Dong K. (2002) Synergistic interaction between two cockroach sodium channel mutations and a tobacco budworm sodium channel mutation in reducing channel sensitivity to a pyrethroid insecticide. *Insect Biochem Molec* 32, 397-404.
- Lubega G. W., Klein R. D., Geary T. G. and Prichard R. K. (1994) *Haemonchus* contortus: the role of two beta-tubulin gene subfamilies in the resistance to benzimidazole anthelmintics. *Biochem Pharmacol* 47, 1705-1715.
- Portillo V., Jagannathan S. and Wolstenholme A. J. (2003) Distribution of glutamategated chloride channel subunits in the parasitic nematode *Haemonchus contortus*. J Comp Neurol 462, 213-222.
- Roos M. H., Boersema J. H., Borgsteede F. H., Cornelissen J., Taylor M. and Ruitenberg
 E. J. (1990) Molecular analysis of selection for benzimidazole resistance in the sheep
 parasite Haemonchus contortus. Mol Biochem Parasitol 43, 77-88.

Sangster N. C. (2001) Managing parasiticide resistance. Vet Parasitol 98, 89-109.

- Sangster N. C. and Gill J. (1999) Pharmacology of anthelmintic resistance. *Parasitol Today* 15, 141-146.
- Silvestre A. and Humbert J. F. (2000) A molecular tool for species identification and benzimidazole resistance diagnosis in larval communities of small ruminant parasites. *Exp Parasitol* **95**, 271-276.
- Vassilatis D. K., Arena J. P., Plasterk R. H., Wilkinson H. A., Schaeffer J. M., Cully D. F. and Van der Ploeg L. H. (1997) Genetic and biochemical evidence for a novel avermectin-sensitive chloride channel in *Caenorhabditis elegans*. Isolation and characterization. *J Biol Chem* 272, 33167-33174.
- Vermunt J. J., West D. M. and Pomroy W. E. (1995) Multiple resistance to ivermectin and oxfendazole in *Cooperia* species of cattle in New Zealand. *Vet Rec* 137, 43-45.

INTRODUCTION TO APPENDICES

Four appendices are included in this thesis. The manuscript presented in Appendix A shows the cloning of a GABA β subunit from *C. oncophora*. Because this subunit has not been implicated in ivermectin resistance, it is not included as a chapter in the thesis. However, the GABA β subunit is necessary for the formation of functional GABA receptors, and because of its relevance in functional expression studies, it is included here.

In Appendix B, data not included in the Chapter 3 manuscript is presented.

In Appendix C, the GenBank flatfiles of all the full-length sequences cloned in the course of this study are presented. Except for GABA- β , we analyzed the genetic variability of all these sequences.

Appendix D contains the Ethics Certificate (Animal Use Protocol), as required by the University for studies involving animal subjects.

APPENDIX A

Cloning a GABA-gated chloride channel cDNA from Cooperia oncophora

Annete I. Njue, Xiao-peng Feng and Roger K. Prichard

A.1 Introduction

GABA is an important inhibitory neurotransmitter in vertebrate and invertebrate nervous systems (McIntire et al. 1993). It mediates neuronal inhibition by binding to GABA_A receptors and activating channels which are selective for chloride ions. The increased permeability to chloride ions results in membrane hyperpolarization. GABA_A receptors have been identified as targets of important neuroactive drugs, including the older cyclodiene insecticides, as well as the macrocyclic lactone endectocides (Kass et al. 1980; Matsumura and Ghiasuddin 1983; Gant et al. 1987). While these two drug classes target the same receptor, they act by different mechanisms- the cyclodienes by blocking the channel, and the macrocyclic lactones by opening it (Clark et al. 1995). Like the cyclodienes, picrotoxin acts as a GABA channel blocker, and insect strains that are resistant to cyclodienes show cross-resistance to picrotoxin (Matsumura and Ghiasuddin 1983).

Macrocyclic lactone anthelmintics are widely used to treat helminth infections in domestic animals. Unfortunately, their usefulness is threatened by the development of resistance. Genetic variability studies indicate that the GABA_A receptor gene HG1 may be involved in ivermectin resistance in *Haemonchus contortus*, a nematode parasite of sheep (Blackhall et al. 2003). Using *Xenopus laevis* oocytes as an expression system, Feng et al. (2002) demonstrated that mutations in the HG1 subunit affect receptor function. The GABA β subunit is an important functional component of the GABA receptor (Sieghart et al. 1999). In their study, Feng et al. (2002) co-expressed the wildtype and mutant HG1 subunits with a GABA β subunit, since the HG1 subunits did not form functional homomeric receptors. Here, we describe the cloning of a GABA β

subunit from *C. oncophora*, which will be useful in heterologous expression studies involving the GABA receptors of this parasitic nematode.

A.2 Materials and Methods

A.2.1 Cloning of the full-length C. oncophora GABA β cDNA

Isolation of total RNA from C. oncophora and first strand cDNA synthesis were carried out as previously described (Chapters 3 and 5). For the initial isolation of the C. oncophora GABAB sequence, cDNA was amplified using Advantage 2 cDNA kit These primers were designed based on an (ClonTech) with degenerate primers. alignment of GABAB predicted protein sequences from Caenorhabditis elegans, Drosophila melanogaster, and Lymnaea stagnalis. Two sets of primers were designed for a nested PCR approach. A fragment of the GABAB cDNA was first amplified with the outer sense primer, Deg F1 (5' TAYGAYATHCGNCTNCGNCC 3'), and the outer antisense primer, Deg R1 (5' TARTTNACNGCNGCRTAYTC 3'). This first-round reaction was then used as template for subsequent amplification using the nested primers ATHTGGGTNCCNGAYTT 3'), and Deg R2 (5' (5' F2 Deg TTNACSTANGADATNCGNGG 3'). The PCR reaction conditions were: an initial denaturation at 94°C for 30 s, followed by 30 cycles of 94°C for 20 s, 50°C for 30 s, and 72°C for 1 min, and a final extension at 72°C for 5 min. The PCR product from the second (nested) reaction was examined on a 1% agarose gel (TBE) stained with 0.5 µg/ml ethidium bromide, and the fragment of the expected length (~ 570 bp) was purified using the Nucleospin Gel Extraction Kit (ClonTech). The purified PCR products were then sub-cloned into a TA cloning vector (Invitrogen) as described by the manufacturer,

and then sequenced using standard M13 forward and/or reverse primers. Three independent clones were sequenced to obtain a consensus. Based on the sequenced fragment, gene-specific primers were designed for the 5' and 3' RACE (Rapid Amplification of cDNA Ends) reactions.

To identify the 5' end of the *C. oncophora* GABAβ cDNA, The SL2 primer (5' GGTTTTAACCCAGTTACTCAAG 3') was used along with two gene-specific antisense primers 5' RACE 1 (5' TCGGACGATTCCACAGCATAAGTAC 3') and 5' RACE 2 (5' AGGTTCATCGAGCAGCTAAGCGTTG 3') in a semi-nested PCR reaction using the Advantage 2 cDNA kit. Amplification conditions were as outlined above, with an annealing temperature of 54°C. The resulting PCR fragment was purified, ligated into a TA cloning vector (Invitrogen) and subsequently sequenced in both directions using vector primers.

To amplify the 3' end of GABA β cDNA, a nested PCR approach was employed using the Marathon cDNA Amplification Kit (ClonTech). Two gene-specific sense sequenced fragment 5' (3' RACE 1, designed from the primers were 3' 2. 5' GATCGTTATGCTCTCATGGGTTCTCG 3' and RACE CACGAGGCAACCAGTGCCCGTGTTG 3'), and used with the two antisense adaptor primers AP1 and AP2, respectively, as outlined by the manufacturer.

A.3 Results

The *C. oncophora* GABA β sequence has been deposited in GenBank under the accession number AY372759. The 1815 bp sequence encodes a protein of 543 amino acids (Figure 1). The protein sequence has structural features that are characteristic of inhibitory cys-

loop receptor subunits, including a pair of cysteine residues in the N-terminal domain, and four hydrophobic transmembrane (TM) domains. The spliced leader sequence, SL2, is found at the N-terminus 44 bp upstream of the initiation codon, and a predicted signal peptide cleavage site occurs between residues 29 and 30 (Figure 1). Potential N-linked glycosylation sites are present in the N-terminal domain (residues 48, 54 and 60). Consensus sites for cAMP-dependent phosphorylation (residue 430) and protein kinase C phosphorylation (residues 374, 427, 472, 475 and 481) occur within the long intracytoplasmic loop between TMs 3 and 4. In Figure 2, the *C. oncophora* GABAβ predicted protein sequence is shown aligned with GABAβ sequences from *H. contortus* (Feng, unpublished) and *C. elegans* (Feng et al, 2002). At the amino acid level, the *C. oncophora* showed 89 % identity to the *H. contortus* sequence, and 69 % identity to the *C. elegans* sequence. The *C. oncophora* sequence is also closely related to GABAβ sequences from *D. melanogaster*, *Anopheles gambiae* and *Lymaea stagnalis*, and shows lower levels of sequence identity to vertebrate GABAβ subunits (Table 1). Figure 1. Sequence of the *C. oncophora* GABA β cDNA. The predicted amino acid sequence is shown above the nucleotide sequence. The transmembrane domains are overlined, and the signal peptide cleavage site is indicated by \clubsuit . The two cysteine residues are indicated by \clubsuit , and the possible N-linked glycosylation sites by \blacktriangledown . Potential phosphorylation sites for cAMP-dependent protein kinase (\bigtriangledown) and protein kinase C (\bigstar) are also indicated.

$\texttt{M} \ \texttt{R} \ \texttt{M} \ \texttt{R} \ \texttt{R} \ \texttt{N} \ \texttt{G} \\ \texttt{G} \texttt{G} \texttt{G} \texttt{T} \texttt{T} \texttt{T} \texttt{A} \texttt{A} \texttt{C} \texttt{C} \texttt{A} \texttt{G} \texttt{G} \texttt{C} \texttt{G} \texttt{C} \texttt{G} \texttt{G} \texttt{G} \texttt{G} \texttt{G} \texttt{G} \texttt{G} G$
SKTVFLVSITLLLFSFTFTRGSEERMQVDR TCAAAGACCGTTTTTCTCAATCACTCTGCTTCTTTTCTCATTCACGTCCACGAGGATCCGAAGAGGATGCAAGTCGATCGA
I R G G T R S L R N S T A S Q N K T S Y S N A S S L L A D L ATCCGCGGTGGAACGCGTTCACTTCGAAATTCAACGGCTTCCCAAAATAAAACGTCGTACTCGAATGCGAGTTCACTTCTAGCGGATCTT
L A D Y D I R L R P G F G G D A L L L T M D I I I A S F D S CTGGCTGATTATGATATTCGGCTACGTCCAGGATTCGGTGGCGATGCATTACTTCTCACAATGGATATCATTATCGCTTCGTTCG
I S E V N M D Y T I T M Y L H Q Y W T D E R L S W G G S V P ATATCAGAAGTCAACATGGACTACACGATCACCATGTATTTACACCAATACTGGACGGAC
I D E M T L S G E F S Q N I W V P D T F L A N D K H S F L H ATAGATGAAATGACTTTGAGTGGAGAATTCTCTCCAAAATATCTGGGTACCGGATACATTCCTCGCCAACGACAAACACTCATTCCTTCAT
E V T E R N K M L R I S V D G K I A Y G M R L T S T L S C S GAAGTCACTGAACGCAACAAAATGCTGAGGATTAGCGTCGATGGAAAAATCGCCTATGGAATGAGGCTCACGTCAACGCTTAGCTGCTCG
M N L R N F P L D S Q N C T V E I E S Y G Y T T S E V L M L ATGAACCTACGCAATTTTCCATTGGATTCGCAAAATTGTACGGTAGAAATAGAATCATATGGCTACACTACATCAGAAGTACTTATGCTG
W N R P K A V H G V E D V D V P Q F T I T G Y Q T E D R V V TGGAATCGTCCGAAAGCTGTTCATGGCGTTGAGGACGTTGATGTCCCACAATTCACCATCACTGGTTATCAAACAGAAGATCGTGTTGTC
S T A T G S Y Q R L S L V F Q L Q R S V G Y F I F Q T Y L P AGCACAGCTACAGGATCGTATCAGCGGCTGTCACTAGTTTTCCAACGTCAGTTGGCTATTTTATTTTTCAAACATATCTACCT
C V L I V M L S W V S F W I N H E A T S A R V A L G I T T V TGTGTTTTGATCGTTATGCTCTCATGGGTCTCGTTCTGGATTAATCACGAGGCAACCAGTGCCCGTGTTGCTTTAGGCATCACTACCGTA
L T M T T I S T G V R Q S L P R I S Y V K S I D V Y L V M C CTCACTATGACAACAATCTCTGCTGCGCGCCAATCACTGCCTCGAATCAGTTATGTGAAAAGTATCGATGTGTATCTGGTGATGTGT
F V F V F A A L L E Y A A V N Y S Y W G R R S R G G G E E G TTTGTGTTCGTCTTGCTGCGTTTACTTGAATATGCGGCCGTGGAACTATTCGTACTGGGGTCGGAGAACGAGGAGGGGGGGG
W P V C N S N K D D R E S A V N I K E W G V P S S L I D E L TGGCCAGTATGCAATTCCAATAAGGACGACCGTGAAAGTGCTGTCAACATCAAGGAGTGGGGTGTACCAAGCAGTCTCATTGATGAGCTT
R Q S T S D P R T H D L N S T A L A G T A P E S L S S A S R AGGCAATCCACTTCTGACCCTCGAACTCGACTCGACCGCTTTGGCTGGAACTGCTCCCGAGTCCCTTTCCTCAGCTTCGAGG \bigtriangledown
K K R Q S S P I P S L C P S G P N G I D E D D S P E Y P R Y AAAAAGCGCCAGTCAAGTCCGATACCGTCTCTTTGCCCCTCTGGACCAAATGGTATTGACGAAGATGACTCACCAGAATATCCACGATAT
A S T V Q G M R A R P S L S A R T A R R L R T T R L R A Q S GCCTCAACCGTTCAGGGTATGAGAGCCAGGCCGTCTTTATCAGCAAGGACCGCACGACGACGACGACAAGGTTGAGAGCGCAAAGT
M T M S L H R M G V R A R K A L P R I R V R D V N V I D K Y ATGACGATGTCGTTGCATCGGATGGGTGTGCGAGCTCGCAAAGCGCTGCCTCGTATTCGGGTCCGAGACGTGAACGTCATCGACAAGTAC
S R T V F P I C F V I F N I F Y W G Y Y S I I Q L (543 aa) TCGCGGACCGTCTTTCCTATTTGTTTTTGTGATTTTCAATAPTTTCTACTGGGGTTACTACTCCATCATACAGCTGTAAATGACCGGTATC
GATAGCCAAAGCTACATCAGTGTTCCTGCTCAACATGAAGCATCCCCTTTTAAAGAAATCAATAAAGATCTCAAAAACAAAAAAAA

AAAAAAAAAAAAAAA (1815 nt)

Figure 2. Alignment of GABA β predicted amino acid sequences from *C. oncophora*, *H. contortus* and *C. elegans*. The grey shading represents amino acid identity. Two cysteine residues are found in the N-terminal extracellular domain (indicated by $\textcircled{\bullet}$). The four transmembrane domains (TM1-4) are also shown.

Figure 2

C. oncophora GABA β	MRMARRINGSKTVFLVSITLLLFSFTFTRGSEERMOVDRIRGGTRSLRNSTALSONKTSVSN 60
H. contortus GABA β	MOTIRGSGSKPVFLASITLLLFSFTFTRGSEEKTOIDRLRGGARSLRNSTIR-NRTSVSN 59
C. elegans GABA β	MRRSKTRRIFHVSITSLLVSTIFCONGTKPHNNSTSDOMSSSSWSNASQTMVSN 53
C. oncophora GABA β	61 ASSLLADLLADYDI RLRPGFGGDALLLTMDI I I ASFDSI SEVNMDYTI TMYLHQYWTDER 120
H. contortus GABA β	60 ASSLLADLLADYDI RLRPGFGGDALLLTMDI I I ASFDSI SEVNMDYTI TMYLHQYWTDER 119
C. elegans GABA β	54 ASSLLSDLLDYDI RLRPGFGGDALLLTMDI I I ASFDSI SEVDMDYTL TMYLHQYWTDER 113
C. oncophora GABA β	121 LSWGGSVPIDEMTLSGEFSQNIWVPDTFLANDKHSFLHEVTERNKMLRISVDGKIAYGMR 180
H. contortus GABA β	120 LSWSSAVPIDEMTLSGEFSHNIWVPDTFLANDKQSFLHEVTERNKMLRISSDGKIAYGMR 179
C. elegans GABA β	114 LRWSNEI PIDEMTLSGEFSQNIWVPDTFLANDKHSYLHEVTERNKMLRINVDGKVAYGMR 173
C. oncophora GABA β H. contortus GABA β C. elegans GABA β	181 LTSTLSCSMNLRNFPLDSQNCTVEIESYGYTTSEVLMLWNRPKAVHGVED VDVPQFTITG 240 180 LTSTLSCSMNLRNFPLDSQNCTVEIESYGYTTSEVLILWNQPKAVHGVEEADVPQFTITG 239 174 LTSTLSCSMNLRNFSLDSQNCTVEIESYGYTTSEVLMKWNYPLAVHGVEQADVPQFTITG 233 174 LTSTLSCSMNLRNFSLDSQNCTVEIESYGYTTSEVLMKWNYPLAVHGVEQADVPQFTITG 233
C. oncophora GABA β H. contortus GABA β C. elegans GABA β	241 YQTEDRVVSTATGSYQRLSLVFQLQRSVGYFIFQTYLPCVLIVMLSWVSFWINHEATSAR 300 240 FQTEDRVVATATGSYQRLSLVFQLQRSVGYFIFQTYLPCVLIVMLSWVSFWINHEATSAR 299 234 FHTEDSIVSTATGSYQRLSLVFQLRRSVGYFIFETYLPCOLIVMLSWVSFWINHEATSAR 299 234 FHTEDSIVSTATGSYQRLSLVFQLRRSVGYFIFETYLPCOLIVMLSWVSFWINHEATSAR 299
C. oncophora GABA β H. contortus GABA β C. elegans GABA β	IM 2TM 3301VAL GI TTVLTMTTI STGVRQSLPRI SYVKSI DVYLVMCFVFVFAALLEYAAVNYSYWGR-359300VAL GI TTVLTMTTI STGVRQSLPRI SYVKSI DVYLVMCFVFVFAALLEYAAVNYSYWGR-358294VAL GI TTVLTMTTI STGVRQSLPRI SYVKSI DI YLVMCFVFVFAALLEYAAVNYSYWGRE353
C. oncophora GABA β H. contortus GABA β C. elegans GABA β	360 RSRGGGEEGWPVCNSNKDDRESAVNIKEWGVPSSLIDELROSTSDPR THDLNSTALA 416 359 RSRGGGEEGWPVCNSNKDDRESAVNIKEWDVPSSLMNELROPTSDPR TRELNSSSTA 415 354 RGKGGGGNEWPVNGANKEDRESAVNVCKW-VPSGLMDGVPOPQ-DRRVEALEEAMSTSNT 411
C. oncophora GABA β	417 GTAPESLSSASR-KKROSSPIPSLCPSGPNGIDEDDSPEYPRYAST-VQGMRARPSL - 471
H. contortus GABA β	416 GDLSGPLSSAAK-RKROSSPIPSLCPSGPNDIDDDSPEYPRYAST-IQGMKARPSL - 470
C. elegans GABA β	412 AAGNNNFESTSKPKKRSSSPIPPLCRAGNTISEESESPDYPRYSTTSLKGARPHASLNHK 471
C. oncophora GABA β H. contortus GABA β C. elegans GABA β	472 SARTARLET TELRAQSMIMSLHEMGV RARKALPRI RVRDVNU DKYSRIVF 523 471 SART RELESTELRAQSVIMSLHEMGI RARKALPRI RVRDVNU DKYSRIVF 522 472 THHLKGRSSARAKERMTLARMNVS-MKQSISQIGRARKKVIPTI RVRDVNLIDKYSRVVF 530
C. oncophora GABA β	524 PICFVIFNIFYWGYYSIIQL 543
H. contortus GABA β	523 PVCFVIFNLFYWGYYTIIQL 542
C. elegans GABA β	531 PVCFIVFNLFYWSYYMMVPS 550

Table 1. Percentage amino-acid identity of *C*. *oncophora* GABA β with other cloned and predicted GABA receptor subunits.

SEQUENCE	% IDENTITY
H. contortus GABAβ	89
C. elegans GABAβ	69
D. melanogaster GABA β	66
An. gambiae GABAβ	64
L. stagnalis GABAβ	63
D. melanogaster Rdl	44
Human GABA ^β	40
Mouse GABAβ	39
Chicken GABAB	38

A.4 Discussion

GABA receptors are members of the ligand gated ion channel superfamily, which includes nicotinic acetylcholine receptors, 5-hydroxytryptamine₃ (5 HT-3) serotonin receptors and glycine receptors (Vassilatis et al. 1997). Receptors in this family share structural similarities. They are pentameric in structure, and are made up of distinct, but homologous subunits (Smith and Olsen 1995). Each subunit in the receptor complex consists of a large N-terminal extracellular domain, four transmembrane (TM) domains, a

large intracellular loop between TM's 3 and 4, and an extracellular carboxy terminal. Eight subunit types (α , β γ , δ , ε , π , ρ , θ) have been identified in mammals, and various combinations of these subunits form receptors which have distinct pharmacological properties and are differentially distributed within the nervous system (Macdonald and Olsen 1994; Barnard et al. 1998; Whiting 1999). Fewer GABA receptor subunit types have been cloned from invertebrates. Three GABA receptor subunits have been cloned in D. melanogaster (Hosie et al. 1997). These GABA receptors are widely distributed in the insect nervous system. Three C. elegans genes have been identified which encode GABA receptor subunits (Bamber et al. 1999), while two subunit types have been identified in the parasitic nematode H. contortus (Laughton et al. 1994). Here, we report the cloning of a GABAB receptor from C. oncophora which shows high sequence identity to GABAB subunits from H. contortus and C. elegans. The GABAB subunit is an important functional component of the GABAA receptor. While there are reports of non- β subunits (α and γ) forming functional homomeric and heteromeric channels, the β subunit is required for robust receptor expression (Sigel et al. 1990).

In *C. elegans*, GABAergic neurons control somatic musculature (McIntire et al. 1993). Avermectin causes paralysis of somatic musculature in *Ascaris suum* and *C. elegans*, and this effect is mediated at least in part through its interaction with GABA receptors (Kass et al. 1980). This effect may also be mediated partly through avermectin's interaction with glutamate-gated chloride channel (GluCl) receptors, since GluCl subunits are expressed in motor neurons (Delany et al. 1998; Jagannathan et al. 1999; Portillo et al. 2003). Genetic variability and electrophysiological studies have demonstrated the involvement of the *H. contortus* GABA HG1 gene in ivermectin

resistance (Feng et al. 2002; Blackhall et al. 2003). Attempts to clone the HG1 homolog from *C. oncophora* were unsuccessful. When expressed in *Xenopus* oocytes, the *H. contortus* HG1 subunit failed to form functional homomeric receptors, and a GABAβ subunit was necessary for the formation of functional GABA-gated and ivermectinsensitive receptors (Feng et al. 2002). Connor et al. (1998) have shown that in *Xenopus* oocytes expressing non- β subunits, the subunits are not localized to the cell membrane, and a GABA β subunit is needed for proper targeting of non- β subunits to the cell surface.

The *C. oncophora* GABA β subunit was expressed in *Xenopus* oocytes. Eggs expressing the homomeric receptors did not respond to 1 mM GABA, 1 mM glycine, 1 mM glutamate, 5 μ M ivermectin or 5 μ M moxidectin (data not shown). Similar results were described for *C. elegans* GABA β homomeric receptors (Feng et al. 2002). Some invertebrate GABA receptor subunits are able to form functional homomeric receptors when expressed in a heterologous system. Bamber et al. (1999) showed that *C. elegans* UNC-49B and UNC-49C GABA receptor subunits formed homomeric GABA receptors when expressed in *Xenopus* oocytes. The *D. melanogaster* Rdl GABA receptor subunits also formed homomeric GABA receptors in *Xenopus* oocytes (Ffrench-Constant et al. 1993). Whether the *C. oncophora* GABA β homomeric receptor is actually correctly targeted to the cell membrane requires further investigation.

In conclusion, we have cloned a GABA β subunit from *C. oncophora*, which shows a high degree of similarity to GABA β subunits from *H. contortus* and *C. elegans*. This subunit may be necessary for the formation of functional heteromeric receptors, and will therefore be useful for future electrophysiological experiments.

- Bamber B. A., Beg A. A., Twyman R. E. and Jorgensen E. M. (1999) The Caenorhabditis elegans unc-49 locus encodes multiple subunits of a heteromultimeric GABA receptor. *J Neurosci* 19, 5348-5359.
- Barnard E. A., Skolnick P., Olsen R. W., Mohler H., Sieghart W., Biggio G., Braestrup C., Bateson A. N. and Langer S. Z. (1998) International Union of Pharmacology. XV.
 Subtypes of gamma-aminobutyric acidA receptors: classification on the basis of subunit structure and receptor function. *Pharmacol Rev* 50, 291-313.
- Blackhall W. J., Prichard R. K. and Beech R. N. (2003) Selection at a gammaaminobutyric acid receptor gene in Haemonchus contortus resistant to avermectins/milbemycins. *Mol Biochem Parasitol* **131**, 137-145.
- Clark J. M., American Chemical Society. Division of Agrochemicals. and American Chemical Society. Meeting (1995) *Molecular action of insecticides on ion channels*, pp x, 356. American Chemical Society, Washington, D.C.
- Connor J. X., Boileau A. J. and Czajkowski C. (1998) A GABAA receptor alpha1 subunit tagged with green fluorescent protein requires a beta subunit for functional surface expression. *J Biol Chem* **273**, 28906-28911.
- Delany N. S., Laughton D. L. and Wolstenholme A. J. (1998) Cloning and localisation of an avermectin receptor-related subunit from Haemonchus contortus. *Mol Biochem Parasitol* 97, 177-187.

- Feng X. P., Hayashi J., Beech R. N. and Prichard R. K. (2002) Study of the nematode putative GABA type-A receptor subunits: evidence for modulation by ivermectin. J Neurochem 83, 870-878.
- Ffrench-Constant R. H., Rocheleau T. A., Steichen J. C. and Chalmers A. E. (1993) A point mutation in a Drosophila GABA receptor confers insecticide resistance. *Nature* 363, 449-451.
- Gant D. B., Eldefrawi M. E. and Eldefrawi A. T. (1987) Cyclodiene insecticides inhibit GABAA receptor-regulated chloride transport. *Toxicol Appl Pharmacol* 88, 313-321.
- Hosie A. M., Aronstein K., Sattelle D. B. and ffrench-Constant R. H. (1997) Molecular biology of insect neuronal GABA receptors. *Trends Neurosci* 20, 578-583.
- Jagannathan S., Laughton D. L., Critten C. L., Skinner T. M., Horoszok L. and Wolstenholme A. J. (1999) Ligand-gated chloride channel subunits encoded by the Haemonchus contortus and Ascaris suum orthologues of the Caenorhabditis elegans gbr-2 (avr-14) gene. *Mol Biochem Parasitol* **103**, 129-140.
- Kass I. S., Wang C. C., Walrond J. P. and Stretton A. O. (1980) Avermectin B1a, a paralyzing anthelmintic that affects interneurons and inhibitory motoneurons in Ascaris. *Proc Natl Acad Sci U S A* 77, 6211-6215.
- Laughton D. L., Amar M., Thomas P., Towner P., Harris P., Lunt G. G. and Wolstenholme A. J. (1994) Cloning of a putative inhibitory amino acid receptor subunit from the parasitic nematode Haemonchus contortus. *Receptors Channels* 2, 155-163.
- Macdonald R. L. and Olsen R. W. (1994) GABAA receptor channels. *Annu Rev Neurosci* 17, 569-602.

- Matsumura F. and Ghiasuddin S. M. (1983) Evidence for similarities between cyclodiene type insecticides and picrotoxinin in their action mechanisms. *J Environ Sci Health B* 18, 1-14.
- McIntire S. L., Jorgensen E., Kaplan J. and Horvitz H. R. (1993) The GABAergic nervous system of Caenorhabditis elegans. *Nature* **364**, 337-341.
- Portillo V., Jagannathan S. and Wolstenholme A. J. (2003) Distribution of glutamategated chloride channel subunits in the parasitic nematode Haemonchus contortus. *J Comp Neurol* **462**, 213-222.
- Sieghart W., Fuchs K., Tretter V., Ebert V., Jechlinger M., Hoger H. and Adamiker D. (1999) Structure and subunit composition of GABA(A) receptors. *Neurochem Int* 34, 379-385.
- Sigel E., Baur R., Trube G., Mohler H. and Malherbe P. (1990) The effect of subunit composition of rat brain GABAA receptors on channel function. *Neuron* **5**, 703-711.
- Smith G. B. and Olsen R. W. (1995) Functional domains of GABAA receptors. *Trends Pharmacol Sci* 16, 162-168.
- Vassilatis D. K., Elliston K. O., Paress P. S., Hamelin M., Arena J. P., Schaeffer J. M., Van der Ploeg L. H. and Cully D. F. (1997) Evolutionary relationship of the ligandgated ion channels and the avermectin-sensitive, glutamate-gated chloride channels. J Mol Evol 44, 501-508.
- Whiting P. J. (1999) The GABA-A receptor gene family: new targets for therapeutic intervention. *Neurochem Int* 34, 387-390.

APPENDIX B

SSCP analysis of intracytoplasmic loop region of Co GluClo3 and GluCl β genes

To determine the role of the Co $GluCl\alpha\beta$ and $GluCl\beta$ genes in ivermectin resistance, fragments of the two genes were examined using SSCP. Two regions of both genes were analyzed:

1. A fragment of the extracellular N-terminal domain between the two dicysteine loops (results presented in Chapter 3), and

2. A fragment of the intracytoplasmic loop between transmembrane domains 3 and 4. This was done to determine whether analysis of different regions of the gene would yield similar results. The results of this analysis are presented in this section. Analysis of the intracytoplasmic loop region of the *GluClc3* gene showed significant differences in allele frequencies between IVS and IVR worms, similar to what was seen with analysis of the N-terminal region (Chapter 3). With the *GluClβ* gene, unique alleles were identified at low frequency in both IVS and IVR groups following analysis of the intracytoplasmic loop region. Sequencing of Co *GluClc3* and *GluClβ* alleles revealed no amino acid changes, and the association of Co *GluClc3* with ivermectin resistance suggests that mutations in regions other than those analyzed here may be involved in ivermectin resistance. Indeed, sequencing of the full-length GluClc3 cDNAs from IVS and IVR worms identified amino acid changes in the N-terminal extracellular domain, in regions

other than those analyzed using SSCP. These changes were found to affect agonist sensitivity (Chapter 4). With the *GluClβ* gene, the finding of unique alleles in IVS and IVR worms at the intracytoplasmic loop region may explain the mutant IVR GluClβ allele reported in Chapter 4, which, when expressed in *Xenopus* oocytes, failed to form a functional channel.

B.1 Analysis of the Co GluClos gene

50 IVS and 53 IVR individual male worms were analyzed, and the different alleles identified by their different banding patterns (Figure 1). A total of 9 different alleles were identified (Figure 2). Allele A was the most common allele in the IVS group, with a frequency of 0.45. The frequency of this allele in the IVR group was 0.3. Allele C was the most common allele in the IVR group, with a frequency of 0.38. The frequency of this allele in the IVR group was 0.38. The frequency of this allele in the IVS group was 0.25. Chi-square analysis revealed a significant difference in allele frequencies between the two groups (p < 0.001), suggesting that the *GluClo3* gene may be involved in ivermectin resistance or linked to another gene which is involved in ivermectin resistance.

Figure 1. SSCP patterns of three individual worms at the Co *GluClo3* gene (intracytoplasmic loop region). The gel shows seven of the nine alleles identified at the region examined.



Figure 2. Co GluClo3 allele frequencies at second region analyzed.



All nine $GluCl\alpha3$ alleles were sequenced. A total of 22 positions were found to be polymorphic (Figure 3). Three of these were in the exon. All three were silent mutations, as they did not result in amino acid change.

Figure 3. Positions of polymorphic sites of the Co GluCla3 gene analyzed. 22 positions were found to be polymorphic, three of which were in the exon (marked by arrows). All were silent mutations.



B.2 Analysis of the Co $GluCl\beta$ gene

58 IVS and 58 IVR individual male worms were examined. Three alleles were detected by SSCP (Figures 4 and 5). Allele A was the most common in both groups, being found at frequencies of 0.98 and 0.97 in the IVS and IVR groups, respectively. Allele B was found only in two IVS worms, and was only present in the heterozygous state. Allele C was found only in three IVR worms, and was also only present in the heterozygous state.

Figure 4. SSCP patterns of three individual worms at the Co $GluCl\beta$ gene.







The three Co $GluCl\beta$ alleles, A, B and C, were sequenced. Unlike the Co $GluCl\alpha\beta$ region, which spanned two introns, the same region of the Co $GluCl\beta$ gene spanned only one intron. This suggests that the position of introns in the Co $GluCl\alpha\beta$ and $GluCl\beta$ genes is not conserved. Alignment of the three Co $GluCl\beta$ alleles revealed three polymorphic sites, two of which were in the intron (Figure 6). One silent exonic mutation was also identified.

Figure 6. Positions of polymorphic sites of the Co $GluCl\beta$ gene analyzed. 3 positions were found to be polymorphic, one of which was in the exon (marked by an arrow).



APPENDIX C

GenBank Flatfiles of Full-length C. oncophora cDNA Sequences Cloned

C.1 C.oncophora GluCla3 subunit

LOCUS DEFIN ACCES VERSI	S NITION S SION ION	AY3 Coo AY3 AY3	72756 peria 72756 72756	1 oncop	818 bp hora GluCl	mRNA alpha-3	lin mRNA	near IN A, complete	V 20-NOV-2003 cds.		
KEYWC	DRDS	•			h						
SOURC	CE	Coo	peria	oncop	nora						
ORGAN	JISM	Coo	peria	oncop.	nora			1	31 L 1 3 L		
		Euk Str	aryota ongyli	a; Met: .da; T:	azoa; Nemat richostrong	gyloidea;	comac ; Coc	orea; knac periidae;	Cooperia.		
REFEF	RENCE	1 ()	bases	1 to 1	1818)						
AUTHC	DRS	Nju	e,A.I.	and	Prichard,R.	к.					
TITLE	2	Gen gen	etic v es in	variab iverm	ility of gl ectin-susce	lutamate- eptible a	-gate and -	ed chloride -resistant	channel strains of		
		Coo	peria	oncop	hora						
JOURN	JAL	Unp	ublish	led							
REFEF	RENCE	2 (3	bases	1 to 1	1818)						
AUTHO	DRS	Nju	e,A.I.	and	Prichard,R.	к.					
TITLE	2	Dir	ect Su	ubmiss.	ion						
JOURN	JAL	Sub	mitted	1 (22-2	AUG-2003) 1	Institute	e of	Parasitolo	gy, McGill		
		Uni	versit	y, 21	111 Lakesł	nore Road	1, St	e. Anne de	Bellevue,		
		QC :	H9X 3V	79, Ca	nada						
FEATU	JRES	Lo	cation	/Qual:	ifiers						
5	source	1.	.1818								
		/0:	/organism="Cooperia oncophora"								
		/ m	ol_typ	e="mRl	NA"						
_		/ 0.0	o_xrei	= tax	on:2/828"						
Ĺ	.DS	20	odon e	: .tart-'	1						
		/n:	/codon_start="								
		/p.	rotein		AAR21855"						
		/t:	ransla	tion=	"MRTSVPLATE	RIGPILALI	ICIVI	TIISTVEGKE	KLKEOEIIORIL		
		NN	YDWRVR	PRGLN	ASWPDTGGPVI	VTVNIYLF	RSISK	CIDDVNMEYSA	QFTFREEWVDAR		
		LA	YGRFEL	ESTEVI	PPFVVLATSEN	IADQSQQIV	vmpd'i	FFQNEKEARF	HLIDKPNVLIRI		
		HKI	HKDGSILYSVRLSLVLSCPMSLEFYPLDRQNCLIDLASYAYTTQDIKYEWKEQNPVQQ								
		KD0	GLRQSL	PSFEL	QDVVTKYCTSI	TNTGEYSO	CARVK	(LLLRREY S YY	LIQLYIPCIMLV		
		VVS	VVSWVSFWLDKDAVPARVSLGVTTLLTMTTQASGINSKLPPVSYIKAVDVWIGVCLAF								
		${\tt IFGALLEYAVVNYYGRKEFLRKEKKKKTRLDDCVCPSERPALRLDLSTFRRRGWTPLN}$									
		RLI	LDVLGR	NADLSI	RRVDLMSRITE	PTLFTVFI	JVFYY	SVYVKQSNLE	;		
ORIGI	IN						÷				
1	ggtttaa	atta	cccaa	gtttg	agtttccgat	: ttttaco	jcgt	caaagcgcag	gtgettegte		
61	taatggo	catc	gatct	agaca	ctgcaagaca	u gcaaago	catg	cgcacttccg	tccctctggc		
121	gactcga	ata	gggcc	aatat	tggccctcat	ctgtato	cgtc	attacaatca	tetetacagt		
181	tgaggg	caag	aggaa	actta	aagaacagga	a gatcatt	ccaa	cgtattetea	acaactacga		
241	241 ttggagagtc		aggee	aaggg	gallgaacge		JCCa	gatactygag	gleeaglyee		
301 .	ggtcaca	igit	tetta	attta	ragage	. cicadaa	aca	argategeet	atacygaata		
301 401	cagtgeu	.Caa	tagaa	agagg	tacagaatt	tataata	yca stta	aggetegeet			
421 101	cgayyat	rcaa	casat	ttaga	tacctastac	, cycyyca , cttttt	rcad	aacragaage	addaegegga		
401 5/1	acatete	rato	gacaa	accaa	acqtqctcat	t cotatt	cac	aaagacgget	ctattctata		
601	tagcota	ada	ttatc	actoo	tattatccto	cccgato	atca	ttggaattct	atccattqqa		
661	tcgacaa	aat	tgcct	tatco	atcttgcate	atatgco	ctac	actactcaad	atatcaagta		
721	tgaatgo	jaaa	gagca	gaatc	ctgtccaaca	aaaggad	cggc	ctacgacagt	cgttaccaag		
781	cttcgaa	ittg	caaga	tgtcg	ttaccaagta	ctgtaco	cagc	aaaaccaata	cgggagaata		
841	cagttgt	gcc	cgggt	caaac	tgcttttaag	aagagaa	atat	agttactacc	tcattcaact		

901	ctacattcca	tgtattatgt	tagtggtcgt	ctcttgggtt	tcattctggc	tcgataagga
961	tgcggtacca	gctcgggtat	ctttgggtgt	caccacattg	ctcacaatga	caactcaagc
1021	tagcggtatc	aactccaaac	ttccacctgt	ttcctatata	aaggcggtgg	acgtgtggat
1081	cggcgtgtgt	ctggcgttca	tctttggagc	tctactcgag	tatgcagttg	tgaattatta
1141	cggtcgaaag	gaattccttc	gtaaagaaaa	gaaaaagaaa	acgcgcttgg	atgactgtgt
1201	ctgtccgtct	gaacgtcctg	ctttgcgact	tgacttgagc	acttttcgtc	ggcggggttg
1261	gaccccttta	aataggctgt	tagatgtgtt	gggtcgaaat	gccgatctct	cacggagggt
1321	ggacttaatg	tcacgaatca	cttttcccac	cctgttcaca	gttttttgg	tgttctacta
1381	ttccgtgtat	gtgaaacaga	gcaacctcga	gtaaccatta	atcttctggc	ggtattttcc
1441	gctccttttc	gtcgctacaa	tgccgtcgcg	gaccgaaacg	acagcaaggg	ctccgatgtg
1501	atcgcacgtt	ttcgatgcct	cctcacttat	tctgattcca	caattcacat	caactcactc
1561	acttcgttct	cagetcaacg	cgcattttct	ccccatccat	ccgtgcatct	tccctgacca
1621	tcctcgcttc	tttctgcttt	ctcctgccat	tctgctgatc	caactctggc	cattcaccag
1681	agccagcaca	gattcatgct	tttcttattg	tgatcattgt	tcgcagttct	tttcatatgt
1741	aagaaacaag	gcagggcttt	tgtttctctt	gtaaataaat	gaagatttga	aaaaaaaaaa
1801	aaaaaaaaaa	aaaaaaa				
11						

C.2 C.oncophora GluCl β subunit

LOCUS	3	AY3	72757 3	L480 bp	mRNA li	near IN	/ 20-NOV-2003
DEFIN	JITION	Coo	peria onco	phora GluCl	beta mRNA, o	complete cd:	5.
ACCES	SSION	AY3	72757				
VERS	ION	AY3	72757				
KEYW	ORDS						
SOUR	'E	Coo	peria oncon	ohora			
ORGAN	JISM	Coo	peria onco	phora			
011071	AT DIT	Euk	arvota: Mei	azoa: Nemat	oda: Chromae	dorea: Rhabo	ditida:
		Str	ongvlida. '	Prichostronc	vloidea: Co	operiidae: (Cooperia.
סבבבנ	FNCE	1 /	hases 1 to	1480)	<u>y</u> 101464, 00		oooportoo.
ATTIC		⊥ (. N=in	o A T and	Prichard R	ĸ		
mTmTT	7	Con	etic varial	vility of al	utamate-date	ad chloride	channel
	Ľ.	don	ecte varias	nectin_succe	ntible and	-registant (strains of
		Gen	es III Iveli	hora	perpre ana	resiscanc,	SCIULIUS OF
	тат	UDD	peria oncor ubliabod	pilora			
JUUR		2 V	bagag 1 to	14901			
REFE	RENCE	<u>ل</u> (،		1400/ Daighard D	V		
AUTHO	JRS	NJU	e,A.I. and art Cubric	Pricharu, R.	κ.		
TITLE	5	Dire	ect Submiss		matituto of	Damaritalo	Macill
JOURI	IAL	Sub	mitted (22.	-AUG-2003) 1	nstitute of	Parasitoro	JY, MCGIII Dellemie
		Uni	versity, 2.	L III Lakesr	ore Road, S	ce. Anne de	Bellevue,
		QC 1	H9X 3V9, Ca	inada			
FEATU	JRES	LO	cation/Qua.	lifiers			
5	source	1.	.1480	•	1		
		/0:	rganısm="Co	poperia onco	phora"		
		/m	ol_type="ml	RNA"			
		/d.	b_xref="tax	con:27828"			
C	CDS	36	1334				
		/c	odon_start=	=1			
		/p:	roduct="Glu	ICl beta"			
		/p:	rotein_id='	'AAR21856"			
		/t:	ranslation=	= "MTFALVFTVA	TVAVMADSSHV	SRRSSGGTQEQI	EILNELLSNYDM
		RVI	RPPPTNYSDP	GPVTVRVNIMI	RMLSKIDVVNM	EYSMQLTFREQU	WLDSRLAYARLG
		YHI	NPPKFLTVPH	KSNLWIPDTFF	PTEKAAHRHLI	DTDNMFLRIHPI	OGKVLYSSRISI
		TS	SCHMQLQLYPI	DLQFCDFDLVS	YAHTMKDIVYQ	WDPTAPVQLKP	GVGSDLPNFQLT
		NI	FINDDCTSHT	ITGSYACLRMQL	TLKRQFSYYLV	QLYGPTTMIVI	VSWVSFWIDMHS
		TA(GRVALGVTTLI	TMTTMQAAINA	KLPPVSYVKVV	DVWLGACQTFVI	FGALLEYÁFVSY
		QD:	SQRQTDQAKSI	RATRKAQKRRAÇ	MELSERDHYQP	PCTCHLYQDYE	PTLRDRLRRYFT
		KPI	OYLPAKIDYYA	RFCVPLGFLAF	NAIYWTSCLVM	VSRLI"	
ORIGI	IN						
1	ggtttaa	tta	cccaagtttg	aggetegeag	taccgatgac	cttcgcgttg	gtgttcactg
61	tagetac	ggt	ggccgttatg	gccgattcgt	cccacgtctc	gcgtcgatcc	agtggtggca
121	cccaaga	gca	agagattett	aacgaactgo	tatccaatta	cgacatgcga	gttcgaccac
181	cgcccac	caa	ctactcagat	ccaacaggto	cagtgacagt	tcgggtgaat	atcatgatta
241	gaatgtt	gtc	aaaaattgat	gttgtcaaca	tggaatacag	tatgcaatta	acatttcgcg
301	aacaatq	act	cgattcgcgt	ctggcatatg	ctcgccttgg	ctaccataac	cctccaaagt
361	ttctcac	tqt	accccatata	aaaagcaacc	tatggatccc	tgataccttc	tttccgacag
421	agaaage	tqc	acatcggcat	ctcatcgata	cggacaacat	gttcctgcga	atacatccag
481	atggaaa	aat	gttgtacago	aqccqaatta	gtatcacgag	ctcqtqccac	atgcagette
541	aactcta	tcc	attogattto	cagttetgtg	attttgacct	agtcagctat	gctcacacaa
601	tgaagga	tat	tatatacac	tgggatecca	coocaccaot	gcagetcaaa	ccagacatca
661	gaagga	cct	gcctaacttt	cagetaacca	atatcactac	aaatgacgac	tgcactagtc
721	acaccaa	tac	tagatcatac	gectatetea	gaatgcaget	taccettaaa	agacaattca
781	attatta	tet	agtccagtta		caacgatgat	agtaattott	tcatgggtat
841	cattta	aat	coatatocat	tetacoreto	atcatateac	cttaggegte	accacactot
	Secury	300	-galacycat		22-2-4-4-		

C.3 C.oncophora beta-tubulin isotype 1

LOCUS	AY2	59994 14	470 bp :	mRNA lir	near INV	/ 09-OCT-2003
DEFINIT ACCESSI	ION Coo ON AY2	peria oncopl 59994	hora beta-t	ubulin isoty	ype 1 mRNA,	complete cds
VERSION KEYWORD	AY2 S .	59994 GI:37	594691			
SOURCE	Coo	peria oncopl	hora			
ORGANIS	M Coo	peria oncopl	hora			
	Euk	aryota; Meta	azoa; Nemat	oda; Chromad	lorea; Rhabo	litida;
	Str	ongylida; T	richostrong	yloidea; Coo	operiidae; (Cooperia.
REFEREN	CE 1 (bases 1 to 3	1470)			
AUTHORS	Nju	e,A.I. and D	Prichard,R.	Κ.		
TITLE	Clo	ning two fu	ll-length b	eta-tubulin	isotype cD1	NAs from
	Coo	peria oncopl	hora, and s	creening for	r benzimidaz	zole
	res	istance-ass	ociated mut	ations in tu	wo isolates	
JOURNAL	Par	asitology 12	27 (2003) I	n press		
REFEREN	CE 2 (bases 1 to 3	1470)			
AUTHORS	Nju	e,A.I. and D	Prichard,R.	Κ.		
TITLE	Dir	ect Submiss:	ion			
JOURNAL	Sub	mitted (21-1	MAR-2003) I	nstitute of	Parasitolog	gy, McGill
	Uni	versity, 21	,111 Lakesh	ore Road, St	ce-Anne-de-I	Bellevue,
	QC	H9X 3V9, Cai	nada			
FEATURE	S Lo	cation/Qual:	ifiers			
sou	rce 1.	.1470				
	/0	rganism="Coo	operia onco	phora"		
	/ m	ol_type="mRI	NA"			
	/d.	b_xref="taxe	on:27828"			
CDS	42	1388				
	/c	odon_start=	1			
	/p:	roduct="beta	a-tubulin 1	sotype 1"		
	/ p:	rotein_ia="/	AAP20434"			
	/ d.	D_XFEL="GI:3		OCCNIO TO CIZEUMI		
	/し. デロ	Lanstation-	MREIVIVQAG		SVIGDINIOI QFI	UFCOSCACNININA
	KG	HVTEGAELVDM	ULDWRKEAEG	CDCLOGEOLTH	SLCCCTCSCMC	PLUTSKIREEYP
	DR	TMASESVVPSPI	VSDTVVEPYN	ATLSVHOLVEN	TDETFCIDNEAL	LYDICFRTLKLT
	NP	TYGDINHLVSV	TMSGVTTCLRF	PGOLNADLRKL	AVNMVPFPRLHI	FFMPGFAPLSAK
	GA	OAYRASTVAEL	FOOMFDAKNMM	AACDPRHGRYL	IVAAMFRGRMSI	MREVDDOMMSVQ
	NK	_ NSSYFVEWIPN	VKTAVCDIPP	RGLKMAATFVGI	NSTAIOELFKR	ISEQFTAMFRRK
	AF:	LHWYTGEGMDEN	MEFTEAESNMN	DLISEYQQYQE	ATADDMGDLDAI	EGAEEPYPEE"
BASE CO	UNT 388	a 346 c 348	g 388 t			
ORIGIN						
1 gg	tttaatta	cccaagtttg	agaactctcc	tcgactacat	catgcgtgaa	atcgttcatg
61 ta	caagccgg	tcaatgcggt	aaccagattg	gatcaaagtt	ctgggaagtg	atctccgatg
121 aa	cacggtat	ccaacccgat	ggaacataca	aaggagaatc	agacctgcag	ttagagagaa
181 tc	aatgttta	ctacaatgaa	gcacatggag	gcaaatatgt	cccacgtgct	gtacttgttg
241 at	ctcgagcc	cggaacaatg	gattctgttc	gttctggacc	atatggacaa	ctattccgtc
301 ca	gataatta	cgtgtttggc	cagtcaggag	cgggtaacaa	ctgggcaaag	ggccactata
361 cc	gagggagc	tgaactcgtc	gacaatgtcc	tagatgttgt	tcgtaaagag	gctgagggct
421 gc	gattgcct	tcagggcttc	caacttacgc	attctcttgg	aggaggtacc	ggatccggta
481 tg	ggcacttt	gcttatttca	aaaattcgtg	aggagtaccc	tgacagaatt	atggcttcgt
541 tc	tctgttgt	tccttcacca	aaggtctccg	acaccgttgt	ggaaccttac	aatgctactc
601 tt	tctgttca	ccaactggta	gaaaataccg	atgaaacgtt	ctgtattgat	aatgaagctc
661 tg	tatgatat	ctgcttccgc	acattgaagc	tcacaaaccc	aacttatgga	gateteaace
721 ac	ctagtgtc	tgtcacaatg	tccggagtca	cgacttgcct	ccgattccct	ggacaactaa
781 at	gctgatct	tcgcaagttg	gccgtgaaca	tggttccatt	ccctcgtctt	cacttcttta



841	tgcccggttt	tgccccattg	tctgctaagg	gtgctcaagc	atatcgcgct	tcaactgtcg
901	ctgagettac	acagcaaatg	ttcgatgcca	agaacatgat	ggctgcctgt	gatcctcgcc
961	atggacgcta	tcttacggtg	gctgccatgt	tccgtggtcg	catgagcatg	cgggaagtag
1021	acqaccagat	gatgtctgta	cagaacaaga	actcgtcata	tttcgttgaa	tggattccaa
1081	acaacgtgaa	aactgccgtt	tgtgatattc	ctcctcgtgg	actgaagatg	gctgctactt
1141	tcgttggtaa	ctcgactgcg	atccaagagc	tgttcaagcg	tatttcggag	cagttcacag
1201	ctatgttccg	acgcaaagct	ttccttcatt	ggtacactgg	tgagggtatg	gacgaaatgg
1261	agttcactga	agctgagtcg	aacatgaacg	atcttatctc	cgaataccaa	cagtaccagg
1321	aagccaccgc	tgacgatatg	ggcgatctcg	atgcagaagg	cgcagaagag	ccgtaccccg
1381	aggagtagtg	atctacccgt	gttgcgctgt	tttttctgtg	tcaatgcgaa	atacacattg
1441	gttgcgttaa	aaaaaaaaaa	aaaaaaaaaa			
11						

C.4 C.oncophora beta-tubulin isotype 2

LOCUS	AY259995 1620 bp mRNA linear INV 09-OCT-2003
DEFINITION	Cooperia oncophora beta-tubulin isotype 2 mRNA, complete cds
ACCESSION	AY259995
VERSION	AY259995 GI:37594693
KEYWORDS	
SOURCE	Cooperia oncophora
ORGANISM	Cooperia oncophora
	Eukaryota; Metazoa; Nematoda; Chromadorea; Rhabditida;
	Strongylida; Trichostrongyloidea; Cooperiidae; Cooperia.
REFERENCE	1 (bases 1 to 1620)
AUTHORS	Njue, A.I. and Prichard, R.K.
TITLE	Cloning two full-length beta-tubulin isotype cDNAs from
	Cooperia oncophora, and screening for benzimidazole
	resistance-associated mutations in two isolates
JOURNAL	Parasitology 127 (2003) In press
REFERENCE	2 (bases 1 to 1620)
AUTHORS	Njue, A.I. and Prichard, R.K.
TTTLE	Direct Submission
JOURNAL	Submitted (21-MAR-2003) Institute of Parasitology, McGill
00012112	University, 21,111 Lakeshore Road, Ste-Anne-de-Bellevue,
	OC H9X 3V9. Canada
FEATURES	Location/Oualifiers
source	11620
Douroo	/organism="Cooperia oncophora"
	/mol type="mRNA"
	/db xref="taxon:27828"
CDS	57 1403
600	/codon_start=1
	/product="beta-tubulin isotvpe 2"
	/protein_id="AAP20435"
	/db xref="GT: 37594694"
	/translation="MRETVHVOAGOCGNOIGAKFWEVISDEHGIOPDGSYKGESDLOL
	FRINVYYNEANGCKYVPRAVLVDLEPGTMDSVRSGPFGALFRPDNFVFGOSGAGNNWA
	KCHYTEGAELVDSVLDVVRKEAEGCDCLOGFPLTHSLGGGTGSGMGTLLIAKIREEYP
	DRIMSSESWPSPKUSDTWVEPYNATLSVHOLVENTDETFCIDNEALYDICFRTLKLT
	NPTYGDLNHLVSVTMSGVTTCLRFPGOLNADLRKLAVNMVPFPRLHFFMPGFAPLSAK
	GAOAYRALTVSELTOOMFDAKNMMAACDPRHGRYLTVAAMFRGRMSMREVDDOMMSVO
	NKNSSYFVEWTPNNVKTAVCDTPPRGLKMAATFVGNSTAIOELFKRISEOFTAMFRRK
	AFI.HWYTGFGMDEMEETEAESNMNDLUSEYOOYOEATADDDGEVEGTVENDTYAEE"
BASE COUNT	$461 = 370 = 365 = 424 \pm$
OPICIN	
1 gattta	
61 ataaaa	tcat ccacattcaa accagacaat acagaaacca gatcggagca aagttctggg
121 aggtta	tete teaceacae eccatecaee constructe atacaaaeea eaaaeteate
101 ttoat	taga acquattaat gtgtactaca acquagctaa tggaqqaaaa tatgttccac
241 gaggag	tact tattaactta gaacctagaa caatagacto tattogatoa gaacctitog
241 yaycay	tatt contrologic aactitutet transcarte transcarga aataactaga
301 grgett	
421 aggagg	and clacalagaa gyayeeyaac eeyeeyaeag egeeeeyyaa geegeeeyaa
421 aygayg	rate aggaategea activates treetaagat contraagag taccotrata
401 gcacly	tate ttetttetee atagtaeett eegeeeegat tteegeteega atagttaee
601 attaca	etac cacacttet atacaccaac ttataaaaa cacaataaa acattetaca
661 thank	and accorting accorting transport coordinate accortant
721 Degree	alya aylallilal yalalliyet leayyalall caayellall adoolatti
701 togoto	and actuated gatetageta aactuatet gaagatata costacto
/or recercing	yaca actyaatyot yatttatyta aattyyttyt taatatyyty tegitteett

841	gcttgcattt	cttcatgcct	ggatttgcac	cactgtcagc	aaaaggagct	caagectace
901	gtgcacttac	cgtttctgag	cttacacaac	agatgttcga	tgctaagaac	atgatggcag
961	cttgtgatcc	acgacatggc	cgttatctca	ccgttgcggc	tatgttcaga	ggccgtatga
1021	gcatgcgcga	ggtcgatgat	cagatgatgt	ctgtacaaaa	caagaactca	tcatacttcg
1081	ttgaatggat	ccctaacaat	gttaaaacag	ccgtgtgcga	catcccaccg	cgcggattga
1141	aaatggctgc	tactttcgtt	ggaaactcga	cagcaatcca	agaactgttc	aagcgcatct
1201	cagaacaatt	cactgccatg	ttccgtcgca	aagcettett	acattggtac	actggtgagg
1261	gtatggacga	aatggaattc	actgaagccg	aatcgaatat	gaatgatctt	gtgtctgaat
1321	atcaacaata	ccaagaagcg	actgctgatg	atgatggcga	agtagaagga	accgttgaga
1381	acgacaccta	cgcagaggag	tagaaaagca	attgcccaat	tcactatgca	tctcatttt
1441	ttgcagctat	catactgcat	acgttgtaaa	agtttctctg	aacggctgtt	ctcaaagcac
1501	tttctatttg	tgattttatt	atgatctatt	cttgctgcta	attatttctt	cgcgagcttc
1561	ccgatgagtt	tgataaagcg	aatgatcagg	ctgaaaaaaa	aaaaaaaaaa	aaaaaaaaaa
11						
C.5 C.oncophora actin

LOCUS	S	AY3	72758 1	455 bp	mRNA lin	near IN	/ 20-NOV-2003				
DEFIN	NITION	Coo	peria oncop	hora actin	mRNA, comple	ete cds.					
ACCES	SSION	AY3	72758								
VERSI	ION	AY3	72758								
KEYWO	ORDS										
SOUR	TE	Coo	peria oncop	hora							
ORGAN	NTSM	Coo	peria oncop	hora							
011011	Eukarvota: Metazoa: Nematoda: Chromadorea: Rhabditida:										
		Str	ongvlida: T	richostronc	vloidea; Co	operiidae; (Cooperia.				
REFE	RENCE	1 (bases 1 to	1455)	-	-	_				
AUTHO	DRS	Niu	Niue A I and Prichard.R.K.								
TTTT.	E.	Gen	etic variab	ility of al	utamate-gate	ed chloride	channel				
	genes in ivermectin-suscentible and -resistant strains of										
		Coo	peria oncop	hora	T						
TOUR	TAT.	Unpublished									
PEFFI	FNCE	$\frac{1}{2} (bases 1 \pm 0.1455)$									
AUTH	JBG	Niu	e.A.T. and	Prichard.R.	К.						
TTTT.	2110	Dir	ect Submiss	ion							
	ώ Μ Τ Λ Τ΄.	Sub	mitted (22-	AUG-2003) T	nstitute of	Parasitolo	rv. McGill				
0001/1	.4 7 .1.1	Uni	versity 21	111 Lakest	ore Road, St	te. Anne de	Bellevue,				
			49X 31/9 Ca	nada	ioro noud, p		,				
זידי איםים	יוסדיכ	LO	UC HYA SVY, Canada Logation (Ouslifiers								
FEAIC		1	1 1/55								
2	Source	10	.1400 raaniem="Co	operia oper	nhora"						
		/ 0. / m/	ol type-"mB	NA"	phora						
		/ di	b vrof-"tav	on•27828"							
(יחר	30	1162	011.27020							
(.05	10	odon start-	1							
		/ Ci	roduct-"act	⊥ in"							
		/p. /m	rotein id-"	 スカロク1 Q 5 7 "							
		/p. /+·	ranclation-	MCDDEVAALA	UDNCSCMCKAC	FACODA PRAVE	PSTVGRPRHOGV				
		/L. MT7/		PAOGKBGTL TI	VDNGSGMCICHO.	MODDIE MODIE I MODIE E br>MODIE E MODIE E MODIE E MODIE E	EVNELRVAPEEH				
		7777	GIIGQIODI NDVA	NDERWINOTMER		OAVT.ST.VASCR	TTCVVIDSCDCV				
		ידעי		UNTERIOLACE	TINTIMITIMI	ERGVSFTTTAE	RETURDIKEKLC				
		VV	YVALDFEOEMATAASSSSLEKSYELPDGQVITVGNERFRCPEALFQPSFLGMESAGIH								
		1 V I 도·ጥ·	ETSYNSTMKCDIDTRKDLYANTVLSGGTTMYPGIADRMOKEITALAPSTMKIKITAPP								
ETSINSIMKCDIDIRKDLIANTVISGGITMIPGIADRMQAEITALAPSIMAIA. BREVGENTGGGETIAGI GERROOMMICKOEVDESCOSTUURCES"											
ODTO	T N T	ER	KI SVWIGGSID	HODDIF QQIMI	.91001010010	I VIII(I(CI					
ORIGI			agattaataa	aggtaggga	astatatasa	aacaaaatta	ctactettat				
1	ggtttte		cayttactca	ageragerag	acycycyac	gacgaggeeg	ctcctccac				
61 101	ggttyac	adl	yyattegyaa	rgagaaga	tagaacaaaat	ggagacgacg	atatagaga				
101	tgtcul		tecategicy	geegaeeeee	ataaaaaaa	gccatggttg	cetterareta				
181	gaaggac	ccg	tacgtaggag	acgaggerea	gicaaayaya	ggtattetta	gggaggaggag				
241	ccccatt	gag	cacggtatcg	teaceaacig	gyatyataty	gagaayattt	gycaccacac				
301	cttctac	aat	gageteegtg	LIGCCCCaga	ayaycaccee	gtettette	atttacca				
361	acttaat	CCL	aaagetaace	gagaaaaaaa	gacceaaate	acycegaga	citicaacac				
421	tccggct	atg	tatgtcgcca	tccaagetgt	gettteeete	tacgetteeg	gacglaccac				
481	tggtgtc	gtt	ttagattccg	grgarggagt	cacccacact	glacclatct	algagggala				
541	tgctctt	CCT	cacgctatcc	tccgtctcga	. tetegetgga	cgigalella	cigacialli				
601	gatgaag	fatc	ctgactgaac	gtggttactc	cttcaccacg	accgctgaac	gtgaaategt				
661	gcgtgac	atc	aaggagaaac	tgtgctatgt	tgeeetegae	LLCyaacaag	adatygetae				
721	tgctgct	tca	tegtectete	tggagaagto	ttacgaactc	ceugaeggte	aayucatcac				
781	cgttgga	aac	gagcgcttcc	gttgccccga	ggeeetette	cageetteet	lettgggtat				
841	ggaatct	gct	ggaatccatg	agacgtette	caattcaatt	algaaatgcg	alalogatat				
901	tcgtaag	gac	ttgtacgcaa	ataccgtttt	atccggtgga	accaccatgt	accouggtat				
961	tgccgat	cgt	atgcagaagg	aaattacggc	letageteee	agcacgatga	aaaccaagat				

1021	tattgctccc	cctgagcgca	aatactccgt	ctggattggt	ggatctatcc	ttgcctcctt
1081	gtccaccttc	cagcagatgt	ggatctccaa	gcaggaatac	gatgagtccg	gaccgtcgat
1141	cgttcaccgt	aaatgcttct	aaaaggcgta	cttctctttt	cgtaatcacg	aacaaaactc
1201	atttttgctc	cattcatgta	ctatatgttg	tgtgtctatg	gtttgtttct	gtctcggtca
1261	gggtccggtg	ctgtatgtac	tactttcagt	agtaatagta	gcccaggtca	tactgaccaa
1321	gttggaaaca	ggtcctagtc	atcaattggc	tatgttcatg	agaatgcaat	aaactttaat
1381	atgcaatgat	aataaagatt	tgaaaattct	aaaaaaaaaa	gcaaaaaaaa	aaaaaaaaaa
1441	aaaaaaaaaa	aaaaa				

C.6 C.oncophora GABA β subunit

LOCUS	AY3	72759 2	L815 bp	mRNA	linear	INV	20-NOV-2003			
DEFINITION	Coo	peria oncop	ohora GABA	-beta subu	nit mRNA,	compl	ete cds.			
ACCESSION	AY3	72759								
VERSION	AY3	72759								
KEYWORDS										
SOURCE	Coo	peria oncom	ohora							
OPCANTSM	Coo	peria oncor	phora							
ONGHIATOTI	Fub	arvota: Mei	azoa: Nem	atoda. Chr	omadorea:	Rhabd	itida:			
	Ctr	aryota, neo	Frichostro	avloidea:	Cooperiic	lae. C	ooperia			
	1 /	bagag 1 ta	10151	igy foraca,	COOPCIIIC	<i>iuc</i> , c	ooperia.			
REFERENCE	т (. т (.	1 (bases 1 to 1815)								
AUTHORS	พวน	e,A.I. and	Prichard,	K.K.	The Empre	monio	anganhara			
TITLE	Seq	uence of GA	ABA-Deta Si	IDUNIC CDN	A ITOM COU	peria	οποορποτα			
JOURNAL	Unp	ublished	4.04.53							
REFERENCE	2 (1	bases 1 to	1815)							
AUTHORS	Nju	e,A.I. and	Prichard,	R.K.						
TITLE	Dir	ect Submiss	sion							
JOURNAL	Sub	mitted (22-	-AUG-2003)	Institute	e of Parasi	itolog	y, McGill			
	Uni	versity, 21	111 Lakes	shore Road	l, Ste. Anr	ie de	Bellevue,			
	QC 1	H9X 3V9, Ca	anada							
FEATURES	Lo	cation/Qual	lifiers							
source	1.	11815								
	/0	rganism="Co	operia one	cophora"						
	/m	/mol_type="mRNA"								
	/ d]	/db_xref="taxon:27828"								
CDS	67	671698								
020	/ С	/codon_start=1								
	/p	/product="GABA-beta subunit"								
	/ກ	/protein id="AAR21858"								
	/+-	/translation="MRMARRNGSKTVFLVSITLLLFSFTFTRGSEERMOVDRIRGGTR								
	SU	SLRNSTASONKTSYSNASSLLADLLADYDIRLRPGFGGDALLLTMDIIIASFDSISEV								
	NM1	NMDYTITMYLHOYWTDERLSWGGSVPIDEMTLSGEFSONIWVPDTFLANDKHSFLHEV								
	TTE1	TERNKMLRISVDGKIAYGMRLTSTLSCSMNLRNFPLDSONCTVEIESYGYTTSEVLML								
	WINTI	MNR PK AVHCVEDVDVPOFTTCVOTEDRVVCTATCCVORLCTVETEDTOTTTDCVCTTFOTV								
	T.D	I.DUALIZIANA GAVAGEMINALZAGIÁI CIMMANA WAMANI GAGABOALZA A GALAGIÁLA MINICEVY NOA PRARA A CALICITALA CAMANA CALACACA CALACACACACACACACACACACACACAC								
		TIMOERIERATIERAVIMURATORACOBCOCCEEOMDRCNGNADDBEGYDNILKERRADD PECAPTANIPMASEMINUEVICANCOPCOCCEEOMDRCNGNADDBEGYDNILKERRADDG								
	1111	UNINCE A E AETTERINE INCLUSI MOLLERA DE CI CONCONCIDATE CI COCOMULTERA ANI I DI MOLLERA COCONCIDER CONCONCIDER CI IDEI DOCERCIDERITA ANI I DI MOLLERA CONCECERA A CONCONCIDERI COCOMULTERA ANI CONCONCIDERI								
	- LLC-	DADA ANDERIDA DAMAGI DYOGAMANGI DADA CAMONGI DYOGAMAGI DYOGANGI DY								
	נפע	PEIPRIASIV	ZGMRARPSLS/	AUTEMANNALT	TOT #	2 THUNNG	VNANNAUENIN			
0.0.7.7.7.7	VRI	DAWATDKIZKI	VEDICEATEI	NIFIWGIISI	τQu"					
ORIGIN										
I ggtttt	aacc	cagttactca	agelgeel	ly aalyget	.gay caleya	iCada	ayuyuyyuy			
61 agagaa	atgc	gaatggccag	aagaaacgo	ja tcaaaga	leeg ttttt	JLCGL	CLCAALCACL			
121 ctgctt	CTTT	teteatteac	gttcacaco	ga ggateeg	laag agagga	itgca	agtegatega			
181 atccgc	ggtg	gaacgcgttc	acttogaaa	at tcaacgg	ctt cccaaa	iataa	aacgtcgtac			
241 tcgaat	gcga	gttcacttct	agcggatci	tt ctggctg	att atgata	attcg	gctacgtcca			
301 ggatte	ggtg	gcgatgcatt	acttetead	ca atggata	itca ttatco	jette	gttcgactca			
361 atatca	gaag	tcaacatgga	l ctacacgat	tc accatgt	att tacaco	caata	ctggacggac			
421 gaacgg	ctct	cgtggggtgg	f ttcagtaco	cg atagatg	jaaa tgacti	tgag	tggagaattc			
481 tctcaa	aata	tctgggtacc	ggatacat	te etegeea	lacg acaaad	cactc	attccttcat			
541 gaagtc	actg	aacgcaacaa	aatgctga	yg attagcg	rtcg atggaa	aaat	cgcctatgga			
601 atgagg	ctca	cgtcaacgct	tagetget	cg atgaacc	tac gcaatt	ttcc	attggattcg			
661 caaaat	tgta	cggtagaaat	agaatcata	at ggctaca	icta catcaç	yaagt	acttatgctg			
721 tggaat	cgtc	cgaaagctgt	tcatggcg	t gaggacg	sttg atgtco	caca	attcaccatc			
781 actggt	tatc	aaacagaaga	tcgtgttgt	cc agcacag	cta caggat	ccgta	tcagcggctg			
841 tcacta	gttt	tccaactgca	acgttcagt	t ggctatt	tta ttttt	caaac	atatctacct			





901	tgtgttttga	tcgttatgct	ctcatgggtc	tcgttctgga	ttaatcacga	ggcaaccagt
961	gcccgtgttg	ctttaggcat	cactaccgta	ctcactatga	caacaatctc	tactggtgtg
1021	cgccaatcac	tgcctcgaat	cagttatgtg	aaaagtatcg	atgtgtatct	ggtgatgtgt
1081	tttgtgttcg	tctttgctgc	tttacttgaa	tatgcggctg	tgaactattc	gtactggggt
1141	cggagatete	gaggaggtgg	cgaggaagga	tggccagtat	gcaattccaa	taaggacgac
1201	cgtgaaagtg	ctgtcaacat	caaggagtgg	ggtgtaccaa	gcagtctcat	tgatgagctt
1261	aggcaatcca	cttctgaccc	tcgaactcac	gatctgaact	cgaccgcttt	ggctggaact
1321	gctcccgagt	ccctttcctc	agcttcgagg	aaaaagcgcc	agtcaagtcc	gataccgtct
1381	ctttgcccct	ctggaccaaa	tggtattgac	gaagatgact	caccagaata	tccacgatat
1441	gcctcaaccg	ttcagggtat	gagagccagg	ccgtctttat	cagcaaggac	cgcacgacgg
1501	ctacgaacga	caaggttgag	agcgcaaagt	atgacgatgt	cgttgcatcg	gatgggtgtg
1561	cgagctcgca	aagcgctgcc	tcgtattcgg	gtccgagacg	tgaacgtcat	cgacaagtac
1621	tcgcggaccg	tctttcctat	ttgttttgtg	attttcaata	ttttctactg	gggttactac
1681	tccatcatac	agctgtaaat	gaccggtatc	gatagccaaa	gctacatcag	tgttcctgct
1741	caacatgaag	catccccttt	taaagaaatc	aataaagatc	tcaaaacaaa	aaaaaaaaaa
1801	aaaaaaaaaa	aaaaa				

//

APPENDIX D

Ethics Certificate (Animal Use Protocol)